## COMPUTATIONAL MODELING OF COGNITIVE PROCESSES FOR CONTINUOUS IN-HOME ASSESSMENT OF COGNITIVE PERFORMANCE

By

Stuart Hagler

### A DISSERTATION

Presented to the Department of Biomedical Engineering of the Oregon Health & Science University School of Medicine in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Biomedical Engineering

September 2014

Department of Biomedical Engineering School of Medicine Oregon Health & Science University

### CERTIFICATE OF APPROVAL

This is to certify that the Ph.D. dissertation of Stuart Hagler has been approved

Michael (Misha) Pavel, Ph.D. Professor, Thesis Advisor

Paul Gorman, M.D. Associate Professor

András Gruber, M.D. Associate Professor

Peter Jacobs, Ph.D. Assistant Professor

Holly Jimison, Ph.D. Professor

Jeffrey Kaye, M.D. Professor

© 2014 Stuart Hagler All Rights Reserved To my wife May and my daughters Emmy and Mika

## **Table of Contents**

| Table of Contents i  |
|--|
| Table of Figuresvii  |
| Table of Abbreviations xi  |
| Table of Symbols xii   |
| Acknowledgments xiv  |
| Abstract xvi   |
| Introduction1  |
| Chapter 1 – Background and Introduction 2                                    |
| 1.1 Overview   |
| 1.2 Background   |
| 1.2.1 Economics of Healthcare for an Aging Population                        |
| 1.2.2 Institution-Centered Model of Preventive Healthcare                    |
| 1.2.3 Institution-Centered Measurement Example: Walking Speed                |
| 1.2.4 Institution-Centered Measurement Example: Trail-Making Test (TMT) 6    |
| 1.3 Introduction   |
| 1.3.1 User-Centered Model of Preventive Healthcare                           |
| 1.3.2 User-Centered, In-Home Cognitive Monitoring of Older Adults            |
| 1.3.3 Computational Models in User-Centered, In-Home Cognitive Monitoring 10 |

| 1.4 Thesis Structure  | 12      |
|---|---------|
| 1.4.1 Part I – In-Home Monitoring of Walking Speed                  | 12      |
| 1.4.2 Part II – In-Home Monitoring of Trail-Making Test Performance | 13      |
| Part I – In-Home Monitoring of Walking Speed                        |         |
| Chapter 2 – Unobtrusive and Ubiquitous In-Home Monitoring: A Metho  | odology |
| for Continuous Assessment of Gait Velocity in Elders                |         |
| 2.0 Abstract  | 17      |
| 2.1 Introduction  |         |
| 2.2 System Description and Data Collection                          |         |
| 2.3 Data Modeling and Analysis                                      |         |
| 2.3.1 The Linear Model  | 23      |
| 2.3.2 Estimation of Gait Velocity                                   | 27      |
| 2.3.3 The Calibrated Sensor Line                                    | 29      |
| 2.3.4 The Uncalibrated Sensor Line                                  | 30      |
| 2.4 Experimental Verification                                       |         |
| 2.4.1 Experimental Description                                      | 31      |
| 2.4.2 Experimental Results  | 33      |
| 2.5 Discussion  |         |
| 2.6 Conclusion  | 39      |
| Part II – In-Home Monitoring of Trail-Making Test Performance       | 40      |
| Chapter 3 – Assessing Executive Function Using a Computer Game:     |         |
| Computational Modeling of Cognitive Processes                       | 41      |
| 3.0 Abstract  | 41      |

| 3.1 Introduction   |         |
|--|---------|
| 3.2 Overview of Two Connect-the-Dots Tasks                         |         |
| 3.2.1 Trail-Making Test  | 44      |
| 3.2.2 Scavenger Hunt   | 47      |
| 3.2.3 Differences Between SH and TMT                               | 50      |
| 3.3 Connect-the-Dots Model   | 51      |
| 3.3.1 Recall and Update Stage                                      | 54      |
| 3.3.2 Search Stage   | 54      |
| 3.3.3 Motor Stage  | 55      |
| 3.3.4 Total Time   | 56      |
| 3.3.5 Relating SH to TMT   | 57      |
| 3.4 Analyzing SH Play  | 58      |
| 3.4.1 Estimating Motor Parameters                                  | 59      |
| 3.4.2 Estimating Cognitive Parameters                              | 60      |
| 3.5 Estimating TMT Scores  | 61      |
| 3.5.1 TMT Score Estimator  | 61      |
| 3.5.2 Estimating Global Parameters                                 | 63      |
| 3.6 Empirical Study  | 64      |
| 3.7 Discussion   | 71      |
| 3.8 Conclusion   | 74      |
| 3.9 Appendix   | 75      |
| Chapter 4 – Towards the Unobtrusive and Ubiquitous In-Home Monitor | ring of |
| Cognitive Performance Using Mouse Dynamics                         | 77      |

| 4.0 Abstract  | 77        |
|---|-----------|
| 4.1 Introduction  | 77        |
| 4.2 Cognitive Performance (Executive Function)                      | 81        |
| 4.2.1 Trail-Making Test   | 81        |
| 4.2.2 Connect-the-Dots Model  | 83        |
| 4.3 Measuring In-Home MD  | 86        |
| 4.3.1 Preliminaries   | 87        |
| 4.3.2 Storing Mouse-Position Data                                   | 88        |
| 4.3.3 Identifying Mouse Movements                                   | 89        |
| 4.3.4 Estimating Motor Parameters                                   |           |
| 4.3.5 Modeling Mouse Movements                                      |           |
| 4.4 Proxies for Recall and Search Performance                       |           |
| 4.4.1 Cut Speed   |           |
| 4.4.2 Digit Symbol Test   | 100       |
| 4.5 Empirical Study   | 100       |
| 4.5.1 Measures of Motor Performance                                 | 102       |
| 4.5.2 Measures of Cognitive Performance (Executive Function)        | 108       |
| 4.6 Discussion  | 117       |
| 4.7 Conclusion  | 121       |
| Chapter 5 – On the Relationship of Set-Switching, Movement Times, a | nd Errors |
| in Trail-Making Test Part B for Older Adults                        | 122       |
| 5.0 Abstract  | 122       |
| 5.1 Introduction  | 123       |

| 5.2 Background  | 125 |
|---|-----|
| 5.2.1 Trail-Making Test   | 125 |
| 5.2.2 Connect-the-Dots Model  | 126 |
| 5.3 In-Home Mouse Dynamics and TMT                                  | 130 |
| 5.4 Single-Task and Dual-Task in TMT                                | 132 |
| 5.5 Cognitive Performance and Scavenger Hunt Computer Game          | 135 |
| 5.5.1 Estimating Motor Performance                                  | 136 |
| 5.5.2 Estimating Recall and Search Performance                      | 138 |
| 5.5.3 Estimating TMT Performance                                    | 139 |
| 5.6 Errors in TMT and SH  | 140 |
| 5.7 Notation Convention   | 141 |
| 5.8 Empirical Study   | 142 |
| 5.9 Discussion  | 155 |
| 5.10 Conclusion   | 161 |
| 5.11 Appendices   | 161 |
| 5.11.1 Appendix 1   | 161 |
| 5.11.2 Appendix 2   | 162 |
| Conclusion  | 166 |
| Chapter 6 – Conclusion and Future Directions                        | 167 |
| 6.1 Conclusion  | 167 |
| 6.1.1 Part I – In-Home Monitoring of Walking Speed                  | 167 |
| 6.1.2 Part II – In-Home Monitoring of Trail-Making Test Performance | 168 |
| 6.1.3 Connect-the-Dots Model  | 169 |

| References                                       |     |
|--|-----|
| 6.3 Appendix                                     |     |
| 6.2.6 Optimal Control Model of Mouse Movements   |     |
| 6.2.5 Measure Dual-Tasking During Mouse Movement |     |
| 6.2.4 Reduce In-Home MD Processing Time          |     |
| 6.2.3 Analyze Errors Made During SH              |     |
| 6.2.2 Augment Computer Game Suite                | 175 |
| 6.2.1 Measure Walking Cadence In-Home            |     |
| 6.2 Future Directions                            |     |
| 6.1.4 Using the Three Techniques Together        | 173 |

## **Table of Figures**

| Figure 2.1. A motion sensor line for measuring walking speed where the four sensors   |
|---|
| are placed 0.61 m apart and are installed on a ceiling typically 2.54 m high 21   |
| Figure 2.2. Schematic of a person walking through a sensor line containing four sensors   |
| with the fields of view and the locations of the $\tilde{x}_i$ and $x_i$ shown  |
| Figure 2.3. Combined walking speed data for all subjects for the 5 sensor lines of 4  |
| sensors (the various shapes indicate data from different sensor lines), using a calibration   |
| factor calibrated to the walking speed measured by the gait mat. The sign of the  |
| estimated speed differentiates "forward" walks (positive values) versus "return" walks  |
| (negative values)   |
|   |
| Figure 2.4. Combined walking speed data for all subjects for the 5 sensor lines of 4  |
| <b>Figure 2.4.</b> Combined walking speed data for all subjects for the 5 sensor lines of 4 sensors (the various shapes indicate data from different sensor lines), using estimated   |
|   |
| sensors (the various shapes indicate data from different sensor lines), using estimated   |
| sensors (the various shapes indicate data from different sensor lines), using estimated calibration factors. The sign of the estimated speed differentiates "forward" walks   |
| sensors (the various shapes indicate data from different sensor lines), using estimated calibration factors. The sign of the estimated speed differentiates "forward" walks (positive values) versus "return" walks (negative values)       |
| sensors (the various shapes indicate data from different sensor lines), using estimated<br>calibration factors. The sign of the estimated speed differentiates "forward" walks<br>(positive values) versus "return" walks (negative values) |
| sensors (the various shapes indicate data from different sensor lines), using estimated<br>calibration factors. The sign of the estimated speed differentiates "forward" walks<br>(positive values) versus "return" walks (negative values) |

Figure 3.2. A typical Scavenger Hunt round. The board for a round of Scavenger Hunt showing the time remaining in the game (27.2 s), search string ('B2A1'), cumulative Figure 3.3. Additive stages move model. The process of selecting the next target in the sequence involves three sequential stages of (1) recalling the next target, (2) serially searching for the next target by considering the available targets one after another, and Figure 3.4. Actual vs. estimated TMT scores across subjects. Each of the 23 subjects has two values shown, one for TMT-A and one for TMT-B, each representing the average of the three administrations of TMT. The model fit has  $R^2 = 0.82$  and p < Figure 4.1. Mouse speed density for a single session. The kernel smoothed density of the interpolated regularly sampled mouse speed time series for a single computer usage session by a single subject. The speed  $v_{cut} = 67$  counts/s separates velocities associated with mouse activity with those associated with the mouse being idle. The session lasted Figure 4.2. Sample mouse-position data. A sample segment of horizontal and vertical mouse-position data (in mouse-space) covering a 5 s period. The dots correspond to individual mouse-position measurements. There are 8 separate mouse movements indicated by alternating series of grey and black dots. The intervals between the separate 

**Figure 4.3.** Example class I and class II mouse movements. The dashed line indicates the observed mouse speed time-course of the movement, and the solid line indicates the

viii

| Figure 5.1. Stages of a TMT-B move. The process of selecting the next target in the         |
|---|
| sequence for TMT-B involves four sequential stages of (1) switching sets (2) recalling      |
| the next target, (3) serially searching for the next target by considering the available    |
| targets one after another, and (4) physically moving the mouse so that the cursor is on the |
| target and clicking. The set-switching stage is made a secondary cognitive task with the    |
| motor stage whenever possible   |
| <b>Figure 5.2.</b> Correlations of $\tau_M$ with other parameters                           |
| Figure 5.3. Actual vs. estimated TMT scores across subjects. Each of the 23 subjects        |
| has two values shown, one for TMT-A and one for TMT-B, each representing the                |
| average of the three administrations of TMT. The model fit has $R^2 = 0.87$ and p <         |
| 0.0001. A line with slope one passing through the origin is shown for reference 152         |

### **Table of Abbreviations**

| ADL             | Activities of Daily Living                           |
|-----------------|--|
| DST             | Digit-Symbol Test                                    |
| LHS             | Left-Hand Side                                       |
| MD              | Mouse Dynamics                                       |
| MMSE            | Mini Mental State Exam                               |
| OLS             | Ordinary Least Squares                               |
| PIR             | Passive Infrared                                     |
|                 |  |
| RFID            | Radio Frequency Identification                       |
| RFID<br>RHS     | Radio Frequency Identification<br>Right-Hand Side    |
|                 |  |
| RHS             | Right-Hand Side                                      |
| RHS<br>RV       | Right-Hand Side<br>Random Variable                   |
| RHS<br>RV<br>SH | Right-Hand Side<br>Random Variable<br>Scavenger Hunt |

# **Table of Symbols**

| $E\left\{ X ight\} ,\left\langle X ight angle$ | Mean Value of <i>X</i>                             |
|--|--|
| $M\left\{X\right\}$                            | Median Value of X                                  |
| $\sigma\{X\}$                                  | Standard Deviation of <i>X</i>                     |
| $IQR\{X\}$                                     | Inter-Quartile Range of <i>X</i>                   |
| D  | Total Movement Distance                            |
| Ν  | Number of Errors on TMT                            |
| S  | Test Score   |
| Р  | Probability of Making an Error on Each Move of TMT |
| Т  | Movement Time                                      |
| W  | Target Width                                       |
| <i>a</i> , <i>b</i>                            | Motor Parameters of Fitts' Law                     |
| р  | Probability of Making an Error on Each Move of SH  |
| ν  | Walking Speed                                      |
| V <sub>cut</sub>                               | Cut Speed  |
| Δ  | Net Movement Distance                              |
| α,β  | Recall Time Transformation Parameters              |
| γ  | Dual-Tasking Transformation Parameter              |
| χ,ξ  | Motor Complexity                                   |

| К                             | Search Complexity                 |
|-------------------------------|-----------------------------------|
| ρ                             | $10^{\nu_{cut}/\nu_0}$            |
| $	au_R$                       | Characteristic Recall Time        |
| $	au_s$                       | Characteristic Search Time        |
| $	au_{\scriptscriptstyle Sw}$ | Characteristic Set-Switching Time |

#### Acknowledgments

I would like to thank my thesis advisor, Dr. Misha Pavel for his support and guidance during the entire process of producing this thesis. In particular, this thesis would never have happened without Misha's encouragement to complete a doctorate when I was working as a member of his research staff. I would also like to thank my Thesis Advisory Committee, all of whom also agreed to serve on the Thesis Defense Committee, Drs. Paul Gorman, Holly Jimison, and Peter Jacobs. I am grateful in particular for the care that they took in going through and commenting on earlier drafts of this document. This thesis would be a much poorer document without their participation. Further, I would like to thank Dr. András Gruber for agreeing to participate in both committees rather late in the process, and Dr. Jeff Kaye for agreeing to serve on the Thesis Defense Committee. I would like to thank the staff and students at OHSU Dr. Daniel Austin, Dr. Zach Beattie, Nicole Larimer, Julia Leach, Nora Mattek, Dr. Zephy McKanna, Johanna Peterson, Thomas Riley, Sara Saperstein, Jon Yeargers, and Tracy Zitzelberger for their support. In particular, I would like to thank Dr. Tamara Hayes who I also worked for as research staff at OHSU and gave me valuable support when I began work as a student. To our loss, she is no longer with us; she will be greatly missed. Finally, I must thank the support from my family, my parents Joe and Helen Hagler, and most importantly, my wife May.

I would like to acknowledge the staff from the Oregon Center for Aging and

Technology (ORCATECH) who assisted in the study presented in Chapter 2, and for making available the mouse movement data from the Intelligent Systems for Detection of Aging Changes (ISAAC) study analyzed in Chapter 4. I would also like to thank Holly Jimison and Misha Pavel for making available the Scavenger Hunt game data analyzed in Chapters 3 and 5.

This work was supported by the Alzheimer's Association, the National Institute of Health, the National Institute on Aging, the National Institute on Standards and Technology's Advanced Technology Projects, and the National Science Foundation under the Grants 1111722, 5RC1AG36121-2, ASMMI0116ST, P30AG008017, P30AG024978, and R01AG024059. This work was partially funded by Intel Corporation. The content is solely the responsibility of the author and does not necessarily represent the official views of the Alzheimer's Association, the National Institute on Standards and Technology's Advanced Technology Projects, the National Institute on Standards and Technology's Advanced Technology Projects, the National Science Foundation, or the Intel Corporation.

#### Abstract

### Computational Modeling of Cognitive Processes for Continuous In-Home Assessment of Cognitive Performance

Stuart Hagler

Department of Biomedical Engineering School of Medicine Oregon Health & Science University

September 2014 Thesis Advisor: Michael (Misha) Pavel, Ph.D.

A rapidly growing proportion of older adults in the populations of the US, EU, and Japan, combined with an increasing prevalence of conditions associated with aging, and further exacerbated by behavioral issues, comprise significant challenges to healthcare delivery. The development of cost-effective, proactive, and preventive approaches, focused on quality-of-life is therefore of utmost societal importance. Replacing institution-centered (clinic-centered) reactive approaches with user-centered sensor and computer-aided care is emerging as a potential solution that would allow early detection and intervention through continuous monitoring and assessment. In contrast to well controlled, in-clinic measurements, however, the context of the user-centered behavioral observations is typically unknown, and the quantities measured by the sensors are usually remote from the quantities of interest. These aspects pose significant challenges for the development of robust algorithms. The only way to mitigate these impediments is the development of computational models that relate the observable quantities to meaningful The focus of this thesis is the development of several algorithmic behavioral metrics. techniques based on computational models of behaviors. These serve as examples of approaches that would enable inference of the patient state from unobtrusive, but continuous monitoring of everyday, real-life behaviors. I examine three techniques for performing in-home monitoring of cognitive performance. The first technique monitors an older adult's walking speed unobtrusively in the home using motion sensors placed on the ceiling or wall. The second technique monitors the older adult's cognitive performance by observing how the older adult plays a specially designed computer game. The third technique monitors the older adult's cognitive performance by observing how the older adult uses a computer in the course of everyday, real-life computer usage. In the case of the first technique, the walking speed serves as immediately behavioral metric, while in the cases of the second and third techniques, we define the behavioral metrics as the parameters of computational models of the behavior of playing the game or using the computer, respectively. The model-based inferences in the second and third techniques characterize the older adults' ability to utilize the component cognitive functions in order to carry out the task of playing the game or using the computer. The computational models that are used in the second and third techniques provide descriptions of how the behaviors are carried out physically, and the associated behavioral metrics characterize the older adults' physical performance of specific aspects of the behavior being measured. In this thesis, I demonstrate that the unobtrusive performance measurements combined with the computational model can provide

estimates of cognitive functionality similar to that obtained in the controlled environment of a clinical assessment using standard neuropsychological tests – the walking speed, and the Trail-Making Test. In addition, I argue that the observations of older adults playing the computer game and going about everyday, real-life computer usage support a computational model of the Trail-Making Test in which set-switching is performed as a dual-task with movement during Trail-Making Test Part B. Introduction

#### **Chapter 1 – Background and Introduction**

#### 1.1 Overview

Larger portions of the populations of the European Union, Japan, and the United States are living to older ages. The associated costs of providing healthcare to a larger older population pose significant challenges to the economies of these regions. The costs of the treatment of a medical issue can be reduced by detecting and treating the issue earlier. Thus, cost-effective methods of early detection of medical issues can reduce the costs of providing healthcare to the growing older population.

The present, institution-centered (clinic-centered), model of providing healthcare through regular, clinical visits to a trained clinician has a number of shortcomings as a cost-effective means of providing early detection disease. These shortcomings relate to the necessary infrequency of clinic visits for economic and practical reasons. Alternative, user-centered, models of overall lifestyle health management by the ubiquitous and continuous application of technology to healthcare provide a potentially cost-effective method of constant healthcare monitoring.

The performance of technologies applied to the ubiquitous and continuous monitoring of subject performance for the purpose of early detection of disease can be enhanced greatly by the use of computational models of the behaviors that the technologies are being measuring. These computational models allow the clinician to make better measurements using the available, cost-effective technologies, and relate the ubiquitous and continuous measurements to standard clinical measurements.

In this thesis, I demonstrate that the unobtrusive performance measurements combined with the computational model can provide estimates of cognitive functionality similar to that obtained in the controlled environment of a clinical assessment using standard neuropsychological tests – the walking speed, and the Trail-Making Test. In addition, I argue that the observations of older adults playing the computer game and going about everyday, real-life computer usage support a computational model of the Trail-Making Test in which set-switching is performed as a dual-task with movement during Trail-Making Test Part B.

#### **1.2 Background**

#### 1.2.1 Economics of Healthcare for an Aging Population

The populations of the European Union, Japan, and the United States are becoming older with a larger proportion of the population of advanced age. [1, 2] In the United States, the population of American aged 65 years or older will double in the next 25 years to about 72 million. [3] By 2030, older adults will account for roughly 20% of the U.S. population. [3] The aging population of these regions comes with a variety of economic consequences. [1, 2] Projections suggest that health and long-term care will account for about half of the increase in age-related social expenditures by 2050. [4] Improving quality of life and providing adequate medical care for the rising number of elderly while keeping health care costs under control has, in recent years, become a major problem. [5] The rising cost of health care potentially represents a significant threat to the long-term economic security of workers and retirees. [6] The costs associated with the health and

long-term care of older adults can be brought down by appropriate and well-coordinate health and social policies that keep down the required amount of health services. [7]

Preventive healthcare seeks to avert or avoid disease. We can divide preventive healthcare into three kinds of disease prevention: Primary prevention – modifying unhealthy behaviors, Secondary prevention – detecting diseases or their risk at early stages, Tertiary prevention – the effort to avoid or divert the complications from diseases after they have developed. [8] Current, institution-centered, medical care focuses on tertiary prevention. [8] A well-defined and effective package of services promoting primary and secondary prevention at reasonable cost should offer good economic value. [8] In the cases of dementia and Alzheimer's disease, the expected benefits of early detection and intervention would make up for the expected costs of putting a system in place to perform early detection and intervention in the projected savings across the disease course. [9-11]

#### 1.2.2 Institution-Centered Model of Preventive Healthcare

The current model for providing healthcare is institution-centered and requires a trained-professional to meet with the subject and make performance measurements. [12] Clinical evaluation of cognitive and motor performance suffers from a number of drawbacks. Frequent visits to the clinic for assessment are impractical as they are of prohibitive cost and it is difficult for subjects to make frequent trips to a clinic. Infrequent measurements report only the net change between measurement times and cannot distinguish between functional changes occurring slowly over time and abrupt functional changes, which may have different causes. Infrequent measurements do not detect changes when they happen which may reduce the ability of a clinician to provide

intervention or reduce the effect of an intervention. Performance in the clinic may be atypical and not representative of the subject's everyday performance.

#### 1.2.3 Institution-Centered Measurement Example: Walking Speed

One commonly used clinical measure of performance is the walking speed. We provide an extended description of the walking speed at this point both to illustrate clinical evaluation techniques through a specific example as well as to provide relevant clinical background for the technique of measuring walking speed continuously in the home that we present in Part I Chapter 2.

A clinician measures the walking speed by observing the time the subject takes to walk a fixed distance along a straight line. The most used walking speed tests use: 2.44 m, 4 m, and 6 m, carried out at usual walking speed. [13] The walking speed is simply the distance walked divided by the observed time taken.

Walking speed has been found to decline as adults age, [14, 15] and has been shown to be associated with survival in older adults. [16, 17] Evidence supports the use of walking speed tests as predictors of adverse results related with health in older adults, [13] and has shown walking speed to be a quantitative estimate of risk of future hospitalization. [18] Slower walking speed has been demonstrated in dementia patients compared to controls [19] and has been shown to precede cognitive impairment [20] and dementia [21], and timed walk has been used as a partial characterization of lower extremity function which has been shown to predict disability [22, 23]. The slowing of walking speed appears to take place secondary to the slowing of processing speed in the path leading to dementia. [24] Other studies have shown a relationship between walking speed and cognition. [25, 26]

#### 1.2.4 Institution-Centered Measurement Example: Trail-Making Test (TMT)

Another commonly used clinical measure of performance is the Trail-Making Test. We provide an extended description of the walking speed at this point both to illustrate clinical evaluation techniques through a another specific example as well as to provide relevant clinical background for the techniques of measuring cognitive performance continuously in the home through interactions with a computer that we present in Part II Chapters 3, 4, and 5.

TMT consists of two connect-the-dots tasks (TMT-A and TMT-B) which the subject completes by using a pen to draw a single line through a series of targets on a test page. Each test takes the form of a standard 8.5"x11" sheet of paper on which is printed 25 small (12mm diameter) targets, seemingly randomly scattered circles containing a label that may be a letter or a number. For TMT-A, the targets are labeled with numbers from 1 to 25, with one number labeling each circle, while in TMT-B targets are labeled with all letters from A to L and all numbers from 1 to 13 with one letter or number labeling each circle. The individual completes the TMT-A by using a pen to draw a line (the "trail") through all the circles in ascending numerical order of the labels (i.e.  $(1,2,3,\ldots,24,25)$ ), while the TMT-B is completed by drawing the line through the circles in ascending alphanumeric order (i.e. '1,A,2,B,...,L,13'). The score on each part of the test is the time the subject needs to complete each task (there is no time limit). If the subject makes an error on the test (drawing the line through an incorrect target), the test administrator stops the subject as soon as the error is noted by the administrator and returns the subject to the last correctly selected target; timing of the subject is not stopped during this error recovery process. The total numbers of errors made on each part are included with the test score. The difference between TMT-A and TMT-B is the use of the numeric and alphanumeric sequence of labels respectively. The alphanumeric label sequences introduce the additional complication of set-switching in which the subject must not only recall the next element in an alphabetic or a numeric sequence, but must also switch sets between the alphabetic and numeric sequences.

Normative TMT data give scores of 42 s on TMT-A and 100 s on TMT-B for ages 75-79, and 55 s on TMT-A and 130 s on TMT-B and for ages 80-84. [27] The test re-test reliabilities for TMT have been observed to be  $R^2 = 0.56$  for TMT-A and  $R^2 = 0.72$  for TMT-B for a cohort of normal adult controls. [28] These controls had observed scores of 47 ± 25 s on TMT-A and 120 ± 86 s on TMT-B. [28] Guidelines for the administration of TMT are provided in [29]. A neuropsychologist normally administers TMT in an office setting once every 6 to 12 months to reduce practice effects associated with repeated completion of standardized tests. [30-32]

TMT is one of the most clinically useful neuropsychological tests and is used routinely in the diagnosis of many neurological conditions (Parkinson's, Alzheimer's and dementias, and general cognitive decline). [33] TMT measures sensorimotor ability, working memory, and set-switching ability. [34, 35] TMT-A indicates mainly sensorimotor abilities, while the TMT-B indicates working memory and set-switching abilities over and above what is indicated by TMT-A. [34] The difference of TMT-A and TMT-B,  $S^A$  and  $S^B$  respectively, given by  $S^B - S^A$  is considered to provide a measure of cognitive and set-shifting ability, [36] while the ratio given by  $S^B / S^A$  has been shown to provide an index of executive function. [37] TMT performance has been found to decline as adults age. [38-43] Performance on the TMT is a strong, independent predictor of mobility impairment, accelerated decline in lower extremity function, and death in older adults living in the community. [44] In older adults, performance on the TMTs has been associated with gambling problems [45], as well as being a predictor of failure on standard driving tests. [46] Poor performance on TMT has been associated with altered dual-task prioritization in older adults. [47]

Normal adult controls do make errors on TMT, with 12% and 35% of normal controls observed to make at least one error on TMT-A and TMT-B respectively. [48] TMT scores and error counts have been shown to be independently meaningful and both to be of clinical utility for assessing individuals for dementia. [49]

#### **1.3 Introduction**

#### 1.3.1 User-Centered Model of Preventive Healthcare

The user-centered model of healthcare is one providing for overall lifestyle health management using technology to implement a system of pervasive healthcare. [12] Pervasive healthcare seeks to apply principles of pervasive computing to healthcare and make techniques of preventive healthcare appear everywhere and anywhere. [50] It addresses new emerging research questions, represents a novel approach, designs new types of technologies, and applies a new kind of research method. [51] Several methodologies have been proposed to address the application of user-centered healthcare to the problem of providing cost-effective, preventive healthcare to a growing population of older adults through the use of technology to develop systems that promote aging in place [52] and the use of pervasive healthcare [53] to help alleviate the burden placed on health care providers. One of the underlying themes of these approaches is to employ technology such as wireless networks combined with novel sensing systems to gather and interpret data in non-health care settings such as the home environment. By virtue of its ubiquitous presence, pervasive healthcare provides an alternative to infrequent clinical measurement capable of providing continuous measurement of real-life behaviors. [53, 52]

We can apply the ideas behind pervasive healthcare to address the problems inherent in using clinical measurements of performance to detect disease by placing technology in the subject's home to make continuous measurements of the subject's performance. Continuous in-home measurement can alleviate the costs associated with frequent trips to the clinic while at the same time providing frequent measurements of performance. They can detect functional changes occurring slowly over time as well as abrupt functional changes. Finally, they are representative of typical subject performance.

#### 1.3.2 User-Centered, In-Home Cognitive Monitoring of Older Adults

Older adults want to maintain their independence as long as possible, [54] and decline in cognitive performance is a major concern for people as they age. [54, 8] Measurable decline in cognitive performance has been observed in nearly half of all people over age 85. [55] In the institution-centered model, cognitive performance may be screened at a physician's clinic, though the results are coarse and do not perform well as an early detector of problems. [56]

The user-centered model of preventive health care provides a technological platform for the long-term monitoring of cognitive performance in older adults. It allows for frequent measurement of cognitive performance using everyday behaviors. Thus, the measured cognitive performance represents typical performance rather than the (possibly somewhat different) performance displayed in the clinic setting. Cognitive monitoring allows not only for the early detection of cognitive decline but also allows one to track the efficacy of interventions [57] used to reverse detected cognitive decline. Such interventions include forms of cognitive health coaching [58] including: (1) cognitive exercise, [59, 58, 60-62] (2) physical exercise, [60, 61, 63, 62] (3) sleep management, [64] (4) socialization, [61, 65] or (5) diet. [61] Studies of potential pharmacological treatments of cognitive decline can also benefit from user-centered cognitive monitoring as a way to measure the efficacy of the drug. [60]

#### 1.3.3 Computational Models in User-Centered, In-Home Cognitive Monitoring

The application of computational models to in-home monitoring has largely centered on the problem of organizing and presenting the data to clinicians or care-givers (see e.g. [66-70]). Our interest is not in the application of computational modeling techniques for making sense of data coming from multiple sources. Instead, our interest is in using computational models to develop a detailed, physical description of the behavior that we are measuring.

Researchers have applied a variety of technologies to user-centered, in-home monitoring of older adults. A sample of these technologies include: motion sensors placed in the home for the monitoring of activity [71, 69, 70], activity monitoring using various wearable sensors [72-77], balance assessment using a Nintendo Wii Fit [78], gait measurement using a Microsoft Kinect [79] or a webcam [80], monitoring of breathing during sleep using sensors attached to the bed [81]. In addition, many smart home technologies involve monitoring resident behavior and appropriately responding to residents' prompts. [82, 83] A related application of user-centered monitoring to the care

of older adults is the placement of sensors in the home for the purpose of detecting falls. [84, 85] A survey of further wearable and mobile sensors may be found in [86].

For the application of any of these technologies to user-centered, in-home monitoring to be immediately clinically useful, one must show that one can use the technologies to derive estimates of established clinical measures. Thus, a key problem in making the user-centered healthcare model work for in-home monitoring is that of relating the quantities that can be measured continuously to clinically established measures of subject performance. A computational model that describes both the behavior that is measured in the home and a related behavior that is measured in the clinic provides a principled way to establish such a relationship between in-home and in-clinic measurements.

A computational (or mathematical) model of a behavior allows one to calculate how a subject carries the behavior out given a description of the task the subject is trying to perform. For a specified task, we are able to compare the behavior predicted by the model to the observed behavior and quantify how well the model performs. In general, a computational model will contain a number of free parameters. The model should predict somewhat different performances of the behavior depending on the values of the free parameters. We can use a computational model to characterize a subject by finding the values of the free parameters in the model that cause the behavior predicted by the model to approximate most closely the behavior measured for the subject for a given specified task.

In a good model, the values of the free parameters should remain essentially constant across a range of specified tasks so that the same model closely approximates the observed behaviors for a number of tasks using the same free parameter values. Thus, in the case of a good model, the free parameters provide a characterization of the subject. When a good model also describes a behavior used in performing a clinical measurement, then we can use the measured parameter values to predict performance on the clinical measurement. By providing a detailed description of the behavior being measured using user-centered, in-home monitoring techniques, a good computational model facilitates a more detailed analysis of the observations by providing a set of free parameters that can be fit to the data and so used to characterize the subject and estimate performance on related clinical measures. The computational model also allows us to put a principled structure on the data so we can make the best use of the available data for making performance estimates.

#### **1.4 Thesis Structure**

In this thesis, we look at techniques for user-centered, in-home monitoring of cognitive performance using available cost-effective technologies. We use the cost-effective technologies to produce cognitive performance estimates in terms of the two clinical measures that we gave detailed descriptions for above: (1) walking speed, and (2) the Trail-Making Test. We make measurements of the walking speed in the home using an inexpensive system of motion sensors placed in the home. We make measurements from which we can estimate TMT performance in the home by observing the subject's interactions with a personal computer.

#### 1.4.1 Part I – In-Home Monitoring of Walking Speed

In Part I, we develop a technique for making measurements of walking speed in the home using a system of passive infrared motion sensors. This technique produces a continuous measurement of a subject's in-home walking speeds.

In Chapter 2 in Part I, we develop the technique for measuring walking speed in the home using passive infrared motion sensors. In this Chapter, we develop a computational model of the process by which the motion sensors detect the subject walking beneath them and use the model to produce an estimate of the walking speed. We validate the walking speed estimates by comparing the estimates to measurements of walking speed made using a GAITRite® Walkway System gait mat.

My specific contributions in Chapter 2 are the setting up and running of the empirical study and the computational modeling of the process by which the system of motion sensors estimate the walking speed.

#### 1.4.2 Part II – In-Home Monitoring of Trail-Making Test Performance

In Part II, we develop two techniques for estimating performance on TMT in the home using measurements of subjects' interactions with a computer. These techniques produce continuous estimates of cognitive performance as described by sets of several measured parameters. We also show that the data obtained from in-home measurements of computer usage support a model of TMT in which the subject dual-tasks set-switching and movement during TMT-B.

In Chapter 3 in Part II, we develop a computational model of the process a subject goes through when playing a simple computer game that was designed to mimic TMT. We use this model to derive a set of several parameters that we can measure by observing how a subject plays the game and that we can use to characterize the subject's cognitive performance. We validate the clinical utility of the measured parameters by constructing a model of TMT and using that model together with the measured parameters to generate an estimate of a subject's performance on TMT. We compare these estimates of subjects' performances on TMT to measurements of actual performances on TMT.

My specific contributions in Chapter 3 are the construction of the computational model that describes TMT and the computer game (the connect-the-dots model), the analysis of the subjects' play of the computer game to produce measurements of the various parameters present in the computational model, and the validation of the technique by producing an estimate of TMT performance.

In Chapter 4 in Part II, we develop a technique to store and analyze mouse position data observed during a subject's everyday (real-life) computer usage. We use this technique to derive a set of several parameters that we can measure by observing how a subject uses the computer mouse. We use the model that we have developed in Chapter 3 to provide a qualitative model of the cognitive process that a subject uses when going about using a computer. The model indicates how we can relate measurements obtained from a subject's everyday computer usage to that subject's performance on TMT. We validate the clinical utility of the measured parameter values by comparing these estimates of subjects' performances on TMT to measurements of actual performances on TMT. We observe that the measured parameter values perform much better in producing an estimate of performance on TMT-B than in producing an estimate of performance on TMT-A.

My specific contributions in Chapter 4 are the technique to identify individual mouse movements, the analysis of the resulting mouse movement data to produce performance metrics related to mouse movements, and the validation of the performance metrics by producing an estimate of TMT performance.

14
In Chapter 5 in Part II, we provide an improvement to the computational model that we have used in Chapters 3 and 4 based on observations made when applying the model to the data that we have obtained in Chapters 3 and 4. We observe that the parameters measured in Chapter 4 perform significantly better when used to estimate subjects' performances on TMT-B than they do when used to estimate performances on TMT-A. As the parameters measured in Chapter 4 are derived from observing how a subject uses a computer mouse, we argue that the simplest model that we can use to account for the observation is one in which subjects move the hand slower on TMT-B than on TMT-A. We argue that this effect is due to movements begin performed as a single-task during TMT-A, and as a dual-task during TMT-B. We proceed to revisit the measurements that we have made in Chapter 3 and show that they are consistent with a model in which the subject is single-tasking during movements in TMT-B.

My specific contributions in Chapter 5 are the modification of the computational model developed in Chapter 3 to include dual-tasking of the set-switching and motor stages, and the analysis of the data from Chapter 3 to show that data support a model in which set-switching and motor are dual-tasked during TMT-B.

15

Part I – In-Home Monitoring of Walking Speed

# Chapter 2 – Unobtrusive and Ubiquitous In-Home Monitoring: A Methodology for Continuous Assessment of Gait Velocity in Elders

Stuart Hagler, Daniel Austin, Tamara L. Hayes, Jeffrey Kaye, Misha Pavel

This work was originally published in IEEE Transactions in Biomedical Engineering 2010; Volume 57, Issue 4, Pages: 813-820. Reprinted with permission

### 2.0 Abstract

Gait velocity has been shown to quantitatively estimate risk of future hospitalization, has been shown to be a predictor of disability, and has been shown to slow prior to cognitive decline. In this paper, we describe a system for continuous and unobtrusive inhome assessment of gait velocity, a critical metric of function. This system is based on estimating walking speed from noisy time and location data collected by a "sensor line" of restricted view passive infrared (PIR) motion detectors. We demonstrate the validity of our system by comparing with measurements from the commercially available GAITRite® Walkway System gait mat. We present the data from 882 walks from 27 subjects walking at three different subject-paced speeds (encouraged to walk slowly, normal speed, or fast) in two directions through a sensor line. The experimental results show that the uncalibrated system accuracy (average error) of estimated velocity was

7.1cm/s (SD = 11.3cm/s), which improved to 1.1cm/s (SD = 9.1cm/s) after a simple calibration procedure. Based on the average measured walking speed of 102 cm/s our system had an average error of less than 7% without calibration and 1.1% with calibration.

#### **2.1 Introduction**

Improving quality of life and providing adequate medical care for the rising number of elderly while keeping health care costs under control has in recent years become a major problem [87]. Several methodologies have been proposed to address this problem that include the use of technology to develop systems that promote aging in place [52] and the use of pervasive healthcare [53] to help alleviate the burden placed on health care providers. One of the underlying themes of these approaches is to employ technology such as wireless networks combined with novel sensing systems to gather and interpret data in non-health care settings such as the home environment.

One specific measure of particular interest for unobtrusive assessment for health monitoring is walking speed. Walking speed has been shown to be a quantitative estimate of risk of future hospitalization [18]. Slower walking speed has been demonstrated in dementia patients compared to controls [19] and has been shown to precede cognitive impairment [20] and dementia [21], and timed walk has been used as a partial characterization of lower extremity function which has been shown to predict disability [22, 23]. Other studies have shown a relationship between walking speed and cognition [25, 26]. Current evaluation of walking speed is typically done both infrequently and in the clinic setting which suffers from at least five shortcomings. First, frequent assessment visits are impractical and cost prohibitive since either it is difficult

for patients to make frequent trips to a doctor's office or other clinical settings or in the case of research assessments, inconvenient for the research team to visit homes frequently. Second, for longitudinal study each testing session is typically scheduled in increments of six months or a year after a baseline visit making it difficult both to evaluate the validity or stability of baseline measurements and to detect short and long term variability [88]. Third, there may be an intentional [18] change in walking speed in the clinical setting or an unintentional [88] change in abilities during a single assessment. These pacing considerations themselves may have important implications for predicting outcomes [26]. Fourth, infrequent measurements report only the net change between measurement times and cannot distinguish between functional changes occurring slowly over time and abrupt functional changes, which may have different causes. Fifth, infrequent measurements do not detect changes when they happen which may reduce the ability of a clinician to provide intervention or reduce the effect of an intervention. By shifting to continuous in-home monitoring of walking speed from the current paradigm, the effect of all of these short comings can be significantly reduced or removed.

There have been several systems proposed for monitoring walking speed and other gait features outside of the clinical setting [89-91]. These systems typically consist of some wearable combination of gyroscopes and/or accelerometers and have demonstrated accuracy and precision in the field. However, these systems suffer from several limitations such as short battery life, the need to download the data or introduce additional hardware and complexity for wireless data collection, and the inconvenience of both a wearable device and having to remember to wear a device. For these reasons the wearable devices are currently inadequate for long-term, in-home, unobtrusive monitoring.

In order to address these concerns and to improve diagnostic ability for clinicians and researchers, we propose a methodology for continuous in-home monitoring of walking speed using passive infrared motion sensors. Specifically, we describe the hardware preparation and deployment, the techniques for data collection, and the data processing algorithms for continuous in-home assessment of walking speed in elders. Finally, we validate our approach by comparing the results of our method for walking speed estimation with the commercially available GAITRite® Walkway System gait mat.

## **2.2 System Description and Data Collection**

In this section, we describe the hardware and methodology used to deploy the walking speed measurement system in a residence. A partial description of this system has been described elsewhere [92, 93] in the more general context of total activity monitoring, as has a simpler version of the proposed approach [94]. Here we specialize and describe in more detail the specific nature of the walking speed measurement system. We begin by describing the sensors and how they are placed in a residence and follow with a description of the wireless network based data collection.

To detect motion we used the X10 model MS16A (X10.com) passive infrared motion sensor which emits a unique programmable bit code at 310 MHz when motion is detected. We restricted the field of view of each motion sensor to  $\pm 4$  degrees and installed four sensors sequentially on the ceiling (average height of 2.54 m) approximately 61cm apart in a confined area such as a hallway or other corridor. This combination tends to force a resident to walk linearly through each sensor pair in the sensor line and ensures that each sensor will only fire when someone passes directly

below. Limiting the field of view precisely and placing sensors in exact locations is not possible, and therefore there is some variability in the physical locations which cause the sensors to fire, as will be discussed shortly. Figure 2.1 shows how these sensors look from a resident's point of view when entering the sensor line.



**Figure 2.1.** A motion sensor line for measuring walking speed where the four sensors are placed 0.61 m apart and are installed on a ceiling typically 2.54 m high.

To collect the wirelessly transmitted sensor firings, we use a WGL 800 wireless transceiver connected to a desktop computer installed in the residence. Simultaneous sensor firings or other interfering sources can result in lost data due to collisions at the wireless transceiver. However, these have been shown to be minor, with a less than 2% overall data loss [93]. The computer timestamps the sensor firings and the data pair is both stored locally and sent via a secure Internet connection to a central database for analysis.

Our experience with the described technology comes from the deployment and monitoring of approximately 250 Portland (OR, USA) metropolitan area homes and retirement community dwellings from between 6 months to over 2 years in ongoing studies. We have instrumented both single and multi person dwellings and have collected data from over 1,200,000 walking events from single person homes with minimal technical challenge or sophistication needed for setup. Installation of the complete system (including additional technologies described elsewhere [93]) takes an average of 1.5 to 2 hours with 2 people. Deploying only the equipment necessary to measure walking speed (computer, sensors, wireless transceiver, and internet) is estimated to take 1 person approximately 1 hour - if the home already contains an Internet-connected computer, this could be done in 20 min. The technologies are managed remotely using custom systems management software that supports data viewing, remote software updates, and remote computer reboots if needed. Other issues, such as replacing motion sensor batteries (battery life is about 1 year) or changing sensors if they become unreliable or defective can typically be handled in a very short visit to the residence, typically 10-15 min. Overall, we have found the system has been simple to install, unobtrusive in the sense of both passive sensor technology and minimal outside intervention, and easy to maintain.

### **2.3 Data Modeling and Analysis**

In this section, we introduce and discuss both the proposed linear model and the estimator for determining the gait velocity from noisy motion sensor data. We start by describing how to determine the precise spatial separation of the sensors from the sensor firings since they will not, as mentioned, be the same as the measured values due to a combination of installation variability and differences between individual sensors. We then use this information to model the walking speed as a linear function of the measured

data degraded by two sources of additive noise. The first source of noise is in units of distance and is due to the sensor firing in slightly different locations during each pass through the sensor line. This error is based on the field of view and sensitivity of an individual sensor. The second source of error is in units of time and is due to the discrepancy between when a sensor fires and when the computer timestamps the firing, which generally causes positive time errors. We conclude the section by proposing a walking speed estimator that minimizes the combination of these errors followed by a discussion of model calibration in the presence of ground-truth data versus estimating the calibration factor when ground-truth data is unavailable.

## 2.3.1 The Linear Model

We start by assuming the sensors are placed at physical positions  $\{\tilde{x}_i\}$  in some spatial coordinate system. For a particular walking event the sensors fire at times  $\{t_i^k\}$  where k indexes the particular sensor line walking event and i indexes the particular sensor which fired. We then define  $\{x_i\}$  to be the average position at which the walker is when the *i*th sensor fires. In other words, for a particular walking event the *i*th sensor fires when the walker is at some random location  $\{x_i + \varepsilon_i\}$  with the errors  $\{\varepsilon_i\}$  being independent random variables with zero expected value. Figure 2.2 illustrates this arrangement.



**Figure 2.2.** Schematic of a person walking through a sensor line containing four sensors with the fields of view and the locations of the  $\tilde{x}_i$  and  $x_i$  shown.

The differences  $\{x_i - \tilde{x}_i\}$  represent likely biases due to the field of view of the sensor and the direction of movement as shown in Figure 2.2. For the sake of simplicity we restrict the present discussion to the analysis of movement in one direction.

Now assuming a walker moves with some known velocity v through a sensor line and we have some absolute reference time,  $\{\tau_i\}$  can be defined as the time at which the walker is expected to be at location  $\{x_i\}$  and trigger the *i*th sensor. If we now include the errors in detection location  $\{\varepsilon_i\}$  explicitly in the measured time we find that the measured times should be  $\{\tau_i + \varepsilon_i / v\}$ .

By further assuming that there is some random delay  $\{\eta_i\}$  between when a sensor fires at location  $\{x_i + \varepsilon_i\}$  and when the computer time-stamps the sensor firing data, the measured time can be written as  $\{t_i^k\} = \{\tau_i^k + \varepsilon_i^k / v^k + \eta_i^k\}$  where the  $\{\eta_i\}$  are independent random variables with some common non-negative expected value and the explicit dependence on the measured time from both position errors and time errors is shown.

Now, consider an event k to comprise a person walking past the line of sensors with a constant, but unknown velocity  $v^k$ . Then for any pair of sensors, i,j we have  $x_j - x_i = v^k \left(\tau_j^k - \tau_i^k\right)$ , from which we find:

$$\left(x_{j}+\varepsilon_{j}^{k}\right)-\left(x_{i}+\varepsilon_{i}^{k}\right)=v^{k}\left[\left(t_{j}^{k}-\eta_{j}^{k}\right)-\left(t_{i}^{k}-\eta_{i}^{k}\right)\right].$$
(2.1)

Using (2.1) we can solve for the velocity  $v^k$ , for any three sensors *i*, *j*, *m*:

$$v^{k} = \frac{(x_{j} - x_{i}) + (\varepsilon_{j}^{k} - \varepsilon_{i}^{k})}{(t_{j}^{k} - t_{i}^{k}) + (\eta_{i}^{k} - \eta_{j}^{k})} = \frac{(x_{m} - x_{j}) + (\varepsilon_{m}^{k} - \varepsilon_{j}^{k})}{(t_{m}^{k} - t_{j}^{k}) + (\eta_{j}^{k} - \eta_{m}^{k})}.$$
(2.2)

We now economize the notation and define  $\varepsilon_{ji}^k \equiv \varepsilon_j^k - \varepsilon_i^k$  and  $\eta_{ij}^k \equiv \eta_i^k - \eta_j^k$  for *i*, *j*, *m*. Rewriting (2.2) yields:

$$(x_{j} - x_{i})(t_{m}^{k} - t_{j}^{k}) + (x_{j} - x_{i})\eta_{jm}^{k} + (t_{m}^{k} - t_{j}^{k})\varepsilon_{ji}^{k} + \varepsilon_{ji}^{k}\eta_{jm}^{k} = (x_{m} - x_{j})(t_{j}^{k} - t_{i}^{k}) + (x_{m} - x_{j})\eta_{ij}^{k} + (t_{j}^{k} - t_{i}^{k})\varepsilon_{mj}^{k} + \varepsilon_{mj}^{k}\eta_{ij}^{k}.$$

$$(2.3)$$

Taking the expectation of both sides and using the facts that:  $E_k \left\{ \varepsilon_{ij}^k \right\} = 0$ ,  $E_k \left\{ t_i^k \varepsilon_j^k \right\} = 0$ , and  $E_k \left\{ \varepsilon_j^k \eta_i^k \right\} = 0$  for  $i \neq j$ , results in:

$$(x_j - x_i) E_k \{ t_m^k - t_j^k \} - E_k \{ t_j^k \varepsilon_j^k \} + E_k \{ \varepsilon_j^k \eta_j^k \}$$

$$= (x_m - x_j) E_k \{ t_j^k - t_i^k \} - E_k \{ t_j^k \varepsilon_j^k \} + E_k \{ \varepsilon_j^k \eta_j^k \},$$

$$(2.4)$$

which simplifies to:

$$\frac{\left(x_{j}-x_{i}\right)}{\left(x_{m}-x_{j}\right)} = \frac{E_{k}\left\{t_{j}^{k}-t_{i}^{k}\right\}}{E_{k}\left\{t_{m}^{k}-t_{j}^{k}\right\}}.$$
(2.5)

From this we may conclude that:

$$\left(x_{j}-x_{i}\right) \propto E_{k}\left\{t_{j}^{k}-t_{i}^{k}\right\} = E_{k}\left\{\Delta t_{ij}^{k}\right\}.$$
(2.6)

The expected value of the random variable  $\Delta t_{ij}^k$  can therefore be used to estimate the spatial separation of the sensors up to a scale factor by computing the average values over a large number of events.

By explicitly writing (2.6) with the proportionality constant c we have:

$$\left(x_{j}-x_{i}\right)=cE_{k}\left\{\Delta t_{ij}^{k}\right\}.$$
(2.7)

Here c has a ready interpretation as the speed a person would have to walk in order for the sensors to register time differences equal to the average time differences calculated from the training data set.

Let us look more closely at the estimated spacings  $(x_j - x_i)$ . Considering Figure 2.2 again, we see that the sensor-line is effectively hovering at some height between the ceiling and the floor. In the figure it has been drawn at the top of the head. Let us imagine that we knew the actual mean firing position for each sensor as a function of the height above the floor, so that if the sensor is triggered by motion at a height *h* it will typically be triggered at a position  $x_i(h)$ . In addition, the body itself does not move at a single constant velocity *v* during gait, but rather different segments move with various velocities over the course of the gait cycle. The effect of this is that the values  $\{\tau_i\}$  (which are the actual measurements) reflect triggering at various heights due to different body segments. In effect, all these factors are averaged over to produce effective sensor spacings based on the subject's height and style of walking together with the sensor characteristics. We expect that for people of similar heights, as well as reasonably

similar styles of gait, that the estimated effective sensor spacings  $(x_j - x_i)$  should be close in value.

## 2.3.2 Estimation of Gait Velocity

When we estimate the gait velocity we must consider two sources of measurement error. First, there is the combined error for sensors i, j resulting from detecting the walker at positions away from the mean detection locations  $x_i \, x_j$  which we denote by  $\varepsilon_{ii}$ . Second, there is the combined error for sensors *i*, *j* resulting from errors in timestamping the moments at which each sensor fired. This is represented by  $\eta_{ij}$ . In general, these two types of errors should be given different weights in accordance with the relative variability captured by the spatial and temporal covariance matrices of the error terms  $\varepsilon_{ii}$ and  $\eta_{ij}$  [95]. The relative weighting of the temporal error term relative to the spatial error term is represented by the parameter  $\rho$ . We proceed initially by assuming that the calibration factor c and the weighting factor  $\rho$  are known values and derive an estimator for the walking velocity,  $\hat{v}$  through the sensor line. With this estimator, we then proceed to consider situations in which we know the actual walking velocity, v and consider the estimator now as a function,  $\hat{v}(c,\rho)$ , which arises when one calibrates the line using a set of data where the velocities are known. Finally we consider the case where c,  $\rho$  are not known and use information from the physical set up of the sensors to estimate a value of c. In this last case we assume that weighting of the errors is a general value across all reasonably similar sensor lines and so take the value for  $\rho$  obtained from our calibrated experiment described in the next section.

In a sensor line of four sensors i, j, l, m:that fire sequentially when a subject walks along the sensor line, we can estimate the walking speed by minimizing the overall error in the dependent and independent variables using the method of total least squares [96] applied to the model:

$$\begin{pmatrix} \rho E_{k} \left\{ \Delta t_{ij}^{k} \right\} \\ \rho E_{k} \left\{ \Delta t_{jl}^{k} \right\} \\ \rho E_{k} \left\{ \Delta t_{lm}^{k} \right\} \end{pmatrix} + \begin{pmatrix} \frac{\rho}{c} \varepsilon_{ij}^{k} \\ \frac{\rho}{c} \varepsilon_{jl}^{k} \\ \frac{\rho}{c} \varepsilon_{lm}^{k} \end{pmatrix} \end{pmatrix} \begin{pmatrix} \frac{c}{\rho \hat{v}^{k}} \end{pmatrix} = \begin{bmatrix} \left( t_{j}^{k} - t_{i}^{k} \right) \\ \left( t_{l}^{k} - t_{j}^{k} \right) \\ \left( t_{m}^{k} - t_{l}^{k} \right) \end{bmatrix} + \begin{bmatrix} \eta_{ji}^{k} \\ \eta_{lj}^{k} \\ \eta_{ml}^{k} \end{bmatrix},$$
(2.8)

where we have rewritten the linear equations in matrix notation, used the fact that  $\eta_{ji}^{k} = -\eta_{ij}^{k}$  to keep the noise term in (2.8) in the standard additive form, and used the estimate of the spatial sensor separation as in (2.7). We proceed assuming that c,  $\rho$  are fixed and known constants for the sensor line.

To compute the velocity estimates, we now construct a matrix containing as column vectors the distances between adjacent sensors, and the time differences between adjacent sensor firings:

$$M^{k}(\rho) = \begin{bmatrix} \rho E_{k} \left\{ \Delta t_{ij}^{k} \right\} & \left( t_{j}^{k} - t_{i}^{k} \right) \\ \rho E_{k} \left\{ \Delta t_{jl}^{k} \right\} & \left( t_{i}^{k} - t_{j}^{k} \right) \\ \rho E_{k} \left\{ \Delta t_{lm}^{k} \right\} & \left( t_{m}^{k} - t_{i}^{k} \right) \end{bmatrix}.$$

$$(2.9)$$

This matrix may be factored per the singular value decomposition into a trio of matrices  $A^{k}(\rho)$ ,  $B^{k}(\rho)$ ,  $\Sigma^{k}(\rho)$  so that the equation  $M^{k}(\rho) = A^{k}(\rho)\Sigma^{k}(\rho)B^{k*}(\rho)$  is satisfied. Letting  $B_{_{12}}^{k}(\rho)$ ,  $B_{_{22}}^{k}(\rho)$  denote the appropriate elements of the matrix

 $B^{k}(\rho)$ , the estimated velocity is given by:

$$\hat{v}^{k} = -\frac{c}{\rho} \frac{B_{22}^{k}(\rho)}{B_{12}^{k}(\rho)}.$$
(2.10)

## 2.3.3 The Calibrated Sensor Line

Taking the weighting factor  $\rho$  still to be fixed and known, we now treat the calibration factor *c* as a variable whose value we may estimate using known velocity data. We are given a set of training data consisting of sensor firing times  $\{t_i^k\}$  and true gait velocities  $\{v^k\}$  for a sample set of walks through the sensor line. We assume the actual walking speeds and the estimated walking speeds satisfy the linear model:

$$v^{k} + \omega^{k} = -\frac{\hat{c}}{\rho} \frac{B_{22}^{k}(\rho)}{B_{12}^{k}(\rho)},$$
(2.11)

where the  $\{\omega^k\}$  are independent random variables with zero expected value. By collecting all the measurements  $v^k$ ,  $b^k(\rho) = B_{22}^k(\rho)/B_{12}^k(\rho)$  into vectors v,  $b(\rho)$ , we can estimate the calibration factor with linear least squares:

$$\hat{c} = -\rho \frac{v^T v}{v^T b(\rho)}.$$
(2.12)

If we knew the actual value  $\rho$  we would be done at this point as we could estimate the calibration factor *c* given the measured time and velocity values. However, where we do not know the actual value of  $\rho$ , we would like to choose a value which yields the best performance of the method for estimating velocities. In particular we would like to find a value of  $\rho$  which gives the best performance across many sensor-lines. Let us consider the calibration factor for the *n*th sensor-line as a function of  $\rho$ , that is:

$$\hat{c}_{n}(\rho) = -\rho \frac{v_{n}^{T} v_{n}}{v_{n}^{T} b_{n}(\rho)}.$$
(2.13)

We may consider the set of velocity estimates also as functions of  $\rho$ :

$$\hat{v}_{n}^{k}(\rho) = -\frac{\hat{c}_{n}(\rho)}{\rho} b_{n}^{k}(\rho) = \left(\frac{v_{n}^{T}v_{n}}{v_{n}^{T}b_{n}(\rho)}\right) b_{n}^{k}(\rho).$$

$$(2.14)$$

This expresses the estimated velocity for the *k*th walk along the *n*th sensor-line  $\hat{v}_n^k(\rho)$  entirely in terms of the measured time, the measured velocity, and the unknown parameter  $\rho$ . The weighting factor may now be found as a value which gives the best sensor-line performance on average.

In practical situations where calibration data are not available we use an average value of  $\rho$  determined from existing data. In particular, we found that the average value that minimized the estimation error in our controlled experiments, described in the next section, was  $\rho = 0.75$ .

#### 2.3.4 The Uncalibrated Sensor Line

If the velocity is not known, then the training data set will contain only the sensor firing times  $\{t_i^k\}$ . In this case we must estimate the value for *c* using the values for the physical sensor positions  $\{\tilde{x}_i\}$ . Choosing any pair of sensors *i*, *j* we may make the estimate:

$$\hat{c} = \frac{\tilde{x}_j - \tilde{x}_i}{E_k \left\{ \Delta t_{ij}^k \right\}}.$$
(2.15)

We would like to choose our pair of sensors so that the expected value  $\tilde{x}_j - \tilde{x}_i$  is as near in value as possible to the distance between mean detection positions  $x_j - x_i$ . In general the

expected error will be minimized by choosing the pair consisting of the outermost sensors of the line.

We do note that care should be taken when using the uncalibrated sensor line crosssectionally, especially over small samples, as the individual instantiations of a sensor line can have sizable differences between  $\hat{c}$  and c as shown in Figure 2.5.

## 2.4 Experimental Verification

#### 2.4.1 Experimental Description

27 subjects (9 male and 18 female, aged 75 to 95 years, mean age 85.2 years, 145cm to 185cm in height, average height 164.8 cm) participated in the experiment; all provided informed consent. The experiment was conducted in a common use room at the facility where the participants live. A single sensor line of eight (8) restricted-field PIR motion sensors was placed on the ceiling with sensors physically spaced at 61cm (2ft) intervals. The ceiling height was 240cm (7.8ft). Beneath the sensor line an 854cm (14ft) long GAITRite<sup>®</sup> Walkway System gait mat was placed so that the ends of the mat aligned with the outermost sensors. Participants were instructed to walk at self-determined "slow", "normal", and "fast" walking speeds. A total of 30 walks were recorded for most participants such that each participant walked five times at each of the three speeds in the two directions available along the sensor line. Five participants did a larger number of walks (36, 42, 42, 44, and 46) but their larger group of walks included the basic 30 which all participants did. Their precise walking speed for each trial was calculated using the gait mat data. Firing times were collected for each PIR sensor during each trial and used to determine the accuracy of the PIR sensors for measuring walking speed.

Based on our experience with several hundred homes, a reasonable sensor line

configuration which may be installed under the space constraints of typical small homes consists of four sensors placed in a line at approximately 61cm (2ft) intervals. The choice of four sensors was influenced by a few different factors. First, since the sensors are known to be noisy, we wanted multiple measurements of the walking speed to use in our estimator to reduce estimator variance. Experiments showed, for example, that moving from three sensors to four sensors reduced variance by a factor of approximately 3.8. Second, due to space constraints in the homes and retirement communities we found that four sensors are all that would reliably fit in most homes. Third, the probability of an individual sensor firing is approximately 0.937. With four sensors in place and assuming we use walks where either three or four sensors fire, we can capture almost 98% of walking events in the home. Finally, we note that using two sensors in not sufficient as this causes equation (2.8) to reduce to a single equation with a single unknown which can be solved exactly, and therefore does not allow mitigation for known noise effects. In accordance with this we have considered sensor data in groups of four adjacent sensors. Thus our line of eight sensors is treated as five individual sensor lines. Furthermore as there is no reason to suppose that the effective sensor spacing is the same in the two directions along which the line may be walked, – in the "forward" or "return" direction through the line (with respect to the experimenter) - we evaluated each direction independently.

For each sensor line of four sensors in each direction only those walking events in which all 4 sensors fired were considered for the purposes of calculating the effective sensor spacing and calibration factor. However, to estimate the velocity for walking events we used all the sensor line data in which 3 or 4 of the 4 sensors fired.

## 2.4.2 Experimental Results

A total of 882 walks from the 27 participants (mean age 85.2 years) were recorded during this experiment with 441 in the "forward" direction and 441 in the "return" direction (as referenced by the experimenter). The numbers of "slow", "normal", and "fast" speed walks were the same in either direction for each given participant. The 8 ceiling-mounted PIR sensors were divided into 5 sensor lines of 4 sensors with a regular 61cm (2ft) physical spacing for analysis. In the "forward" direction the sensor lines had all four sensors fire  $350 \pm 36$  times, and in the "return" direction all 4 sensors fired  $330 \pm$ 29 times. The effective sensor spacing for each sensor line was calculated and normalized as in the foregoing using only the events where all 4 sensors fired.

Participants walked in the "forward" direction with a speed of  $104 \pm 30.6$  cm/s, and in the "return" direction with a speed of  $100 \pm 29.3$  cm/s as measured by the gait mat. We estimated velocity using a sensor line only in those cases where 3 or 4 of the 4 sensors in the line fired. Figure 2.3 shows the directional walking speed estimates versus the measured values for the combined sensor line data after calibration using velocity data from the gait mat. Figure 2.4 shows the directional walking speed estimates using estimated calibration factors. In both figures the "return" direction is differentiated from the  $v^k$  "forward direction" by introducing a negative sign on all the velocity estimates. Additionally, in both figures the line x = y (corresponding to perfect estimation) has been plotted as a dashed line to demonstrate that both calibrated and uncalibrated estimates are distributed around the correct values. Further, the distribution of points in Figure 2.4 is wider than in Figure 2.3 demonstrating that the calibration procedure does improve estimation. Also, of particular note is the fact that distribution of estimates in both

figures is more centered and densely packed around the true value in the "return" direction, indicating that velocity estimates in the "return" direction are better than in the "forward" direction (i.e., the same sensors performed better in one direction than in the other).



**Figure 2.3.** Combined walking speed data for all subjects for the 5 sensor lines of 4 sensors (the various shapes indicate data from different sensor lines), using a calibration factor calibrated to the walking speed measured by the gait mat. The sign of the estimated speed differentiates "forward" walks (positive values) versus "return" walks (negative values).

To be more precise about the discussion of the walking speed estimates, we denote the estimated speed for the *k*th walk through sensor line *i* for both directions by  $\hat{v}_i^k$  and the actual speed by . The accuracy of the system was evaluated by computing the average difference  $\{\hat{v}_i^k - v^k\}$ . For the case of the calibrated sensor lines the mean of this difference is 1.1cm/s and the standard deviation is 9.1cm/s. In the case of the uncalibrated sensor lines the mean is 7.1cm/s, and the standard deviation is 11.3cm/s.



**Figure 2.4.** Combined walking speed data for all subjects for the 5 sensor lines of 4 sensors (the various shapes indicate data from different sensor lines), using estimated calibration factors. The sign of the estimated speed differentiates "forward" walks (positive values) versus "return" walks (negative values).

Figure 2.5 shows the relationship of the estimated calibration factors to the true (measured) calibration factors, with the line x = y drawn for comparison. This shows that the uncalibrated sensor line with  $\hat{c}$  as in (2.15) tends to underestimate the true calibration factor, this making the velocity estimates slightly higher than in the calibrated case. This also demonstrates the need to be careful when comparing uncalibrated sensor line estimates cross-sectionally.



**Figure 2.5.** Combined calibration factor data (c value) for the 10 possible sensor lines (5 "forward" and 5 "return"). The direction the triangle is pointing indicates whether the direction is "forward" (up) or "return" (down).

#### **2.5 Discussion**

The proposed system for unobtrusive and continuous monitoring of in home walking speeds has been shown to accurately estimate velocity when compared to the GAITRite® Walkway System gait mat standard. The mean estimation errors of 7.1cm/s and 1.1cm/s for the uncalibrated and calibrated sensor lines when compared to the average speed of 102cm/s result in average errors of 6.96% and 1.08%, respectively. Further, the standard deviations of the error distributions for the uncalibrated and calibrated sensor lines are 11.1% and 8.92% when compared to the average speed of walking. This shows that each individual estimate is accurate, and local averaging and other statistical techniques can be used to increase precision (reduce the error variance further).

These positive results demonstrate the feasibility of the proposed method and address several deficits in the current paradigm of assessing gait episodically or in clinic settings. First, with this system the variability of walking speed can now be monitored continuously over the short term (e.g., walk-to-walk variability) in addition to longer time scales (e.g., month-to-month, yearly) without expensive and inconvenient clinic visits. Second, subjects in high-risk groups can be monitored more closely and rapidly than is currently feasible. Third, researchers can have access to more frequent measurements of walking speed, which facilitates the refinement and better understanding of walking speed as it relates to health outcomes and correlations presently in the literature that are based on single or infrequent measurements. Fourth, wide scale analysis of multiple subjects can be performed relatively easily which we anticipate will open further areas of population-based research and diagnostic ability not discussed here. We do note that while our proposed system is less expensive than repeated clinical visits, the cost of the sensors, computer, internet service, and transceiver may currently be cost prohibitive for studies that might involve thousands of subjects who are widely dispersed. However, since historically the cost of equipment and services has continued to drop as better and faster technology becomes available in the marketplace, it is likely that deployment of these kinds of systems to larger cohorts will be facilitated. In addition, less expensive motion sensors may work adequately and simple application specific computers may be built cheaper than off the shelf models which could be deployed today. Further work is needed to identify the most cost efficient approaches to maximize scalability of in-home assessment platforms.

One of the largest challenges to the broad use of our approach to continuous monitoring of walking speed, and to in-home monitoring in general, is the differentiation of multiple residents. This problem is typically addressed by requiring the participants to wear or care some type of radio frequency identification (RFID) tag. We are currently working on both pattern recognition and model-based approaches to distinguish between multiple residents based on the walking events. This will allow the expansion of this methodology from single resident homes to multiple resident dwellings without the need for additional equipment or hardware.

Future work will address comparisons of the in-home continuous method with standard clinical tests of walking, mobility, and physical performance such as the Short Physical Performance Battery, Unified Parkinson's Disease Rating Scale, and various other timed walks of different durations (e.g., 4-meter, 10-meter, 400 meter) thus facilitating interpretation of these established clinical metrics with our new framework. Other future work will include relaxing the assumption that velocity is fixed over a walking event in order to measure the step-to-step variability in each walking event. In this case the velocity of the *k*th walking event becomes some function of time v(t). Retaining our definition of the error  $\{\varepsilon_i\}$  above we find that the time at which the sensor fires may be expressed as  $\{\tau_i + \upsilon_i\}$ , where  $\{\upsilon_i\}$  satisfies:

$$\varepsilon_i = \int_{\tau_i}^{\tau_i + v_i} v(t') dt'.$$
(2.16)

The values  $\{\eta_i\}$  are still defined as above which gives a relation to the measured time values of  $\{t_i^k\} = \{\tau_i^k + \upsilon_i^k + \eta_i^k\}$ . Finally, for any pair of sensors we have the relation:

$$\left(x_{j}+\varepsilon_{j}^{k}\right)-\left(x_{i}+\varepsilon_{i}^{k}\right)=\int_{t_{i}^{k}-\eta_{i}^{k}}^{t_{j}^{k}-\eta_{j}^{k}}v^{k}\left(t'\right)dt'.$$
(2.17)

which generalizes (2.1). We anticipate that adjusting the model along the lines of (2.16) will allow us to derive additional gait parameters from the current and future data.

## **2.6** Conclusion

In this paper we have proposed a new system for continuous in home assessment of walking speed based on PIR sensors and a wireless network for data collection. We have shown that this method is both accurate and precise when compared to the standard of the GAITRite® Walkway System gait mat. This method allows the convenient in home collection of a large number of walking events otherwise gathered infrequently in a clinical setting. Since walking speed has been shown to be an indicator or predictor of many diseases and other health issues such as cognitive decline and hospitalization, we feel that the continuous monitoring of this measure and its applications is an important and useful area of future research.

Part II – In-Home Monitoring of Trail-Making Test Performance

# Chapter 3 – Assessing Executive Function Using a Computer Game: Computational Modeling of Cognitive Processes

Stuart Hagler, Holly B. Jimison, Misha Pavel

This work was originally published in IEEE Journal of Biomedical and Health Informatics 2014; Volume 18, Issue 4, Pages: 1442-1452. Reprinted with permission

### **3.0 Abstract**

Early and reliable detection of cognitive decline is one of the most important challenges of current healthcare. In this project we developed an approach whereby a frequently played computer game can be used to assess a variety of cognitive processes and estimate the results of the pen-and-paper Trail-Making Test (TMT) – known to measure executive function, as well as visual pattern recognition, speed of processing, working memory, and set-switching ability. We developed a computational model of the TMT based on a decomposition of the test into several independent processes, each characterized by a set of parameters that can be estimated from play of a computer game designed to resemble the TMT. An empirical evaluation of the model suggests that it is possible to use the game data to estimate the parameters of the underlying cognitive processes and using the values of the parameters to estimate the TMT performance.

Cognitive measures and trends in these measures can be used to identify individuals for further assessment, to provide a mechanism for improving the early detection of neurological problems, and to provide feedback and monitoring for cognitive interventions in the home.

#### **3.1 Introduction**

Quantitative assessment of cognitive function is an important component of caring for the aging as well as those with other dysfunctions such as traumatic brain injury and many other conditions affecting cognitive functions. The goal of this study is to find ways to assess and monitor subjects' cognitive performance in the subjects' home using information technology. In this paper, we show how a simple computer game in conjunction with computational model can be used for sensitive assessment and monitoring of components of executive function in individual subjects.

The computer game we consider in this paper bears a close relationship to a commonly administered, neuropsychological test – the (pen-and-paper) Trail-Making Test (TMT). Typically administered as one test in a larger battery of tests, TMT is made up of two parts – TMT-A and TMT-B – each resembling a child's connect-the-dots puzzle. Each part, as with the puzzle, is completed by drawing a single, continuous line through all the "dots" in a specified order. The subject's score on each part of TMT is the time the subject took to draw the line to the last "dot." TMT is known to measure visuo-perceptual ability, working memory, and set-switching ability. [34, 35]

Computer-based implementations of neuropsychological tests, such as TMT, have potentially many advantages over traditional, pen-and-paper implementations, including: (1) uniformity of administration across subjects, and (2) more consistent scoring of

42

performance. They also allow the possibility of decomposing performance on the test into performance on individual parts of the test. Researchers have examined the use of computer-based neuropsychological testing [97-99] and have found them to be promising for the cognitive assessment of older adults. [98, 99] In particular, computerized implementations of TMT have been developed (e.g., [100, 101]), however differences between performance on a computerized implementation of TMT and the standard penand-paper TMT, as measured by the scores on TMT-A and TMT-B, have been shown. [101] An alternative to simply implementing a computer-based TMT is to have the subject perform the pen-and-paper TMT while the test administrator notes the duration of the subject's moves the pen to each "dot" by selecting a button on a computer GUI each time a "dot" is selected. [102] This approach allows the performance on TMT to be decomposed into performance on each movement to a "dot."

Our approach is to focus on the time taken to make each move to each "dot" rather than on the time taken to draw the line through the whole set of "dots." Given the information about subject performance gained by examining all the individual moves to "dots" we can then estimate the time the subject would need to draw a line through a set of dots – such as those given on TMT. To obtain sufficiently accurate estimates of the underlying processes requires data for a large number of moves. To acquire the needed move data, we have constructed a simple computer game in which the subject completes a series of rounds each of which consists of a set of randomly placed "dots" which the subject connects by using a computer mouse to select the "dots" in a specified sequence.

We develop a model for each move to a "dot" assuming a sequence of the three independent processes based on Donders' additive stages: [103, 104] recall, search, and

43

motor. The motor stage, describing the movement of the pen or mouse from one "dot" to the next, is based on Fitts' law, characterizing rapid movements into specified target regions. [105-107]

## 3.2 Overview of Two Connect-the-Dots Tasks

TMT is a pen-and-paper neuropsychological test that measures a subject's visuoperceptual ability (ability to interpret visual information), working memory (ability to hold items in memory while completing a complex task), and set-switching ability (ability transition from a task involving one class of objects to a task involving a different class). [34, 35] Scavenger Hunt (SH) is a point-and-click, mouse-driven computer game with game mechanics designed to mimic the testing mechanics of TMT and yet be both challenging and fun so that people are willing to play it routinely over time in a home environment. Both TMT and SH are built around the *connect-the-dots task* that forms the basis of the child's puzzle giving the task its name. In this task, the subject must select a number of "dots" in sequence by drawing a line through them. We used both TMT and SH to validate our model of the connect-the-dots task given in Section 3.3. The following sections provide an overview of TMT and SH.

We call the interval from the selection of one "dot" to the selection of the next "dot" a *move*.

#### 3.2.1 Trail-Making Test

The pen-and-paper TMT is comprised of two separate tests: TMT-A and TMT-B. TMT-A and TMT-B are printed on a standard 8.5"x11" sheet of paper with 25 small (12mm diameter) circles, the *targets*, placed in a seemingly random pattern on the sheet. All targets in both tests have the same diameter or *width*, and contain a *label* which may a letter or a number. In TMT-A, a label is a number from 1 to 25, while in TMT-B a label is a letter from A to L or a number from 1 to 13. Labels only appear once on the test page. In addition, two targets on each test page are indicated by the presence of the words "Begin" and "End" near to (but outside of) these targets. TMT-A and TMT-B refer to test pages each with a specific arrangement of targets, and the same test pages are used every time TMT is administered. Figure 3.1 shows a portion of TMT-A.



**Figure 3.1.** A section of the TMT-A neuropsychological test. Note the words "Begin" and "End" indicating the locations of the first and last targets.

Prior to beginning TMT-A or TMT-B, the test administrator instructs the subject on how to correctly complete the test. This is done by walking the subject through a shorter -8 target – sample test.

TMT-A and TMT-B each start with the subject being given the test page face down, the subject not having seen the test page. The test begins when the test administrator instructs the subject to start the test, and the subject turns over the test page and begins. The subject completes each test by drawing a single line, the *trail*, through all 25 targets in the specified order. In TMT-A, the targets are selected in ascending numerical order of the target labels (i.e. '1,2,3,...,24,25'), while in TMT-B, the order is ascending alphanumeric order of the target labels (i.e. '1,A,2,B,...,L,13'). The "Begin" is printed on the test page lies near the first target of the sequence and the "End" near the last.

The subject's performance on TMT is given in the form of a *score* on each of TMT-A and TMT-B, that is, the time taken to correctly draw a line through all targets on the test page in the specified order beginning when the test administrator instructs the subject to begin, and ending when the test administrator notes that the subject has reached the last target.

The test administrator also makes sure that the subject completes the test correctly, interrupting the test whenever the subject is observed to make an incorrect target selection (i.e. selecting a target out of sequence), as soon as the test administrator notices the error. Whenever such an *error* occurs, the test administrator instructs the subject to return to the last correctly selected target. Timing is not suspended during this process, and is included in the total time and thus in the test score, although, the number of errors for TMT-A and TMT-B are recorded by the test administrator. We call the sequence just outlined, the process of *recovering* from the error.

Guidelines for the administration of TMT are provided in [29]. TMT is normally administered in an office setting by a neuropsychologist once every 6 to 12 months to

46

reduce practice effects associated with repeated completion of standardized tests. [30-32] Normative data for subjects with education in the range of the subject used in our study show TMT scores for ages 75-79 of 42 s on TMT-A and 100 s on TMT-B, and for ages 80-84 of 55 s on TMT-A and 130 s on TMT-B. [27] TMT is one of the most clinically useful neuropsychological tests and is routinely used in the diagnosis of many neurological conditions (Parkinson's, Alzheimer's and dementias, and general cognitive decline). [33] However, it is an expensive test and the infrequent assessment leads to delays in the detection of cognitive issues.

## 3.2.2 Scavenger Hunt

The point-and-click, mouse-driven SH computer game is intended to mimic the mechanics of TMT while presenting the subject with arbitrary target configurations rather than the two fixed target configurations present on TMT-A and TMT-B test pages. SH was designed to be more engaging and fun, and yet present cognitive tasks that would test cognitive functions similar to those of TMT. We have been able to demonstrate that older adults are able to learn the game and play it routinely on computers in their homes. [67]

The subject plays SH by completing a series of *rounds* each being a single connect-thedots task. SH rounds must be completed in 30 s (imposing a speed-accuracy tradeoff); if the subject fails to do so the round is lost and the game of SH ends. SH play continues from round to round until either a round is lost, or the subject elects to stop playing.

Figure 3.2 shows a typical SH round. We call the large pane within the GUI containing the words "Scavenger Hunt" the *board*. The upper left hand corner of the board shows the amount of time left in the round. The upper right hand corner of the board shows the

cumulative *score* for all the rounds completed so far. The box in the center of the upper center of the board contains the *search string* 'B,2,A,1'. The remainder of the board shows the array of *markers* for this round. The set of markers includes both *targets*: '1', '2', 'A', and 'B', and additional *distractors*: '5', 'C', and 'L'. The *number* of markers on the board for the round shown is 7. The subject would *play* this round of SH by using the computer mouse to connect-the-dots by *selecting* targets the targets 'B', '2', 'A', '1', in that order, by clicking on them. The *trail* made by the subject in SH is path taken by the mouse in the course of selecting the targets.



**Figure 3.2.** A typical Scavenger Hunt round. The board for a round of Scavenger Hunt showing the time remaining in the game (27.2 s), search string ('B2A1'), cumulative game score (3269), targets ('1','2','A','B'), and distractors ('5','C','L').

SH indicates correctly selected markers by coloring them green for the remainder of the round; no line indicating a trail is drawn. A subject makes an error when playing a round of SH by selecting any marker other than the one currently being looked for (i.e. the lowest unselected marker in the search string). SH indicates that an error has occurred by

drawing a red "X" over the selected marker which remains until another marker is selected.

SH displays the subject's cumulative score on the game board for the rounds that have been completed. This score is used as feedback and motivation for the subject and not used to infer cognitive function or predict TMT scores. In this paper, we only refer to TMT-A and TMT-B test scores.

Search strings in SH may be ascending or descending alphabetical sequences (e.g., 'A,B,C,...' or '...,C,B,A'), ascending or descending numeric sequences (e.g., '1,2,3,...' or '...,3,2,1'), ascending or descending alphanumeric sequences (e.g., '1,A,2,B,...' or '...,B,2,A,1'), and English language words selected out of a fixed lexicon (e.g., 'H,O,R,S,E').

A marker in SH appears as a circle containing a single letter or number. The centers of the markers are arranged on the board in a 4x8 grid with a spacing of 80 pixels. We refer to the position of all the markers on the board in this grid as the *layout* of the markers. Markers may have a diameter of 63 pixels or 77 pixels. At the normal viewing distance of about 25 cm, the markers subtend approximately 3 degrees of visual angle. This size assures 100% recognition for subjects with corrected vision to 20/20. In any round, all the markers have the same *width* (diameter). SH has two types of markers: (1) targets, and (2) distractors. *Targets* are those markers containing a character that appears in the search string and that the subject must select in the course of completing the connect-the-dots task for the round. *Distractors* are markers that contain characters not appearing in the search string.

A SH round has a variable number of targets, typically about 4 to 10, as well as additional distractors. The board for each SH round is generated at random. In order to track accurately the subject's performance over time, SH has a particular type of test pattern that appears regularly and often to serve as a baseline reference on a subject's performance over time; these rounds have the search string '1,2,3,4', and no distractors. These rounds make up about one in four SH rounds.

SH was designed to try to make the repetition of a very simple task as interesting to the subject as possible. The use of a smaller number of targets was believed to make the game faster, and the variability of the number of targets together with additional distractors was believed to add more variety to the game. Ascending numeric and alphanumeric sequences were included to facilitate comparison to TMT, and descending numeric and alphanumeric sequences as well as ascending and descending alphabetic sequences and lexical sequences were included to add further variety to the game.

#### *3.2.3 Differences Between SH and TMT*

While SH was designed to mimic TMT, the two tasks are clearly not identical, with differences including: (1) SH is played in-home at the subject's leisure, while TMT is an in-clinic test, (2) SH is a computer game while TMT is a pen-and-paper test, (3) a SH round has a 30 s time limit while the subject is instructed to complete TMT-A or TMT-B as quickly as possible, (4) in SH the search string remains visible to the subject for the duration of the round while in TMT the subject is told the search string verbally before beginning the test, (5) the presence of the words "Begin" and "End" on TMT, (6) that SH marks a selected target by changing the color to green where TMT marks a selected target by trail passing through it, (7) SH boards can contain both targets and distractors
while TMT contains only targets, and (8) SH board typically contain about 4-10 markers while TMT always contains 25. In Section 3.3.1 - 3.3.4, we develop a model of the connect-the-dots task that is intended to produce a set of performance measures for each subject based on analysis of moves made playing SH. It is expected that these performance measures would be related to comparable measures based on analysis of moves made in TMT (were the move data available). However, comparison of the resulting performance measures between SH and TMT is complicated by the differences listed here. Instead of assuming that the performance measures are the same in the two cases, we suppose some set of transformations exist relating each performance between the cases of SH and TMT, and, for simplicity, that these transformations are the approximately the same for all subjects. These transformations are developed in detail in Section 3.3.5, and given in Eq. (3.4).

## **3.3 Connect-the-Dots Model**

The connect-the-dots task is completed by drawing a single line through a sequence of "dots" in a specified order. On a high level, performance on the connect-the-dots task can be characterized by the time taken to complete the entire task and the number of errors made in the course of completing the task (as is done in TMT); however, we choose to characterize subject performance by characterizing the performance across each individual move the subject makes in the course of completing the task.

The particular goal in the present paper is to show that measuring subject performance on moves observed in SH play can be used to estimate the subject's scores on TMT. The connect-the-dots model we develop in this section describes each move in the connectthe-dots task. By applying the model to the SH we can take all the observed moves from SH rounds and estimate a set of parameters characterizing how a subject makes a move in connect-the-dots tasks like those in SH. Conversely, given a set of parameters characterizing how a subject makes a move during TMT, we can construct an estimator of the TMT score in the case where no errors are made.

For the sake of simplicity, the proposed model does not characterize errors or the process of recovering from errors; consequently, in our analysis, we ignore the small proportion of rounds of SH in which the subject made any error. We ignore whole rounds to avoid any affects of the error on other moves, whether the error was due to something happening during an earlier move, or caused the subject to carry out subsequent moves differently than they otherwise would. Unfortunately, subjects do make errors on TMT, and we have to account for those errors in the estimators of the TMT-A and TMT-B scores (see Section 3.4 for information on the numbers of errors made). We account for the observed errors by estimating the time delays due to the errors and including these in the prediction of the TMT-A and TMT-B scores. This approach is useful in estimating the relative contribution of the correctly executed moves and errors in the final test scores.

The connect-the-dots model decomposes a move into a sequence of three statistically independent stages as shown in Figure 3.3: (1) the *recall and update* stage during which the subject calls to mind the next target in the search string, (2) the *search* stage during which the subject searches among the unselected targets game board to locate the current target, and (3) the *motor* stage during which the subject moves the mouse or pen to the located target to select it. The general methodology corresponds to Donders' additive stages. [103, 104]



**Figure 3.3.** Additive stages move model. The process of selecting the next target in the sequence involves three sequential stages of (1) recalling the next target, (2) serially searching for the next target by considering the available targets one after another, and (3) physically moving the mouse so that the cursor is on the target and clicking.

The statistical independence is based on the idea that each stage is affected by different aspects of the task and that the effect is limited to that stage. We expect that the duration of the recall and update stage would vary with the type of the search string (i.e. it should take a different amount of time to recall the next target when the search string is purely alphabetic or numeric as opposed to an alphanumeric search string). The duration of the search stage should depend on the number of additional distractors and unselected targets on the board, with the time spent in search decreasing on average as the subject moves to the end of the round. [108] Finally, the length of the motor stage should depend only on the distance on the board from the previously selected target to the new target – assuming that the target size is constant.

We now describe the detailed characterization of the stages of the model: recall and update, search and motor.

## 3.3.1 Recall and Update Stage

The recall and update stage is characterized by the *recall time*  $T_R$  required by the subject to recall the next target in the search string. The recall time is a random variable (RV) with expected value  $\langle T_R \rangle = \tau_R$ , and some standard deviation. We suppose that the time  $T_R$  spent by the subject recalling the next target in a sequence may vary across the classes of search strings available in SH (i.e. alphabetic, numeric, alphanumeric, and lexical), but is assumed to be the same for all the targets in sequences of a given class. We denote values for  $\tau_R$  intended to estimate recall for TMT-A-like connect-the-dots task by  $\tau_R^A$ , and for TMT-B-like tasks by  $\tau_R^B$ .

# 3.3.2 Search Stage

The search stage is characterized by the *search time*  $T_s$  required by the subject to locate the next target in the search string after it has been recalled. We treat search as a series of discrete steps, [108] where the total number of steps in any search is a RV. In each step, the subject considers a different marker (a target or a distractor) on the board. The subject compares the marker being considered during that step to the target being searched for. If the subject decides that the marker being considered is the same as the target they are searching for, they select the marker; otherwise the subject continues the search to another step and considers another marker. Each step of search takes some fixed time  $\tau_s$ . We suppose that the time  $\tau_s$  spent by the subject on each step of search may vary across the classes of search strings available in SH (i.e. alphabetic, numeric, alphanumeric, and lexical), but is assumed to be the same for all the targets in sequences of a given class. We denote values for  $\tau_s$  intended to estimate search step time for TMT-A-like connect-the-dots task by  $\tau_s^A$ , and for TMT-B-like tasks by  $\tau_s^B$ .

The number of steps in a given search depends on the number of markers remaining on the board (i.e. the initial number of markers less the number of targets that have been selected thus far). We suppose that the subject searches the remaining markers at random, with no memory of any of the remaining markers from searches made in previous moves in the same round; we further suppose that during the search stage, the subject has perfect memory and considers each marker only once (we discuss the validity and utility of these assumptions in Section 3.7). Let us consider a SH round with *n* targets and *d* distractors. Suppose the subject is searching for the v-th target. The subject has already found v-1 targets, so there are n-v+1 targets still on the board. The expected number of steps for the search is (n-v+d+1)/2. The expected value for the total search time  $T_s$  for this target is given by:

$$\langle T_s \rangle = \left( \left( n - \nu + d + 1 \right) / 2 \right) \tau_s. \tag{3.1}$$

The distribution of  $T_s$  for a given value  $n-\nu+d$  is uniform on the discrete values  $\tau_s$ , ...,  $(n-\nu+d+1)\tau_s$ .

# 3.3.3 Motor Stage

The motor stage is characterized by the *motor time*  $T_M$ , required by the subject move the mouse or pen from one target to the next. We suppose it to be independent of the search string. The movement made by the mouse or pen is a rapid movement into a target area given by the size of the marker being selected, and is expected to be consistent with Fitts' law. [105-107] We treat the motor time as a RV whose mean value satisfies Fitts' law and has some standard deviation. Fitts' law expresses the relationship between the distance D from the initial position to the center of the target, the target width W, and the expected motor time,  $T_M$ , required to complete the move. Defining  $D_v$  to be center-to-center distance between the v-1-th and v-th targets, assuming a common width W for all targets, the expected motor time taken to move from the v-1-th to the v-th target is given by:

$$\langle T_M \rangle = a + b \log_2 \left( D_{\nu} / W + 1 \right). \tag{3.2}$$

The value  $\log_2(D/W+1)$  provides a measurement of the amount of information the subject must process to complete the movement as measured in bits; so the value *b* provides a measure of how much time the subject spends processing each bit of information.

#### 3.3.4 Total Time

The expected time needed to complete an error-free connect-the-dots task with n targets and d distractors is simply the sum of the expected times for all of the component moves (we use the dot to distinguish multiplication  $a \cdot (b)$  from the expression of a function a(b)):

$$T = n \cdot (\tau_{R} + a) + \kappa (n, d) \tau_{s} + \chi (D_{1}, ..., D_{\nu}, W) b,$$
  

$$\kappa (n, d) = \sum_{\nu=1}^{n} ((n - \nu + d + 1)/2),$$
  

$$\chi (D_{1}, ..., D_{\nu}, W) = \sum_{\nu=1}^{n} \log_{2} (D_{\nu}/W + 1).$$
(3.3)

We can see that the expected time required to complete a connect-the-dots task is linear in the parameters characterizing the subject's cognitive and motor abilities:  $\tau_R$ ,  $\tau_S$ , a, and b. We call  $\kappa(n,d)$  and  $\chi(\{D_v\},W)$  the search complexity, and the motor complexity respectively.

Due to the way in which  $\tau_R$  and *a* appear in Eq. (3.3), their values cannot not be estimated separately. Instead, the best we can do is to estimate their sum  $\tau_R + a$ .

#### 3.3.5 Relating SH to TMT

Due to the differences between SH and TMT outline in Section 3.2.3, we do not expect the values  $\tau_{R}$ ,  $\tau_{S}$ , *a*, and *b* to relate trivially to their counterparts  $\tau_{R}'$ ,  $\tau_{S}'$ , *a'*, and *b'*. Differences 7 and 8 from Section 3.2.3, the presence of distractors and the numbers of targets, have already been handled in the model developed in this section. We suppose that the differences between SH and TMT do not affect the search or motor stages, so difference 6 regarding how selected targets are indicated is assumed not to affect search, and difference 2 regarding SH being a computer game and TMT being a pen-and-paper test is assumed not to effect movement from one target to the next. The remaining differences -1, in-home versus in-clinic, 2, presence of time limit, 4, visibility of the search string, and 5, the presence of the words "Begin" and "End" - are assumed to only affect recall. We model the net effect of these differences on recall using the linear transformation  $\tau_{R}' = \alpha + \beta \tau_{R}$ . However, as a practical matter, we cannot separate the values  $\tau_{R}$  and a, but rather we must work with  $\tau_{R} + a$ , so we use the approximate transformation  $\tau_{R}' + a' \approx \alpha + \beta \cdot (\tau_{R} + a)$ . The full set of transformations is:

$$\tau_{R}' + a' \approx \alpha + \beta \cdot (\tau_{R} + a),$$
  

$$\tau_{S}' = \tau_{S},$$
  

$$b' = b.$$
(3.4)

This set of transformations is assumed to be the same for all search strings (i.e. the values  $\alpha$  and  $\beta$  are the same when relating SH to TMT for TMT-A-like search string and for TMT-B-like search strings). We also assume that our subject population is sufficiently homogeneous that we can use the same transformation for every subject.

# **3.4 Analyzing SH Play**

We validated the model of connect-the-dots tasks developed in Section 3.3 by constructing an estimator of the subject's scores on the TMT-A and TMT-B using measurements taken from that subject's play of SH. We used SH data from rounds using ascending and descending alphabetic and numeric search string to construct the TMT-A estimator and data from rounds using ascending and descending alphanumeric search string to construct the TMT-B estimator. We chose to combine round data in this way so that more data would be available for each subject and we would be able to retain as many subjects as possible for analysis (see Section 3.4 for more information).

We now consider how to estimate a subject's cognitive and motor parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and *b* using the SH move data which consists only of timestamps indicating when buttons were selected by the subject and the relative positions of the buttons on the board. We produce the estimates in a two-step process. In the first step, we estimate the subject's motor performance as described by the Fitts' law *b* parameter, from the time and position data, using SH rounds with search string '1,2,3,4' and no distractors using the model developed in Section 3.3. In the second step, we use the estimated motor performance from the first step to remove the effect of motor performance from observed moves in SH rounds with alphabetic, numeric, and alphanumeric search strings (i.e. rounds with search strings of the same classes as that in

the TMT-A and TMT-B respectively), and then estimate the subject's cognitive recall and search parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ , and  $\tau_S^B$  also using the model developed in Section 3.3.

#### 3.4.1 Estimating Motor Parameters

The first step in our two step process of estimating a subject's cognitive and motor parameters is to estimate the subject's Fitts' law motor parameter b. We use a data set consisting only of moves from SH rounds with the search string '1,2,3,4' and no distractors. As there is some uncertainty in the position of the mouse at the beginning of the round, we ignore the move to the first target for each round.

For each move, we know the inter-target distances  $D_i$ , target widths  $W_i$ , observed move times  $t_i$ , numbers of targets  $n_i$ , and the position of the target in the search string  $v_i$ (so for the string '1,2,3,4', the target '1' has v = 1, the target '2' has v = 2 and so on). So, for a particular move, the expected total time taken to move is:

$$\langle t_i \rangle = (\tau_R + a) + ((n_i - \nu_i + 1)/2)\tau_s + b \log_2(D_i/W_i + 1).$$
(3.5)

As the values  $\tau_R + a$ ,  $\tau_s$ , and b are unknown at this point, we have to fit all three to the data. We can estimate their values by finding the values  $c_0$ ,  $c_1$ , and  $c_2$  that minimize the total squared error given by:

$$\varepsilon = \sum_{i} \left( \langle t_{i} \rangle - c_{0} - |c_{1}| \left( (n_{i} - \nu_{i} + 1)/2 \right) - |c_{2}| \log_{2} \left( D_{i} / W_{i} + 1 \right) \right)^{2}.$$
(3.6)

We constrain the result so that  $c_1$  and  $c_2$  are non-negative. From this step, we only retain the estimated value  $b = |c_2|$ .

# 3.4.2 Estimating Cognitive Parameters

The first step in our two step process of estimating a subject's cognitive and motor parameters is to use the subject's Fitts' law motor parameter *b* estimated in the first step to remove the motor component of the move time and estimate the subject's cognitive parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ , and  $\tau_S^B$ . For estimation of  $\tau_R^A + a$  and  $\tau_S^A$ , we use a data set consisting only of moves from SH rounds with the ascending or descending alphabetic or numeric search strings excluding SH rounds with search string '1,2,3,4' and no distractors; and for estimation of  $\tau_R^B + a$  and  $\tau_S^B$ , we use a data set consisting only of moves from SH rounds with ascending or descending alphanumeric search strings. As there is some uncertainty in the position of the mouse at the beginning of the round, we ignore the move to the first target for each round.

As we did above in Eq. (3.5), we estimate the expected time taken for each move by summing the expected times for each of the three additive stages. In this case, however, we must also include values for the numbers of distractors  $d_i$  present for each move, giving the expected move time:

$$\langle t_i \rangle = (\tau_R + a) + ((n_i - \nu_i + d_i + 1)/2)\tau_s$$
  
+  $b \log_2(D_i / W_i + 1).$  (3.7)

As the values  $\tau_R + a$  and  $\tau_s$  are unknown at this point, we have to fit both to the data. We can estimate their values by finding the values  $c_0$  and  $c_1$  that minimize the total squared error given by:

$$\varepsilon = \sum_{i} \left( \langle t_i \rangle - c_0 - |c_1| \left( (n_i - \nu_i + d_i + 1) / 2 \right) -b \log_2 \left( D_i / W_i + 1 \right) \right)^2.$$
(3.8)

We constrain the result so that  $c_1$  is non-negative. From this step, we retain the estimated values  $\tau_R + a = c_0$  and  $\tau_S = |c_1|$ .

# 3.5 Estimating TMT Scores

Using the procedure given in Section 3.4, we can produce estimates for the cognitive and motor parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and *b* for each subject using data from play of SH. We now use these estimates to produce estimators of performance on TMT-A and TMT-B for each subject.

# 3.5.1 TMT Score Estimator

We begin our construction of the estimators for the performance on TMT by assuming we had the actual values of the cognitive and motor parameters  $\tau_R^{A'} + a'$ ,  $\tau_R^{B'} + a'$ ,  $\tau_s^{A'}$ ,  $\tau_s^{B'}$ , and b' that describe the subject's performance on TMT. Let us consider first the time spent completing TMT-A or TMT-B after the first target has been selected. The search complexity  $\tilde{\kappa}$  for this portion of the test can be calculated from the definition in Eq. (3.3); while the motor complexities  $\tilde{\chi}^A$ , and  $\tilde{\chi}^B$  for this portion of the test can be found using the definition in Eq. (3.3) and direct measurement of the layout of the markers on the test page. Using the superscript X as a place-holder for either A or B, indicating whether TMT-A or TMT-B is being considered, the expected total time spent completing the test after the first target has been found when no errors are made is:

$$\langle T^{X} \rangle = 24 \left( \tau_{R}^{X'} + a' \right) + \tilde{\kappa} \tau_{S}^{X'} + \tilde{\chi}^{X} b'.$$
 (3.9)

One aspect of difference 2 in Section 3.2.3 is that the subject begins the test by turning over the test page. This adds some amount of amount  $T^0$  to the final time. We suppose

 $T^0$  is the same for all subjects for both parts of TMT. In addition, the move to the first target begins at some unknown position. We suppose that the motor portion of the time taken to make the move to the first target is about average for the motor times on the test, or  $a + (1/24)\tilde{\chi}^X b'$ . When the first move is included, the search complexity is now  $\kappa$  as given in Eq. (3.3) for the full test rather than  $\tilde{\kappa}$ . The expected total time spent completing the entire test when no errors are made is:

$$\langle T^{X} \rangle = T^{0} + 25 (\tau_{R}^{X'} + a') + \kappa \tau_{S}^{X'} + (25/24) \tilde{\chi}^{X} b'.$$
 (3.10)

We denote the number of errors the subject made on the TMT-A and TMT-B respectively by  $N^A$  and  $N^B$  and treat time required by the subject to make and recover from an error as a random variable with mean  $\theta$ ; we further assume that the random variable is the same for all subjects. Thus, the expected test score (or expected total time spent completing the test when errors are made) given a expected numbers of errors  $\langle N^A \rangle$  and  $\langle N^B \rangle$  is:

$$\langle S^{X} \rangle = \langle T^{X} \rangle = T^{0} + 25 \left( \tau_{R}^{X'} + a' \right) + \kappa \tau_{S}^{X'}$$
  
+  $\left( 25/24 \right) \tilde{\chi}^{X} b' + \theta \langle N^{X} \rangle.$  (3.11)

Finally, we must replace the cognitive and motor parameters  $\tau_R^{X'} + a'$ ,  $\tau_S^{X'}$ , and b' describing performance on TMT by their counterparts that have been estimated from SH. Replacing the TMT cognitive and motor parameters  $\tau_R^{X'} + a'$ ,  $\tau_S^{X'}$ , and b' by their SH counterparts  $\tau_R^X + a$ ,  $\tau_S^X$ , and b using the transformation between the two connect-thedots tasks is given in Eq. (3.4) gives the estimator for expected scores on TMT given expected numbers of errors:

$$\left\langle S^{X} \right\rangle = \left(T^{0} + 25\alpha\right) + 25\beta \cdot \left(\tau_{R}^{X} + a\right) + \kappa \tau_{S}^{X} + \left(25/24\right) \tilde{\chi}^{X} b + \theta \left\langle N^{X} \right\rangle.$$

$$(3.12)$$

# 3.5.2 Estimating Global Parameters

The subject-specific cognitive and motor parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and *b* appearing in Eq. (3.12) have been estimated using SH move data. The global parameters  $T^0 + 25\alpha$ ,  $\beta$ , and  $\theta$  (assumed to be the same for all subjects) now need to be estimated. We estimate the global parameters by finding the values of  $T^0 + 25\alpha$ ,  $\beta$ , and  $\theta$  that cause our estimators given in Eq. (3.12) to have optimal performance combined for both TMT-A and TMT-B across all subjects.

We index our subjects so that every subject has some index *i* in the data related to TMT-A, and some index *j* in that related to TMT-B. For each subject we have measurements average test scores  $\langle S_i^A \rangle$  and  $\langle S_j^B \rangle$ , and the average numbers of errors made in each part  $\langle N_i^A \rangle$  and  $\langle N_j^B \rangle$ . As the values  $T^0 + 25\alpha$ ,  $\beta$ , and  $\theta$  are unknown at this point, we have to fit all three to the data. We can estimate their values by finding the values  $c_0$ ,  $c_1$ , and  $c_2$  that minimize the total squared error given by:

$$\varepsilon = \sum_{i} \left( \left\langle S_{i}^{A} \right\rangle - c_{0} - 25 |c_{1}| \cdot \left(\tau_{R}^{A} + a\right)_{i} -\kappa \tau_{S,i}^{A} - \left(25/24\right) \tilde{\chi}^{A} b_{i} - |c_{2}| \left\langle N_{i}^{A} \right\rangle \right)^{2} +\sum_{j} \left( \left\langle S_{j}^{B} \right\rangle - c_{0} - 25 |c_{1}| \cdot \left(\tau_{R}^{B} + a\right)_{j} -\kappa \tau_{S,j}^{B} - \left(25/24\right) \tilde{\chi}^{B} b_{j} - |c_{2}| \left\langle N_{j}^{B} \right\rangle \right)^{2}.$$

$$(3.13)$$

We constrain the result so that  $c_1$  and  $c_2$  are non-negative. We retain the estimated values  $T^0 + 25\alpha = c_0$ ,  $\beta = |c_1|$ , and  $\theta = |c_2|$ .

#### **3.6 Empirical Study**

30 older adults (25 female and 5 male, average age  $80 \pm 6.0$  years, average level of education  $15 \pm 2.7$  years, MMSE =  $28 \pm 1.1$ , ADL =  $0.071 \pm 0.30$ ) participated in a one year study in which a set of computer games that included SH was placed into their homes.

SH was developed along with 8 other adaptive computer games to measure cognitive performance of individuals on a regular basis by monitoring their computer interactions during game play on their home computers. [109, 59] The set of computer games, including SH, was placed in subjects' homes for a period of one year. Subjects were encouraged to play all the games often, but were free to play the games as little as they wanted. Play of the computer games by the subjects was monitored, and the relevant information needed to reconstruct a subject's play in any of the games was recorded in a central database in a format allowing us to reconstruct any round of SH played. Subjects were given a battery of cognitive tests, including TMT, administered by trained clinical staff according to standard administrations procedures, at the beginning of the study, 6 months into the study and at the end of the study.

We restricted the analysis to those subjects for whom, when only error-free rounds of SH with alphabetic, numeric, and alphanumeric search strings were considered, we could find at least 25 moves total (across all such rounds) for rounds using alphabetic or numeric and at least 25 moves total for rounds using alphanumeric search strings. Data from SH rounds with alphabetic and numeric search strings were combined in the analysis. Rounds with search string '1,2,3,4' and no distractors, were excluded, as were the first moves within each round for the purpose of determining whether a subject had

enough data. This restriction left a cohort of 23 older adults (20 female and 3 male, average age  $81 \pm 6.8$  years, average level of education  $15 \pm 2.9$  years, MMSE =  $28 \pm 0.89$ , ADL =  $0.058 \pm 0.16$ ).

The numbers of moves across the remaining cohort 23 subjects for SH rounds with the search string '1,2,3,4' and no distractors ranged from 24 to 1236 (median of 108), for rounds with alphabetic or numeric search strings (not including moves from rounds with search string '1,2,3,4' and no distractors) the numbers of moves ranged from 35 to 4618 (median of 259), and for rounds with alphanumeric search string ranged from 43 to 4819 (median of 273). Data for rounds ascending and descending alphabetic or numeric search strings were pooled together for estimating TMT-A performance as were data for ascending and descending alphanumeric search strings for estimating TMT-B performance so that as many subjects as possible could be retained for analysis.

Following the first step of our two step procedure for estimating subject cognitive and motor parameters from SH data given in Section 3.4, for each subject we estimated the value for the Fitts' law parameter *b* (Eq. (3.2)) by minimizing the total error expressed in the model given in Eq. (3.6) using all observed error-free SH rounds with search string '1,2,3,4' and no distractors. The observed means and standard deviations for the values of  $R^2$  and p for the fit of the model given in Eq. (3.6) across the 23 subjects were  $R^2 = 0.26 \pm 0.11$  and  $p = 0.0080 \pm 0.028$ , and the mean and standard deviation of the numbers of moves available for each subject for the estimated Fitts' law parameter *b* across the full cohort of 23 subjects was:

$$b = 300 \pm 110 \,\mathrm{ms}\,/\,\mathrm{bit.}$$
 (3.14)

The SH rounds chosen for the estimation of the Fitts' law parameter b were intended to be those for which the motor component would be strongest. The model fit all subjects at a significance level of p < 0.05. The low R<sup>2</sup> values are expected due to the uniform distribution of the number of search steps given a number of unselected targets on the board described in Section 3.3.2. Our average value for b is close to the independently measured value of 166 ms/bit measured for point-select methods of selecting icons in a computer interface, [110] with the measured value being about one deviation below our average b. We consider this further in Section 3.7.

Continuing to the second step of our two step procedure for estimating subject cognitive and motor parameters from SH data, we estimated the recall and search performances  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ , and  $\tau_S^B$  for each subject using the previously estimated values of b by minimizing the total error expressed in the model given in Eq. (3.8) using data from alphabetic and numeric search strings to estimate  $\tau_R^A + a$  and  $\tau_S^A$ , and alphanumeric search strings to estimate  $\tau_R^B + a$  and  $\tau_S^B$ . The observed means and standard deviations for the values of  $R^2$  and p for the fit of the model given in Eq. (3.8) across the 23 subjects for estimation of the parameters  $\tau_R^A + a$  and  $\tau_S^A$  were  $R^2 = 0.097 \pm$ 0.061 and  $p = 0.065 \pm 0.17$  with the number of moves available use in the estimation having mean and standard deviation  $n = 670 \pm 1200$ , and for estimation of the parameters  $\tau_R^B + a$  and  $\tau_S^B$  were  $R^2 = 0.087 \pm 0.050$  and  $p = 0.049 \pm 0.12$  with the numbers of moves available use in the estimation being  $n = 730 \pm 1300$ . In the case of the estimation of  $\tau_R^A + a$  and  $\tau_S^A$ , four of the subjects had fits with p > 0.05, and among these subjects the numbers of moves available for estimation had means and standard deviations of  $n=52\,\pm$ 

19; similarly for the case of the estimation of  $\tau_R^B + a$  and  $\tau_S^B$ , four of the subjects had fits with p > 0.05, and among these subjects the numbers of moves available for estimation had means and standard deviations of n = 82 ± 38. Two subjects had fits with p > 0.05 in both cases. Removing the appropriate four subjects with poor fits in the each of the two cases gives, for the remaining 19 subjects, in the former case R<sup>2</sup> = 0.11 ± 0.057, p = 0.0032 ± 0.0077, and n = 800 ± 1300, and, for the remaining 19 subjects, in the latter case R<sup>2</sup> = 0.10 ± 0.043, p = 0.0039 ± 0.0095, and n = 830 ± 1300. The observed means and standard deviations of the estimated cognitive parameters across the full cohort of 23 subjects were:

$$\tau_{R}^{A} + a = 380 \pm 250 \,\mathrm{ms},$$
  
 $\tau_{R}^{B} + a = 660 \pm 280 \,\mathrm{ms},$ 
  
 $\tau_{S}^{A} = 96 \pm 43 \,\mathrm{ms},$ 
  
 $\tau_{S}^{B} = 110 \pm 41 \,\mathrm{ms}.$ 
(3.15)

The model, again, fit most subjects to a statistical significance level of p < 0.05, and the cases where this level of significance was not met appear to be attributable to the smaller amount of data available. Again, the low R<sup>2</sup> values are expected due to the uniform distribution of the number of search steps given a number of unselected targets and distractors on the board described in Section 3.3.2. Due to the fact that we are measuring combined cognitive and motor values  $\tau_R^A + a$  and  $\tau_R^B + a$  rather than the purely cognitive values  $\tau_R^A$  and  $\tau_R^B$ , we could not compare the measured values to existing research. However, we could estimate the set-switching (the time needed for the subject to switch from considering the sequence of numbers to considering the sequence of letters and vice versa in TMT-B) by taking the difference of  $\tau_R^A + a$  and  $\tau_R^B + a$ . The average estimated

set-switching time of 280 ms compared well with independently measured values of about 200 ms. [111] The average estimated values for  $\tau_s^A$  and  $\tau_s^B$  for each step in visual search differed from the independently measured value of 240 ms [112] by about a factor of two. We consider these further in Section 3.7.

The observed TMT-A and TMT-B scores and numbers of errors across all the tests taken by the subjects being included in this analysis and their standard deviations were  $S^A = 45 \pm 11$  s and  $N^A = 0.0073 \pm 0.14$ , and  $S^B = 100 \pm 28$  s and  $N^B = 1.0 \pm 0.64$ . The observed scores are near those given in Section 3.2.1 (i.e. [27]) for subjects around the age and education of those used in our study. The observed TMT test-retest reliability for the test pairs: (1) beginning and 6 months, (2) beginning and 1 year, and (3) 6 months and 1 year, was observed to have R<sup>2</sup> of 0.32, 0.20, and 0.13 for TMT-A, and R<sup>2</sup> of 0.43, 0.30, and 0.59 for TMT-B. As we used the full year's worth of data to estimate subject performance, we characterized subject performance on TMT using averages of the test scores over the year.

We next constructed estimators of the mean TMT scores given the mean numbers of errors made on the tests using the procedure given in Section 3.5. The motor complexities for the TMT-A and TMT-B for all moves after the first target has been selected were measured from the test pages using a ruler, giving  $\tilde{\chi}^A = 66$  bits, and  $\tilde{\chi}^B = 74$ bits. Using the values of the cognitive and motor parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and *b* that we estimated for the 23 subjects (Eqs. (3.14) and (3.15)), we estimated the global parameters  $T^0 + 25\alpha$ ,  $\beta$ , and  $\theta$  by minimizing the total error expressed in the model given in Eq. (3.13) using, for each subject the means of the three test scores and the means of the numbers of errors made in the course of each test administration. The model was fit with  $R^2 = 0.82$  and p < 0.0001, and the estimated global parameter values were:

$$T^{0} + 25\alpha = -9.1s,$$
  
 $\beta = 2.2,$  (3.16)  
 $\theta = 30s.$ 

For comparison, we looked at the performance of a simple linear regression of the test scores on the number of errors; the fit in this case had  $R^2 = 0.58$  and p < 0.0001.

Inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$  and  $\theta$  were statistically significant, while that of  $T^0 + 25\alpha$  was not. The values in Eq. (3.16) suggested that subjects required 30 s to recover from an error. We consider this further in Section 3.7. In Figure 3.4, we show how the estimated average TMT-A and TMT-B test scores using these values of the global parameters compared to the actual average test scores for each subject.



**Figure 3.4.** Actual vs. estimated TMT scores across subjects. Each of the 23 subjects has two values shown, one for TMT-A and one for TMT-B, each representing the average of the three administrations of TMT. The model fit has  $R^2 = 0.82$  and p < 0.0001. A line with slope one passing through the origin is shown for reference.

It was of interest to see how the model developed in this paper would perform in the case where no errors were made on TMT. We restricted the analysis to only include administrations of TMT in which both TMT-A and TMT-B had no errors. There were 16 subjects who had at least one error-free administration of TMT. We fitted a truncated version of our model in Eq. (3.13) lacking the terms in  $\theta N^A$  and  $\theta N^B$ , and used the average of all error-free TMT administrations for the test scores. The model fit with R<sup>2</sup> = 0.55 and p < 0.0001, and estimated global parameter values of  $T^0 + 25\alpha = -9.1$  s and  $\beta = 3.1$ . Inspection of the confidence intervals for the coefficient estimates showed that the estimate of  $\beta$  was statistically significant while that of  $T^0 + 25\alpha$  was not. The results were close to those found by retaining tests where errors were made in Eq. (3.16).

If the subject whose data give the outlier is removed from the data set, and the procedure repeated, the fit became  $R^2 = 0.73$  and p < 0.0001, and the global parameters

were found to be  $T^0 + 25\alpha = -6.3$  s,  $\beta = 2.0$ , and  $\theta = 29$  s, with  $\beta$  and  $\theta$  being statistically significant while  $T^0 + 25\alpha$  was not. These results were very close to those found when retaining the outlier in Eq. (3.16), so we retained the outlier in our analysis. The linear regression of the test scores on the number of errors had  $R^2 = 0.51$  and p < 0.0001. When we limited ourselves to error-free administrations of the TMT, we were left with the same set of 16 subjects as in the previous paragraph.

#### **3.7 Discussion**

The connect-the-dots model developed in this study is a very simple model of the connect-the-dots task, and incorporates a number of simplifying assumptions, particularly in the search stage. Our simple model relating the measurements from SH to the case of TMT using a single set of transformations taken to be approximately the same for all subjects is based on the assumption of a relatively homogeneous set of subjects. A broader range of subjects may need to be grouped into classes each with a different set of transformations. A further limitation to our analysis presented here is the exclusion of SH rounds with errors, possibly limiting our ability to collect data for (mildly) cognitively impaired subjects.

We treated the time spent recalling the next target in the sequence as a single RV that appears once in each move. Because our empirical measurements did not allow to separate the effects of cognitive and motor segments, we were unable to estimate the recall times  $\tau_R$ , but had to measure a combined cognitive and motor parameter  $\tau_R + a$ . As a result, we could only average estimated set-switching time of 280 ms measured by taking the difference of  $\tau_R^A + a$  and  $\tau_R^B + a$  to independent measurements; though they compared well to the independently measured value of about 200 ms. [111] The model we developed to describe the transformation from the SH case to the TMT case should have taken the form  $\tau_R' = \alpha + \beta \tau_R$ , but, since we could only use values  $\tau_R + a$ , we had to make the approximation  $\tau_R' + a' \approx \alpha + \beta \cdot (\tau_R + a)$  which causes the motor parameter *a* to change value between the two cases. Even were this not a problem, the transformation  $\tau_R' = \alpha + \beta \tau_R$  causes the set-switching time to change between the SH case and the TMT case, where it seems reasonable that the set-switching time would not change.

The model we use for visual search is that of a serial search that does not benefit from the memory from previous search stages for previous "dots," during any given round, but has perfect memory within the search stage for the current "dot." Using this model, we estimated times spent on each step of search of about 100 ms in both cases ( $\tau_s^A$  and  $\tau_s^B$ ). The natural sequence of images produced by the eye during visual search has been independently measured to be about 240 ms per image. [112] This suggests that our model over-estimates the average number of steps in the visual search for a given target by about a factor of two. A more sophisticated model of search would include aspects related to: (1) visual acuity and how much target information the subject can take in at each search step, (2) the ability of the subject to remember target locations from previous searches, (3) the degree to which the subject becomes confused and considers the same target multiple times during a single search, and (4) the ability of the subject to ignore already selected targets and whether they spend much time considering these targets in later searches. What our simple model of search does capture is the observation that the total search times for moves are, on average, longer earlier in the connect-the-dots task than they are later in the task. [102]

We used Fitts' law to describe the motor portion of a move. An average value for the Fitts' law parameter of  $b \approx 300$  ms/bit was measured across the subjects analyzed. We can compare this value to independently measured values for Fitts' law for several methods of using a computer mouse to select a icon: (1) a = 135 ms and b = 249 ms/bit for drag-select, (2) a = 230 ms and b = 166 ms/bit for point-select, and (3) a = 135 ms and b = 249 ms/bit for stroke-through. [110] Point-select should be closest to the button selection process happening in SH, so our average value of b is roughly twice as large of the independently measured value (although that value is approximately one standard deviation below our average b).

We included the time spent recovering from an error as a single global parameter with the same value for all subjects. The estimated 30 s recovery time for each error is somewhat long (we do not have data on the duration of errors during the administrations of TMT). However, the average number of errors on TMT-A was near zero and that on TMT-B near one, so it is likely that the value estimated for  $\theta$  largely reflects the time needed to recover from errors during TMT-B. The value for  $\theta$  may well be inflated by a correlation where subjects making more errors also require more time to recover from each one. Alternatively, it is possible that, when taking the TMT, subjects also made a number of "near errors" in which they came close to selecting and incorrect "dot," but corrected the error themselves. If the number of these "near errors" is correlated to the number of actual errors, then the large error recovery time may reflect time taken up in a "near error" process as well.

Possibly related to variability in the numbers of errors between administrations of TMT and variability in the error-recovery time is the low test-retest reliability observed in Section 3.6. It appears from our results that errors can contribute a large amount of time to the test scores, and errors are discrete events, so differing numbers of errors from one test to another appear to be able to result in substantially different scores from one test to another.

Moving forward, it is important to better understand the process of making errors in the course of the connect-the-dots task, and there are a variety of approaches to doing this. There are four approaches one might take in considering the errors. (1) Look at the rate at which errors are made in playing SH and see whether this rate predicts the average number of errors made on TMT, or allows us to produce good estimates of the average TMT score without using the observed number of errors on TMT. (2) Look at how observed errors in SH relate to moves made immediately before and after. In our analysis, we have dropped all SH rounds in which any errors were made. The amount of data available, particularly for (mildly) cognitively impaired subjects, would be increased by dropping moves expected to be affected by observed errors from the data set rather than whole rounds. (3) Look at the moves in SH in which errors are made and see if the time spent recovering from the error can be used to predict error recovery times in TMT. (4) Look at outliers in the move times and see if the frequency and average duration of these outliers could be related to the number of errors or error recovery time in SH or TMT, suggesting that the outliers may be "near errors."

# **3.8 Conclusion**

The key objective of this work included (1) the development of techniques that would allow frequent assessment of cognitive functions of individuals at risk, for example associated with aging and (2) the demonstration of the utility of computational modeling in assessment of cognitive function. The general approach was based on using computer games that would enable neurophysiologic assessments comparable to those obtained with traditional neuropsychological tests such as the TMT. In this study, we used a simple game that is similar to the TMT and was developed for this purpose as a part of prior study. [109, 59, 67] In addition to the estimation of the TMT performance, the objective of using the game was to derive a more refined assessment of the various cognitive components recruited in the execution of TMT.

The potential benefit of our approach to modeling computer interactions is that we can model cognitive performance over time for individuals in the home in a more scalable and less expensive manner than current standard practice. Cognitive measures and trends in these measures can be used to identify individuals for further assessment, to provide a mechanism for improving the early detection of neurological problems, and to provide feedback and monitoring for cognitive interventions in the home.

# The following appendix was not included in the published version of this Chapter

## 3.9 Appendix

We can analyze the model in Eq. (3.13) further using the same set of 23 subjects analyzed in Section 3.6. Rather than fitting the model in Eq. (3.13) once for all 23 subjects, we fit it 23 times, once for each subset of 22 subjects. These models fit with a mean and standard deviation of  $R^2$  of  $R^2 = 0.82 \pm 0.013$  and in all cases p < 0.0001. The means and standard deviations of the estimated global parameter values were:

$$T^{0} + 25\alpha = -9.1 \pm 1.0 \text{ s},$$
  

$$\beta = 2.2 \pm 0.096,$$
  

$$\theta = 30 \pm 1.2 \text{ s}.$$
  
(3.17)

The global parameter values estimated for each of the 23 models in Eq. (3.17) are close to those estimated in the single model using all 23 subjects in Eq. (3.16). We arrive at the same model using any set of 22 subjects that we do when using all 23 subjects.

We define  $S^x$  to be the actual average TMT scores and  $\tilde{S}^x$  to be the estimated average TMT scores for the individual subject left out of the estimation of the global parameter values in Eq. (3.17). The means and standard deviations of the resulting differences in the actual and estimated average TMT scores  $S^x - \tilde{S}^x$  using the estimated global parameter values for each of the 23 subject for TMT-A and TMT-B were:

$$S^{A} - \tilde{S}^{A} = -5.0 \pm 14 \,\mathrm{s},$$
  

$$S^{B} - \tilde{S}^{B} = 5.0 \pm 19 \,\mathrm{s}.$$
(3.18)

We can compare the values in Eq. (3.18) to the means and standard deviations of the errors when all 23 subjects are fit using a single model:

$$S^{A} - \tilde{S}^{A} = -5.0 \pm 13$$
s,  
 $S^{B} - \tilde{S}^{B} = 5.0 \pm 17$ s. (3.19)

The errors in the case where the model is fit using 22 subjects and the estimated global parameters were used to estimate TMT performance for the remaining subject were comparable to the errors when all 23 subjects were used to estimate the global parameter values.

# Chapter 4 – Towards the Unobtrusive and Ubiquitous In-Home Monitoring of Cognitive Performance Using Mouse Dynamics

# 4.0 Abstract

Early detection of decline in cognitive performance in older adults facilitates early medical intervention to treat the decline. Detection of decline requires on-going monitoring cognitive performance by which a baseline level of performance is established and decline from the baseline is noted. Detection of cognitive decline can be made earlier as the monitoring becomes more frequent. We look at the unobtrusive and ubiquitous monitoring of in-home (computer) mouse dynamics during everyday computer usage as a means of monitoring cognitive performance. We characterize cognitive performance by executive function as measured by the Trail-Making Test (TMT), and use measures derived from mouse dynamics to estimate cognitive performance by estimating performance on TMT. We further attempt to indicate the direction of future work to improve the measures we use here and to derive further measures from mouse dynamics.

# **4.1 Introduction**

Early detection of declines in cognitive performance is an important component of providing proper medical treatment for maintaining the health and well-being of an older adult. Detection of decline requires regular measurement of cognitive performance by which a baseline level of performance is characterized and declines from the baseline can be detected. As it takes several measurements of cognitive performance to overcome noise in the measurements and definitively detect cognitive decline, the frequency of the measurements of cognitive performance place a limit on how early a decline in cognitive performance can be detected. Currently, medical professionals perform regular measurements of cognitive performance during regular visits to a clinic. Clinical visits are infrequent and of short duration, limiting how quickly cognitive decline can be detected. An alternative to clinical visits is unobtrusive in-home monitoring through the placement of sensors in the subject's home. Unobtrusive in-home monitoring can produce measurements of cognitive performance more frequently and so provide for earlier detection of decline.

Unobtrusive in-home measurements of cognitive performance for older adults have been developed using measurements of in-home activity. [71, 69, 70] Some in-home measurements of specific sorts or aspects of activity that have been developed include walking speed [Chapter 2, [79, 80]], time spent using a computer [113], typing speed [114], time away from home [115], and sleep [116, 117]. Unobtrusive, in-home measurement techniques have also been applied to fall risk assessment and fall detection for older adults. [84, 85]

A further, potentially rich, source of information about the subject that one can measure unobtrusively in the home is how the subject moves a computer mouse. The measurement of (computer) mouse dynamics (MD) as an unobtrusive in-home measure of performance has been explored in a limited way in [118]. Measurements of MD have been explored for use as a biometric for authenticating a computer user's identity. To this end, variety of methods have been put forward which characterize mouse movements by a number of geometrical (the shape of the path of the mouse movement) and dynamical measurements (the timing of the mouse movement), and use machine learning techniques to create classifiers capable of identifying individual subjects. The verification of the subject's identity can be done statically (i.e. the subject completes a set of specified tasks which are compared to recorded previous performances of the same tasks) (e.g. [119]), or continuous (i.e. the subject uses the computer normally and measurements of how the computer is used are compared to characteristic data for the subject) (e.g. [120-122]). The continuous techniques for verification of a subject's identity during computer usage are a form of unobtrusive monitoring.

Rather than identify the subject using MD, we would like to use MD as a measure of subject performance. Cognitive performance and motor performance have been shown to be intertwined, with seemingly motor-only tests functioning as predictors of cognitive decline. The timed up and go test, in which the subject rises from a chair, walks three meters, turns around, returns to the chair and sits down, has been shown to predict cognitive decline; [123, 124] declining walking speeds [20, 125, 126] and motor speeds as characterized by finger-tapping speeds [20, 127] have been shown to indicate cognitive impairment. Measurements of MD during controlled experiments have been used to investigate a variety of cognitive processes, including the cognitive dynamics of the negation-operator in linguistics, [128] the cognitive dynamics in shape and pigment processing in face categorization, [129] and the underlying response confidence during recognition decisions, [130] as well as the effect of semantic priming on mouse movements. [131]

In the present paper, we begin the process of developing techniques to use in-home measurements of MD to monitor the cognitive and motor performance of a subject. We provide solutions to practical problems related to recording and storing large amounts of raw mouse position data for the purpose of on-going analysis. We also provide an algorithm for identifying individual mouse movements within the stored raw data. We show that we can use the mouse movements identified by the algorithm to produce performance metrics, and that these performance metrics capture some amount of intersubject variability by using them to predict subject performance on the Trail-Making Test (TMT) – a standard clinical test in which the subject draws a line through a series of targets printed on a page. To provide a conceptual schema for understanding the MD data and how best to analyze it, we use a mathematical model that describes how a subject completes TMT. We suggest that the process of drawing a line through a series of targets bears a resemblance to the process of navigating a computer by moving a mouse cursor through a series of control widgets.

The data available for analysis cover 210 older adults observed over a period of 6 years. The very large quantity of mouse data available necessitated that the techniques developed require as few computational resources as possible so that our computer could perform the calculations in a reasonable amount of time. The solution of these problems only constitutes the first steps of a wider program of measuring cognitive performance from MD. In addition to looking at these three problems, we indicate possible directions for carrying out the next steps for improving the measures of cognitive performance we develop here and producing further measures from in-home MD.

## **4.2 Cognitive Performance (Executive Function)**

We begin by presenting a conceptual schema for understanding MD and its relationship to cognitive and motor performance. We characterize cognitive performance using executive function, and, in turn, characterize executive function using the score on TMT, a clinical neuropsychological test known to measure executive function. [37, 34, 35] The subject completes TMT by drawing a line through a series of targets printed on a page. We can analyze the TMT score into performance on component tasks using the connectthe-dots model developed in Chapter 3. In this model, we decompose the process of drawing the line from one target to the next into a series of three additive stages. The first two stages are purely cognitive in nature, while the third describes the physical movement to the next target using Fitts' law. [105-107] The third stage relates TMT to the MD of physically moving the mouse. In the interest of making the present treatment as self-contained as possible, we reproduce the model here in detail.

The remainder of Section 4.2 provides a review of material appearing in Chapter 3

#### 4.2.1 Trail-Making Test

TMT is one of the most clinically useful neuropsychological tests and is used routinely in the diagnosis of many neurological conditions (Parkinson's, Alzheimer's and dementias, and general cognitive decline). [33] As we show, the process by which a subject completes TMT resembles the way subject controls a computer using a computer mouse. A natural way of constructing a measure of executive function from MD is to construct an estimator of a subject's score on TMT using MD; and this is how we approach the analysis of subjects' MD. So in order to better understand how to approach the MD data in order to estimate cognitive performance using executive function, we begin by looking in detail at the structure of TMT.

TMT consists of two connect-the-dots tasks (TMT-A and TMT-B) which the subject completes by using a pen to draw a single line through a series of targets on a test page. Each test takes the form of a standard 8.5"x11" sheet of paper on which is printed 25 *targets* – small (12mm diameter), seemingly randomly scattered circles containing a label that may be a letter or a number. For TMT-A, the targets are labeled with all numbers from 1 to 25, with one number labeling each circle, while in TMT-B targets are labeled with all letters from A to L and all numbers from 1 to 13 with one letter or number labeling each circle. The individual completes the TMT-A by using a pen to draw a line (the "trail") through all the circles in ascending numerical order of the labels (i.e. '1,2,3,...,24,25'), while the TMT-B is completed by drawing the line through the circles in ascending alphanumeric order (i.e. '1,A,2,B,...,L,13'). We call the process of drawing a line from one target to the next in the sequence a move. The score on each part of the test is the time the subject needs to complete each task (there is no time limit). If the subject makes an error on the test (drawing the line through an incorrect target), the test administrator stops the subject as soon as the error is noted by the administrator and returns the subject to the last correctly selected target; timing of the subject is not stopped during this error recovery process. The total numbers of errors made on each part are included with the test score. The difference between TMT-A and TMT-B is the use of the numeric and alphanumeric sequence of labels respectively. The alphanumeric label sequences introduce the additional complication of *set-switching* in which the subject

must not only recall the next element in an alphabetic or a numeric sequence, but must also switch sets between the alphabetic and numeric sequences.

TMT characterizes the subject's executive function using two pairs of numbers: (1) the scores on each of the two parts, and (2) the numbers of errors made in completing each of the two parts. Normative data for subjects with education in the range of the subject used in this study show TMT scores for ages 75-79 of 42 s on TMT-A and 100 s on TMT-B, and for ages 80-84 of 55 s on TMT-A and 130 s on TMT-B. [27] The test re-test reliabilities for TMT have been observed to be  $R^2 = 0.56$  for TMT-A and  $R^2 = 0.72$  for TMT-B for a cohort of normal adult controls, scores of  $47 \pm 25$  s on TMT-A and  $120 \pm 86$  s on TMT-B. [28] Guidelines for the administration of TMT are provided in [29].

# 4.2.2 Connect-the-Dots Model

The connect-the-dots model developed in Chapter 3 describes the process by which a subject makes each move in TMT by dividing the move into a sequence three additive stages. [103, 104] The stages are (1) *recall and updating*, in which the subject recalls the next target in the target sequence, (2) *search*, in which the subject searches for the recalled target, and (3) *motor*, in which the subject moves the mouse to the target according to Fitts' law.

The recall and updating stage is characterized by the *recall time* required by the subject to recall the next target in the sequence. We treat the recall time as a random variable (RV)  $T_R$  with expected value  $\langle T_R \rangle = \tau_R$  (the *characteristic recall time*), and some standard deviation.

The search stage is characterized by the *search time*,  $T_s$ , required by the subject to locate on the test page the next target in the sequence after it has been recalled. Search is

treated as a series of discrete steps. [108] We assume that each step of search takes some fixed time  $\tau_s$  (the *characteristic search time*) and that the variation in search time is due to each search taking a variable number of steps; we treat the number of steps as a RV. At each step of search, the subject considers a randomly chosen unselected target on the test page; if that marker is the desired target, the subject ceases search and moves on to the motor phase, otherwise the subject considers another randomly chosen unselected target. We assume that the subject does not consider any unselected target twice during a particular search. For a move during a part of TMT, the subject is searching for the v-th of 25 targets, the subject has already found v-1 targets, and there are 26-vtargets still on the board. The expected number of steps for the search is (26-v)/2, so the expected value for the search time for this target is given by:

$$\langle T_s \rangle = \left( \left( 26 - \nu \right) / 2 \right) \tau_s. \tag{4.1}$$

The distribution of  $T_s$  for a given value  $\nu$  is uniform on the discrete values  $\tau_s$ , ...,  $(26-\nu)\tau_s$ .

The motor stage is characterized by the *motor time*  $T_M$ , a RV representing the time the subject takes to move the mouse to the next target. We assume that the time taken to make each move is consistent with Fitts' law. Defining  $D_v$  to be center-to-center distance between the v-1-th and v-th targets, assuming a common width W for all targets, the expected time taken to move from the v-1-th to the v-th target is given by:

$$\langle T_{M} \rangle = a + b \log_2 \left( D_{\nu} / W + 1 \right). \tag{4.2}$$

The value  $\log_2(D/W+1)$  provides a measurement of the amount of information the subject must process to complete the movement as measured in bits; so the value *b* 

provides a measure of how much time the subject spends processing each bit of information.

The expected total time  $\langle T \rangle$  required to complete TMT-A or TMT-B where no errors are made is the sum of the expected times spent making moving to each target in turn, that is:

$$\langle T \rangle = 25(\tau_R + a) + \kappa \tau_s + \chi(D_1, \dots, D_{25}, W)b,$$
  

$$\kappa = \sum_{\nu=1}^{25} ((26 - \nu)/2),$$
  

$$\chi(D_1, \dots, D_{25}, W) = \sum_{\nu=1}^{25} \log_2(D_\nu / W + 1).$$
(4.3)

We call  $\kappa$  the *search complexity*, and  $\chi(D_1, \dots, D_v, W)$  the *motor complexity*. We note that the estimate for expected total required to complete the task is linear in the cognitive and motor parameters  $\tau_R$ ,  $\tau_S$ , a, and b.

Eq. (4.3) provides an estimate of the expected time required to complete one part of TMT without any errors using the cognitive and motor parameters  $\tau_R$ ,  $\tau_S$ , a, and b. We allow that different values may apply to each part of TMT and characterize subject performance on TMT as a whole by the set of parameter values  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ ,  $a^A$ ,  $a^B$ ,  $b^A$ , and  $b^B$  where the superscript indicates whether the parameter is for TMT-A or TMT-B. When the subject makes no errors, TMT characterizes executive function by two values – the two test scores – the connect-the-dots model characterizes executive function by up to eight values. The search complexity  $\kappa$  for TMT-A or TMT-B (they are the same value) can be calculated from the definition in Eq. (4.3); while the motor complexities  $\tilde{\chi}^A$ , and  $\tilde{\chi}^B$  for the remainder of the test after the first target has been

selected can be calculated using the definition in Eq. (4.3) and direct measurement of the layout of the targets on the test page.

In practice, we must modify the estimate in Eq. (4.3) to include times related to other effects present in the test-taking process. When errors are allowed to occur on either part of TMT a parameter  $\theta$  giving the expected time needed to recover from an error must also be included. The act of turning over the test page and moving to the first target adds some time  $T^0$  in addition to the recall, search and motor times to the total time. Using X as a place-holder for either of superscripts A or B, given a number of errors  $N^X$ , a subject's expected score on TMT-A or TMT-B (the expected time required to complete the test)  $\langle S^X \rangle$  is:

$$\left\langle S^{x} \right\rangle = T^{0} + 25 \left( \tau_{R}^{x} + a^{x} \right) + \kappa \tau_{S}^{x}$$

$$+ \left( 25/24 \right) \tilde{\chi}^{x} b^{x} + \theta N^{x}.$$

$$(4.4)$$

#### 4.3 Measuring In-Home MD

In Section 4.2, we provided a conceptual schema for understanding the in-home MD and its relationship to cognitive performance. The schema characterizes cognitive performance as the score on TMT, that is, the time taken to complete a sequence of simple motor tasks. It uses the connect-the-dots model that decomposes each of the simple motor tasks as an additive sequence of cognitive and motor stages. We would like to estimate cognitive performance using in-home MD by deriving estimates of performance for each of the cognitive and motor stages from measurements made from in-home MD. To begin this process of producing estimates of cognitive and motor performance on TMT, we must first put some structure on the raw mouse data by identifying the individual mouse movements and (implicitly) the intervals between mouse
movements when the mouse is idle. Once we have this structure, we can begin to do a more sophisticated analysis of the stored mouse data.

In this Section, we develop a technique for parsing the raw mouse data into a sequence of movements and idle intervals between movements. We also show how to estimate the motor parameters for Fitts' law using the identified mouse movements. We expect the motor parameters to provide an estimate of performance during the motor stage in TMT. We further begin the process of providing a more sophisticated analysis of the stored mouse data by deriving a simple physical model of mouse movements and using it to derive a simple, initial division of the mouse movements into two classes.

### 4.3.1 Preliminaries

A subject uses a computer in a series of *sessions* that begin when the subject logs-on to the computer and ends when the subject finally logs-off or leaves for an extended period. Within each session, the subject controls the computer by a series of movements made by the computer mouse. During a computer session, the mouse to alternates between intervals in which it is *idle* (not moving), and periods in which it is *active* (moving).

MD occurs in two spaces related by a non-linear transformation: (1) the first is the *pointer-space*, which involves movement of the pointer on the computer screen (2) the second is the *mouse-space*, which involves movement of the computer mouse on the tabletop. Distances in the former are expressed in *pixels* (distances on the computer screen), while those in the latter are expressed in *counts* (distances on the tabletop). One count indicates the smallest distance the mouse must travel before the computer recognizes that it has moved. The physical distance of movement corresponding to one count varies with the mouse hardware. The analysis of computer mouse movement in

[118] mistakes what are, in fact, mouse-space measurements for pointer-space measurements.

The transformation relating the pointer- and mouse-spaces is given by the *pointer* ballistics that relates the pointer velocity  $\vec{v}_{pointer}$  in the pointer-space to the mouse velocity  $\vec{v}_{mouse}$  in the mouse-space by a transfer function  $\vec{v}_{pointer} = G(\vec{v}_{mouse})$ , which appears roughly quadratic. [132] The intent of the transfer function is to facilitate the subject's ability to make both small and large movements of the pointer using reasonable movements of the mouse. [132]

#### 4.3.2 Storing Mouse-Position Data

We think of MD over the course of a session as being a stream of regularly sampled data. This is a very large amount of data, and we would like to store it as compactly as we reasonably can while being able to use it to recover as much information about the mouse movements as we can. The way we do this is to store the data with a variable sampling rate so that we sample the mouse-position more frequently when there is activity and less frequently when there is little activity.

The algorithm that we implement to record the mouse-position data uses eventtriggered sampling. We record the initial mouse-position (in mouse-space) at the beginning of data acquisition. We record subsequent mouse-positions and times whenever the event occurs that the mouse's position exceeds a distance of five counts from the last recorded position along a Manhattan metric in the mouse-space coordinate system for the duration of data acquisition.

#### 4.3.3 Identifying Mouse Movements

The event-triggered sampling procedure in Section 4.3.2 records the mouse movement during the session as an irregularly sampled time series of mouse-positions  $\vec{x}(t_n)$ . We would like to characterize MD during the session by identifying individual mouse movements in  $\vec{x}(t_n)$ . First, we interpolate a regularly sampled time series of mousepositions  $\vec{X}(t_k)$  that approximates the original raw mouse data. The sampling rate of the interpolated time series  $\vec{X}(t_k)$  is taken to be the highest observed sampling rate in the irregularly sampled time series  $\vec{x}(t_n)$ , typically 16 ms. We interpolate by assuming the mouse moved with constant velocity along a straight line between recorded data-points. From  $\vec{X}(t_k)$ , we construct a mouse speed time series  $|\vec{v}(t_k)|$  using  $\vec{X}(t_k)$  by taking  $|\vec{v}_k|$  $= |\vec{v}(t_k)| = |\vec{X}(t_{k+1}) - \vec{X}(t_k)| / \Delta t$ . Using the speeds that occur in the time series  $|\vec{v}(t_k)|$ ,

we can estimate the distribution of the logarithm of the mouse speeds over the course of the session. Figure 4.1 shows the kernel-smoothed density estimate of a typical distribution of the logarithm of the mouse speeds  $\{|\vec{v}_k|\}$  for a session lasting 12 min. The data are analyzed on the logarithm of the speed domain  $\{\log_{10}(|\vec{v}_k|/v_0)\}$  where  $v_0 = 1$ count/s. We observe that the distribution of the logarithm of the mouse speeds in Figure 4.1 is multimodal; this is typical for the distribution the logarithm of the mouse speeds in a computer session. We use the modes of this distribution to distinguish active intervals from idle intervals in the session. We identify time intervals associated with speeds in the highest mode of the distribution as being "active," and all other intervals as being "idle." To do this we associate the *cut speed*  $v_{cut}$ , which demarcates the boundary between active (higher velocity) intervals and idle (lower velocity) intervals, with the local minimum in the distribution between the mode containing the highest velocity intervals and the next lower mode. In Figure 4.1, the cut speed is the local minimum at about 67 count/s.



**Figure 4.1.** Mouse speed density for a single session. The kernel smoothed density of the interpolated regularly sampled mouse speed time series for a single computer usage session by a single subject. The speed  $v_{cut} = 67$  counts/s separates velocities associated with mouse activity with those associated with the mouse being idle. The session lasted about 12 min and 544 mouse movements were identified by the procedure.

We now return to the irregularly sampled time series of mouse-positions  $\vec{x}(t_n)$ . We can construct a mouse speed time series  $|\vec{v}(t_n)|$  for  $\vec{x}(t_n)$  just as we did for  $\vec{X}(t_k)$  albeit with a variable  $\Delta t$ . In this way, we associate a speed with the interval between any two consecutive mouse-position samples. We identify as "idle" any intervals with speed less than  $v_{cut}$  and as "active" those with speed greater than or equal to  $v_{cut}$ . We then identify a mouse movement with any consecutive sequence of active intervals between two idle intervals. Figure 4.2 shows horizontal and vertical mouse-position data (in mouse-space)

for a 5 s period. The mouse movement data (dots) have been parsed into five separate movements (indicated by alternating series of grey and black dots) after the calculation of  $v_{cut}$ . The intervals between the separate movements correspond to intervals with speeds below  $v_{cut}$ .



**Figure 4.2.** Sample mouse-position data. A sample segment of horizontal and vertical mouse-position data (in mouse-space) covering a 5 s period. The dots correspond to individual mouse-position measurements. There are 8 separate mouse movements indicated by alternating series of grey and black dots. The intervals between the separate movements correspond to intervals with speeds below  $v_{ent}$ .

We have observed that the distribution of the logarithm of the velocity is multimodal. It appears that we can use this observation to divide the raw mouse data into individual mouse movements. If this holds, then we have a way of identifying individual mouse movements using information contained in the stored mouse data itself without introducing any further arbitrary assumptions about how the subject carries out the mouse (such as how long the shortest idle interval between movements can be). The reader can view the remainder of this paper as laying out the structure of the mouse data when one identifies individual mouse movements by this technique. We consider limitations of this technique for identifying mouse movements in Section 4.6.

#### 4.3.4 Estimating Motor Parameters

The procedure given in Section 4.3.3 parses the raw mouse data for a computer session into a sequence of mouse movements and idle intervals between mouse movements. Each identified mouse movement has an associated time interval T – the time taken to make the movement – and an associated distance  $D_{\text{mouse}}$  that the mouse travelled in mouse-space. We can now use this parsed mouse movement data to estimate the portion of the executive function performance of the subject associated with the motor stage of the connect-the-dots model given in Section 4.2.2 as characterized by the motor parameters *a* and *b* in Fitts' law.

We assume that all identified mouse movements satisfy Fitts' law, the parameters a and b in Fitts' law are the same for all identified mouse movements, and that each identified mouse movement is a movement to a target of (unknown) width W. We also assume that Fitts' law is related to the visual feedback given by the pointer on the screen and that Fitts' law holds in pointer-space. Thus, each identified mouse movement satisfies the equation.

$$T_{M} = a + b \log_{2} \left( D_{\text{pointer}} / W \right). \tag{4.5}$$

The parameters *a* and *b* in Eq. (4.5) apply to all identified mouse movements, and each mouse movement is has its own values  $T_M$ ,  $D_{pointer}$ , and *W*. We would like to estimate the motor parameters *a* and *b* of Fitts' law in pointer-space using the observed times  $T_M$  and distances moved  $D_{mouse}$ .

We assume that, for typical mouse movements, the relationship between  $D_{\text{pointer}}$  and  $D_{\text{mouse}}$  is approximately linear and can be approximated by  $D_{\text{pointer}} \approx \eta D_{\text{mouse}}$ . This gives an approximate expression for Fitts' law in the mouse-space:

$$T_{M} \approx (a + b \log_{2}(\eta / W)) + b \log_{2} D_{\text{mouse}} + b \log_{2}(1 + (W / \eta) / D_{\text{mouse}}).$$

$$(4.6)$$

To simplify Eq. (4.6) further, we suppose that:

$$\left| \left( W / \eta \right) / D_{\text{mouse}} \right| < 1. \tag{4.7}$$

We confirm the validity of this supposition in Section 4.5. As we do not know the target width W for a given mouse movement we replace the target width with an approximate typical value  $\tilde{W}$ ; thus, we arrive at the following approximate expression for Fitts' law in mouse-space:

$$T_{M} \approx \tilde{a} + b \log_2 D_{\text{mouse}} + \left( b \tilde{W} / \eta \right) / D_{\text{mouse}}.$$
(4.8)

The parameter  $\tilde{a}$  is related to the motor parameter *a* by the expression:

$$a = \tilde{a} - b \log_2\left(\eta / \tilde{W}\right). \tag{4.9}$$

It turns out empirically that  $\eta/\tilde{W}$  is much less than one and thus  $\log_2(\eta/\tilde{W})$  is a negative number. As a result, estimated values of  $\tilde{a}$  are negative numbers.

In principle, we can estimate the motor parameters a and b for a computer session by taking the measured movement times  $T_M$  and distances moved  $D_{\text{mouse}}$  for all the identified mouse movements and calculating the values  $\tilde{a}$ , b, and  $b\tilde{W}/\eta$  that minimize the differences between the left-hand side (LHS) and right-hand side (RHS) of Eq. (4.8) in the ordinary least squares (OLS) sense. We can then use Eq. (4.9) to calculate a. In

practice, to avoid numerical problems, it is preferable to ignore the value estimated for  $b\tilde{W}/\eta$  using Eq. (4.8), and instead estimate the value  $\log_2(\eta/\tilde{W})$  across the population of subjects by fitting a linear model to the measured values of  $\tilde{a}$  and b.

## 4.3.5 Modeling Mouse Movements

In Section 4.3.4, we achieved our goal for the present paper of estimating the portion of the executive function performance of the subject associated with the motor stage of the connect-the-dots model. This is the first step to using in-home MD to estimate executive function as characterized by TMT. Future steps involve estimating the subjects' performance of the search and recall stages of the connect-the-dots model. We now try to derive more information from individual mouse movements beyond Fitts' law with the idea that a better understanding of individual mouse movements may help construct measures of search and recall. Also a better understanding of individual mouse movements we use in Section 4.3.4 to estimate Fitts' law to a set that gives better estimates of the motor parameters  $\tilde{a}$  and b.

Each identified mouse movement consists of a time series of position values. The data available for each mouse movement has additional structure beyond the distance moved and the time taken to make the movement. We can potentially gain more information about what was happening during the movement by looking at this structure for each movement. To do this, we fit a simple mouse movement model to each identified mouse movement. In the present paper, we use the model for the purpose of dividing mouse movements into two classes and seeing whether Fitts' law differs across the two classes or one class is more useful for making estimates of Fitts' law for estimating executive function using the TMT scores.

We construct the model of mouse movements by first assuming that the movements subjects make are, in some sense, better than the alternative movements. We formulate this mathematically by saying that the movements subject make minimize the value of some cost functional. [133-138] There must be some aspect of the movement (e.g., position, velocity, acceleration, etc.) that the subject has complete control over. By this, we mean that there is some aspect of the movement whose value the subject can choose at will, discontinuously. We assume the subject controls the jerk –  $\ddot{x}$  – and that the subject makes the movement in a way that keeps the magnitude of the jerk relatively small. [134-137] We write the cost functional associated with the movement that takes a time  $T_M$  in the form:

$$J[x] = \int_{0}^{T_{M}} L(x, \dot{x}, \ddot{x}, \ddot{x}) dt,$$

$$L(x, \dot{x}, \ddot{x}, \ddot{x}) = (1/2)\ddot{x}^{2} - \Phi(x, \dot{x}, \ddot{x}).$$
(4.10)

The minus sign appears in the RHS of Eq. (4.10) by analogy with the Lagrangian formulation of classical mechanics (see e.g., [139] Chapter 2); since the function  $\Phi(x, \dot{x}, \ddot{x})$  may take on both positive and negative values the minus sign in Eq. (4.10) places no restriction on the model being used. The movement the subject makes, the optimal orbit x(t) is the one that minimizes Eq. (4.10). We treat a mouse movement as beginning at rest at a position  $-D_{\text{mouse}}$  and travelling along an approximately straight line for the movement time  $T_M$  to the origin where it returns to rest. We express these initial and final conditions by requiring that the movement satisfy:

$$\begin{aligned} x(0) &= -D_{\text{mouse}}, \quad x(T_M) = 0, \\ \dot{x}(0) &= 0, \qquad \dot{x}(T_M) = 0, \\ \ddot{x}(0) &= 0, \qquad \ddot{x}(T_M) = 0. \end{aligned}$$
 (4.11)

The term in the jerk  $\ddot{x}$  in the integrand of Eq. (4.10) represents the cost associated with the value of the jerk in making the movement; the term in  $\Phi(x, \dot{x}, \ddot{x})$  represents any other costs associated with the movement. We assume the only cost represented by  $\Phi(x, \dot{x}, \ddot{x})$ is associated with ending in the target region. As the target is a purely geometric object, we expect the cost  $\Phi(x, \dot{x}, \ddot{x})$  to be approximately a function of position alone, that is  $\Phi(x)$ . We approximate the unknown cost  $\Phi(x)$  using a Taylor series expansion truncated to second-order in x:

$$\Phi(x) \approx \phi_0 + \phi_1 x \pm (1/2) \phi^{\pm} x^2, \quad \phi^{\pm} \ge 0.$$
(4.12)

The super script of  $\phi^{\pm}$  simply indicates whether sign of the second term on the RHS in Eq. (4.12) is positive or negative. The target is located at the end of the movement, that is, the origin. As the subject simply needs to end the movement within the target region, any point within the target region is as good a place to end the movement as any other. We model this by taking  $\Phi(x)$  to be approximately constant over the target region; this is done by taking  $\phi_1 = 0$ , giving:

$$\Phi(x) \approx \phi_0 \pm (1/2) \phi^{\pm} x^2, \quad \phi^{\pm} \ge 0.$$
 (4.13)

We do not try to impose any further physical requirements on the cost represented by  $\Phi(x)$ , and leave it in this general mathematical form.

The cost functional in Eq. (4.10) is minimized by setting the first variation to zero, that is (see e.g., [140] part I, Chapter 4):

$$\frac{d^3}{dt^3}\frac{\partial L}{\partial \ddot{x}} - \frac{d^2}{dt^2}\frac{\partial L}{\partial \ddot{x}} + \frac{d}{dt}\frac{\partial L}{\partial \dot{x}} - \frac{\partial L}{\partial x} = 0.$$
(4.14)

The *L* in Eq. (4.14) is the Lagrangian of the system, that is, it is the integrand of the cost functional in Eq. (4.10). The solution to Eq. (4.14) is the Euler-Lagrange equation, Taking the Lagrangian in Eq. (4.10) and using the form of  $\Phi(x)$  in Eq. (4.13) and plugging this into Eq. (4.14), we find the Euler-Lagrange equation for the model of mouse movements:

$$\overset{\dots}{x} \approx \pm \phi^{\pm} x.$$
 (4.15)

Eq. (4.15) gives the general form of an optimal orbit x(t) that minimizes the cost functional in Eq. (4.10). The orbit of a particular mouse movement is found by calculating the solution of Eq. (4.15) using the conditions given in Eq. (4.11). The solution to Eq. (4.15) can be found using the standard technique of assuming the solutions have the form  $x = \exp(rt)$ , plugging this form of the solution into Eq. (4.15) and calculating the values of r needed to satisfy the conditions in Eq. (4.11). We denote the orbits that solve Eq. (4.15) by  $x_{\phi}^{\pm}(t)$  according to the sign associated with the RHS in Eq. (4.15). Denoting by  $c_i$  constant values that are determined by the conditions in Eq. (4.11), and by  $R_i^{\mp}$  the sixth-roots of -1 and +1 respectively, these solutions take the form:

$$x_{\phi}^{\pm}(t) = \sum_{i=1}^{6} c_{i} \exp\left(R_{i}^{\mp}\left(\phi^{\pm}\right)^{1/6}t\right).$$
(4.16)

We call orbits of the form  $x_{\phi}^{-}(t)$  class I movements, and orbits of the form  $x_{\phi}^{+}(t)$  class II movements. In Figure 4.3, we show the data and fit of the models for mouse movements for example class I and class II movements traversing approximately the same distances in approximately the same times. It can be seen that class I movements reach the peak

velocity nearer the beginning of the movement while class II movements reach the peak velocity nearer the end.



**Figure 4.3.** Example class I and class II mouse movements. The dashed line indicates the observed mouse speed time-course of the movement, and the solid line indicates the time-course of the fitted model. Due to the way in which mouse movements are stored and estimated (see Section 4.3) the observed mouse speeds appear as a sequence of average velocities over time intervals. Note that for class I movements, the peak speed is nearer to the beginning of the movement and for class II movements, the speed is nearer to the end.

The cost  $\Phi(x)$  has a somewhat different functional form depending on the sign in Eq. (4.13), and it is the different sign and corresponding functional form of  $\Phi(x)$  that drives the division of mouse movements into two classes. An understanding of the circumstances in which a subject makes a class I or class II movement is needed in order to better understand what the cost  $\Phi(x)$  means. Additionally, the class of a movement may inform the analysis of the idle intervals preceding and following the movement.

#### 4.4 Proxies for Recall and Search Performance

In Section 4.3, we have provided a technique for parsing the raw mouse data into mouse movements and idle intervals between mouse movement, and a simple division of mouse movements into two classes. We have also shown how to estimate the parameters of Fitts' law for the mouse movements. We expect the estimated Fitts' law parameters to provide an estimate of performance of the motor stage in TMT. In the model we have developed to describe TMT, there are two cognitive stages, recall and search, in addition to the motor stage. In lieu of explicit measurements of the performance of these stages from the mouse data, we provide two proxy values that provide information about these two stages. The first is derived from in-home MD in the process that we used to identify the individual mouse movements. The second is the score on a further standard clinical neuropsychological test. We include the second proxy for the light it might shed on future work for extracting measures of recall and search stages from in-home MD.

#### 4.4.1 Cut Speed

In Section 4.3.3, we defined the cut speed  $v_{cut}$  and used it as a means to parse the raw mouse data into individual computer mouse movements. We can think of the cut speed  $v_{cut}$  as determined by two quantities. The first is a rough threshold indicating the lowest speed at which meaningful, controlled MD occurs during a computer session. The second is the length of the shortest idle intervals between movements. Looked at this way, we expect that the cut speed  $v_{cut}$  contains some information about cognitive performance. We define a measure related to the cut speed  $v_{cut}$  to use as a measure of cognitive performance. The measure of cognitive performance we use is the quantity that we actually measure in Section 4.3.3 and from which we derive the cut speed  $v_{cut}$ , namely:

$$\rho = \log_{10} \left( v_{cut} / v_0 \right). \tag{4.17}$$

The constant in Eq. (4.17) is  $v_0 = 1$  count/s.

# 4.4.2 Digit Symbol Test

Another standard clinical instrument used to measure executive function is the Digit Symbol Test (DST). [141] In this test, the subject is given a set of symbols to each of which a single digit has been assigned, and containing a list of the symbols (with symbols repeated) next to a blank space. For each symbol, the subject write in the blank space the digit associated with the symbol. The assignment of digits to symbols remains visible to the subject during the test. The score on this test is the number of correct digit assignments the subject makes in 90 s. DST tests the subject's ability to remember an assignment of arbitrary symbols to digits. We expect DST to provide information about the recall stage of TMT (where the subject must remember the next target in the sequence), and how quickly a subject recalls the next needed control widget (e.g., icons, buttons etc.) to use when navigating a computer application.

#### **4.5 Empirical Study**

210 older adults (121 female and 89 male, average age  $83 \pm 5.4$  years, average level of education  $16 \pm 2.6$  years, MMSE =  $28 \pm 1.9$ , ADL =  $0.51 \pm 1.2$  for the first year on record for the subject) were observed over a period of 6 years during which their computer usage was monitored. [114, 126] The subjects' computers ran using the Microsoft Windows XP operating system. We recorded and stored computer mouse-position data for all subjects in a database following the procedure described in Section

III.B. TMT and DST were administered as part of a battery of neuropsychological tests to each subject once each year. The observed means and standard deviations of the scores and numbers of errors on TMT-A and TMT-B across all administrations were  $S^A = 46 \pm 22$  s and  $N^A = 0.16 \pm 0.93$ , and  $S^B = 120 \pm 58$  s and  $N^B = 1.3 \pm 2.0$ . As data for neuropsychological tests were only available for each year, we reduced the measures for in-home MD to characteristic values for each (calendar) year; so the data we analyzed were the set of available one-year periods across all subjects. There were 929 one-year periods across subject with  $4.4 \pm 1.5$  one-year periods per subject.

Our aim was to validate the measurements made using in-home MD by estimating cognitive performance using executive function as characterized by the TMT scores. In order to perform this validation in a meaningful way, we limited our data set to those one-year periods in which we expected that the fraction of the TMT scores attributable to the motor stage (cf. Section 4.2.2) was largest. That is, we limited the data set to those one-year periods in which the subjects indicated the highest level of cognitive performance. We retained in the data set only those one-year periods in which the subject had perfect scores on MMSE (MMSE = 30) and ADL (ADL = 0), and made no errors on TMT-A and no errors on TMT-B. In addition, two one-year periods had exceptionally large scores on TMT, one having  $S^A = 91$  s and the other having  $S^B = 201$  s. We excluded these two one-year periods as outliers.

These restrictions left a cohort of 55 older adults (30 female and 25 male, average age  $83 \pm 5.8$  years, average level of education  $16 \pm 2.5$  years, for the first year on record for the subject in the restricted set). There were 102 one-year periods across subject with 1.9  $\pm$  1.1 one-year periods per subject. The observed means and standard deviations of the

scores on TMT-A and TMT-B across these one-year periods were  $S^A = 35 \pm 9.4$  s,  $S^B = 76 \pm 22$  s. Of the 55 subjects, 27 had more than one one-year period among the 102 one-year periods used for analysis, and the total number of one-year periods associated with these 27 subjects were 74 one-year periods. We used these 74 one-year periods to estimate how well the score on one completion of TMT estimated the score on the next completion on TMT available within the 74 one-year periods. The correlations were, for TMT-A,  $R^2 = 0.43$  and p < 0.0001, and, for TMT-B,  $R^2 = 0.43$  and p < 0.0001.

# 4.5.1 Measures of Motor Performance

We first look at the application to the technique presented in Section III of identifying individual mouse movements within the raw mouse data to observed in-home MD data. The aim is to show that identified mouse movements have the expected properties (e.g., they have reasonable time durations, and they yield the expected values for the motor parameters in Fitts' law). We then calculate the measures of motor performance (i.e. the motor parameters in Fitts' law) from the identified mouse movements.

We applied the technique presented in Section 4.3.3 of identifying the individual mouse movements within the raw mouse data to the monitored in-home MD to the full set of 929 one-year periods for the full cohort of 210 subjects. For any computer session that exceeded one hour, we analyzed only the first hour of the session. We did this because we found that the computer we were using to analyze the data had memory problems when the length of the sessions being analyzed became too long.

A total of 130446 computer sessions were observed over the full set of 929 one-year periods for the full cohort of 210 subjects. In order to keep the amount of data analyzed in each session manageable, we only considered session data up to the end of the first

hour of usage in a given session. The procedure to find a cut speed  $v_{cut}$  was applied to all sessions. It failed to find a value in 4850 cases. So, in about 96% of sessions, an appropriate  $v_{cut}$  was found and mouse data from the session could be divided into mouse movements. The means and standard deviations of observed cut speeds across subjects were  $v_{cut} = 67 \pm 19$  count/s.

We applied the techniques presented in Sections 4.3.4 and 4.3.5 of quantifying motor performance using Fitts' law and modeling the time-course of the identified mouse movements themselves to the restricted set of 102 one-year periods for the restricted cohort of 55 subjects.

For each one-year period, we separated the identified mouse movements into two sets corresponding to the class I and class II mouse movements defined in Section 4.3.5. We fit the optimal orbits  $x_{\phi}^{\pm}(t)$  calculated from Eq. (4.16) to each identified mouse movement and associated with each identified movement a value  $\phi^{\pm}$ . To improve the fit, for each movement additional initial and final measurements were interpolated for each movement and assigned an average velocity of zero. Solutions were required to satisfy Eq. (4.11) where the time  $T_M$  was the total movement time including the interpolated initial and final measurements, and  $D_{\text{mouse}}$  was the total distance traveled by the mouse (i.e. not the simply the distance between the initial and final positions  $\Delta_{\text{mouse}}$ ). In practice, we found it better to normalize the time in Eq. (4.16) so that it ran from 0 to 1. When this is done, a parameter  $(\phi^{\pm})^{1/6} T_M$  naturally appears in the exponential terms. As the  $c_i$  in Eq. (4.16) are determined by the conditions in Eq. (4.11), the parameter

 $(\phi^{\pm})^{1/6}T_{M}$  is what is actually fit to the data. As very large numbers of individual mouse movements were fit to the model, we sped up the process by choosing  $(\phi^{-})^{1/6}T_{M}^{I}/2\pi$  and  $(\phi^{+})^{1/6}T_{M}^{II}/2\pi$  being elements of the set {0.2, 0.25,...,2.0}. We chose the ranges after considering a subset of the data and calculating values  $\phi^{\pm}$  using a fitting technique with no constraint on the frequency values. Each movement was classified as being class I or class II movements according to whether a model in  $\phi^{-}$  or  $\phi^{+}$  gave a better fit to the data. We carried out subsequent analysis separately for each of the two sets of mouse movements.

For each one-year period, we calculated the ratio of the number of class I movements to the total number of movements. The mean and standard deviation of this ratio across one-year periods was  $0.73 \pm 0.035$ .

Models of Fitts' law (see e.g. [142]) use a motion that accelerates quickly to a peak speed and spends that larger part of the movement slowing while approaching the target (i.e. class I movements in our terminology). As one of the two movement classes arising from the model in Section 4.3.5 produced movements of the form expected for Fitts' law, and the other did not, we analyzed the data separately for the two classes of movements.

The sets of movements for each one-year period classified as class I movements had the following properties. We began by trying to characterize the fits of the model given in Section 4.3.5 to classify the mouse movements to evaluate how well that model was working. For each one-year period, we calculated the medians (M{ $\cdot$ }) and interquartile ranges (IQR{ $\cdot$ }) of (R<sup>I</sup>)<sup>2</sup> and p<sup>I</sup> values (i.e. R<sup>2</sup> and p values for class I movements) for each fit. We then took the means and standard deviations of the medians and

interquartile ranges across all one-year periods, and found  $M\{(R^{I})^{2}\} = 0.40 \pm 0.069$  and  $IQR\{(R^{I})^{2}\} = 0.45 \pm 0.026, \text{ and } M\{p^{I}\} = 9.2 \cdot 10^{-5} \pm 2.7 \cdot 10^{-4} \text{ and } IQR\{p^{I}\} = 0.016 \pm 0.0016 \pm$ 0.018. Although the fits are largely statistically significant, it appears we need a more sophisticated model of mouse movements to capture the variety of mouse movements. We next looked at the distributions of the time durations of the identified mouse movements. The means and standard deviations of the medians and interquartile ranges of the movement times  $T_M^I$  were M{  $T_M^I$  } = 390 ± 92 ms, IQR{  $T_M^I$  } = 370 ± 97 ms. For comparison, the lower limit of applicability of Fitts' law occurs at  $D_{\text{pointer}}^{I}/W = 1/2$ , which, for these subjects, corresponds to a movement time of about 190 ms. So, the middle 50% of the identified mouse movements had reasonable time durations. We consider the distribution of the movement times further in Section 4.6. Finally, we looked at how straight the movements were. The means and standard deviations of the medians and interquartile ranges of the total distances the movements travelled  $D_{\text{mouse}}^{I}$ were M{  $D_{\text{mouse}}^{I}$  } = 110 ± 31 count, IQR{  $D_{\text{mouse}}^{I}$  } = 190 ± 58 count. The means and standard deviations of the medians and interquartile ranges of the ratios of the net distance travelled  $\Delta_{\text{mouse}}^{I}$  to the total distance travelled were M{  $\Delta_{\text{mouse}}^{I} / D_{\text{mouse}}^{I}$  } = 0.91 ± 0.022, IQR{  $\Delta_{\text{mouse}}^{I} / D_{\text{mouse}}^{I}$  } = 0.10 ± 0.028. So, most of the identified mouse movements were relatively straight.

The sets of movements for each one-year period classified as class II movements had the following properties. The means and standard deviations (across all one-year periods) of the medians and interquartile ranges of  $(R^{II})^2$  and  $p^{II}$  values for each fit (in each one-year period) of the model given in Section 4.3.5 used to classify the movements were  $M\{(R^{II})^2\} = 0.43 \pm 0.068$  and  $IQR\{(R^{II})^2\} = 0.47 \pm 0.036$ , and  $M\{p^{II}\} < 0.001$  and  $IQR\{p^{II}\} = 0.010 \pm 0.012$ . The means and standard deviations of the medians and interquartile ranges of the movement times  $T_M^H$  were  $M\{T_M^H\} = 340 \pm 61$  ms,  $IQR\{T_M^H\} = 320 \pm 91$  ms. Again, the lower limit of applicability of Fitts' law occurs at  $D_{pointer}^H/W = 1/2$ , which, for these subjects, corresponds to a movement time of about 300 ms. So, the middle 50% of the identified mouse movements had reasonable time durations. We consider the distribution of the movement times further in Section 4.6. The means and standard deviations of the medians and interquartile ranges of the total distances the movements travelled  $D_{mouse}^H$  were  $M\{D_{mouse}^H\} = 92 \pm 19$  count,  $IQR\{D_{mouse}^H\} = 160 \pm 45$  count. The means and standard deviations of the medians of the medians of the medians and interquartile ranges of the total distance travelled were  $M\{\Delta_{mouse}^H/D_{mouse}^H\} = 0.94 \pm 0.010$ ,  $IQR\{\Delta_{mouse}^H/D_{mouse}^H\} = 0.077 \pm 0.040$ . Again, the movements are relatively straight.

We next used the two sets of identified mouse movements to estimate the motor performance for each one-year period as characterized by Fitts' law using Eq. (4.8). As Fitts' law is a description of rapid, targeted mouse movements, we limited the data sets to movements that we expected Fitts' law to describe, those that were (1) relatively straight  $(\Delta_{\text{mouse}} / D_{\text{mouse}} \ge 1/2)$  and (2) not too long in duration ( $T_M \le 4$  s, corresponding to five times the average expected duration of each movement in TMT or  $D_{\text{pointer}} / W \approx 7$ million). We characterized the motor performance for each one year period separately for class I and class II movements by estimating the parameters  $\tilde{a}$ , b, and  $\tilde{W} / \eta$  in Eq. (4.8) using OLS regression on all the remaining identified mouse movements of the given class. The means and standard deviations of the  $R^2$  and p values for the fits were  $(R^I)^2 = 0.66 \pm 0.079$  and  $p^I < 0.0001$ , and  $(R^{II})^2 = 0.54 \pm 0.092$  and  $p^{II} < 0.0001$ , and the means and the standard deviations of the estimated parameters were:

$$\tilde{a}^{I} = -710 \pm 180 \,\mathrm{ms},$$
  
 $b^{I} = 160 \pm 36 \,\mathrm{ms} \,/\,\mathrm{bit},$   
 $\tilde{W}^{I} \,/\,\eta = 14 \pm 2.5 \,\mathrm{count},$   
 $\tilde{a}^{II} = -800 \pm 360 \,\mathrm{ms},$   
 $b^{II} = 170 \pm 61 \,\mathrm{ms} \,/\,\mathrm{bit},$   
 $\tilde{W}^{II} \,/\,\eta = 17 \pm 32 \,\mathrm{count}.$ 
(4.18)

The average values for  $b^{I}$  and  $b^{II}$  were close to the independently measured value of 166 ms/bit for point-select methods of selecting icons. [110] We consider this further in Section 4.6. We also note that the mean values of  $\tilde{W}^{I}/\eta$  and  $\tilde{W}^{II}/\eta$  are close to the same value.

We confirmed the data supported the linear relationship between  $\tilde{a}$  and b expressed in Eq. (4.8) by performing a linear regression using all one-year periods with the appropriate model separately for class I and class II movements. The R<sup>2</sup> and p values for the fits were (R<sup>I</sup>)<sup>2</sup> = 0.97 and p<sup>I</sup> < 0.0001, and (R<sup>II</sup>)<sup>2</sup> = 0.99 and p<sup>II</sup> < 0.0001, and the resulting models were:

$$\tilde{a}^{I} = [96 \,\mathrm{ms}] - [4.9 \,\mathrm{bit}] b^{I},$$
  
 $\tilde{a}^{II} = [200 \,\mathrm{ms}] - [5.8 \,\mathrm{bit}] b^{II}.$ 
(4.19)

Examination of the 95% confidence intervals showed that all the fitted parameters in Eq. (4.19) were statistically significant. We used Eq. (4.19) to estimate the values of the motor parameters  $a^{I}$  and  $a^{II}$  that appear in Fitts' law, these were:

$$a^{I} = 96 \pm 31 \text{ms},$$
 (4.20)  
 $a^{II} = 200 \pm 42 \text{ms}.$ 

The average values for  $a^{\prime}$  and  $a^{\prime\prime}$  were close to the independently measured value of 230 ms for point-select methods of selecting icons. [110] We consider this further in Section 4.6.

Finally, we looked at the relationship of class I to class II movements as characterized by the parameters  $b^{I}$  and  $b^{II}$  by performing linear regression using a simple linear model with all one-year periods. The R<sup>2</sup> and p values for the fits were R<sup>2</sup> = 0.79 and p < 0.0001, and resulting model was:

$$b^{II} = -[77 \,\mathrm{ms}\,/\,\mathrm{bit}] + [1.5]b^{I}.$$
 (4.21)

Examination of the 95% confidence intervals showed that all the fitted parameters in Eq. (4.21) were statistically significant. By the measures we have looked at here, the two classes of mouse movements do not appear to differ appreciably. They have approximately the same distributions of time durations and straightness, and have about the same values for the motor parameters in Fitts' law with those motor parameters strongly correlated with each other.

We looked again at the set of 27 subjects that had more than one one-year period in the data set. We used these 74 one-year periods to estimate how well the estimates of  $\tilde{a}^{I}$ ,  $b^{I}$ ,  $\tilde{a}^{II}$ , and  $b^{II}$  in one one-year period estimated the value of the same quantity on the (chronologically) next one-year period in the set by the same subject. The correlations were, for  $\tilde{a}^{I}$ ,  $R^{2} = 0.82$  and p < 0.0001, for  $b^{I}$ ,  $R^{2} = 0.86$  and p < 0.0001, for  $\tilde{a}^{II}$ ,  $R^{2} = 0.93$  and p < 0.0001, and, for  $b^{II}$ ,  $R^{2} = 0.94$  and p < 0.0001.

#### 4.5.2 Measures of Cognitive Performance (Executive Function)

We have shown in Section 4.5.1 that the identified mouse movements look, on average, like what we expect mouse movements to looks like in terms of the time durations and

straightness of the movements, and the estimates of the motor parameters of Fitts' law. We next look at how well the identified mouse movements found using the technique presented in Section 4.3 capture subject-to-subject variability in executive function as characterized by TMT. We do this by using the recall and search performance proxy  $\rho$  that is also measured from the in-home MD data. We also look at the effect of including a second recall and search performance proxy, the DST score, to suggest directions for future work extracting recall and search performance measures from the MD data. This analysis continued the use of the restricted set of 102 one-year periods for the restricted cohort of 55 subjects.

We first looked at the ability of the motor parameters estimated from MD to estimate the TMT scores  $S^A$  and  $S^B$ . Using the superscript X as a placeholder for A or B indicating the test score being fit, and the superscript Y as a place holder for I or II indicating the class of mouse movements used in estimating the motor performance, we fit four models of the form:

$$S^{X} = c_{0}^{X,Y} + c_{1}^{X,Y} \tilde{a}^{Y} + c_{2}^{X,Y} b^{Y}.$$
(4.22)

The fits had R<sup>2</sup> and p values of  $(R^{A,I})^2 = 0.037$  and  $p^{A,I} = 0.15$ ,  $(R^{B,I})^2 = 0.071$  and  $p^{B,I} = 0.027$ ,  $(R^{A,II})^2 = 0.066$  and  $p^{A,II} = 0.034$ , and  $(R^{B,II})^2 = 0.16$  and  $p^{B,II} = 0.00020$ . Examination of the 95% confidence intervals showed that all the fitted parameters except  $c_1^{A,I}$  and  $c_2^{A,I}$  were statistically significant.

We next modified the model in Eq. (4.22) by including  $\rho$  (see Eq. (4.17)) as a proxy for recall and search performance. As  $\rho$  is measured using only in-home MD measurements, the resulting models generate estimated of TMT performance using only information from in-home MD. We characterized each one-year period by the mean value of  $\rho$  across all sessions in the one-year period; we denote this value by  $\langle \rho \rangle$ . The mean and standard deviation of  $\langle \rho \rangle$  across all one-year periods was:

$$\langle \rho \rangle = 1.8 \pm 0.044.$$
 (4.23)

We also looked at the relationship between  $\langle \rho \rangle$  and  $\tilde{a}$  and b using linear models; the R<sup>2</sup> and p values for the fits of the best performing models were (R<sup>I</sup>)<sup>2</sup> = 0.58 and p<sup>I</sup> < 0.0001, and (R<sup>II</sup>)<sup>2</sup> = 0.64 and p<sup>II</sup> < 0.0001, and the resulting models were:

$$\langle \rho \rangle = [2.0] - [5.0 \cdot 10^{-4} \,\mathrm{ms}] a^{I} - [8.2 \cdot 10^{-4} \,\mathrm{bit} \,/ \,\mathrm{ms}] b^{I},$$
  
 $\langle \rho \rangle = [2.0] - [3.0 \cdot 10^{-4} \,\mathrm{ms}] a^{II} - [5.4 \cdot 10^{-4} \,\mathrm{bit} \,/ \,\mathrm{ms}] b^{II}.$  (4.24)

Examination of the 95% confidence intervals showed that all the fitted parameters in Eq. (4.24) were statistically significant. In Section 4.4.1, we describe  $\rho$  as being the logarithm of the lowest speed at which meaningful, controlled MD occurs. There appears to be a strong relationship between this lowest speed and the motor parameters of Fitts' law. We consider this further in Section 4.6.

We looked again at the set of 27 subjects that had more than one one-year period in the data set. We used these 74 one-year periods to estimate how well the estimate of  $\rho$  in one one-year period estimated the value of the same quantity on the (chronologically) next one-year period in the set by the same subject; the correlations for  $\rho$  were  $R^2 = 0.82$  and p < 0.0001.

The connect-the-dots model tells us that estimates for cognitive performance using the scores on TMT-A and TMT-B should include the motor parameters a and b, or equivalently,  $\tilde{a}$  and b to account for the movement of the hand. Combining Eqs. (4.4)

and (4.9), we find that the form of the connect-the-dots model indicates that  $\tilde{a}$  and b should appear linearly in the estimators of either TMT score. In the absence of other measurements of recall and search performance we use average value  $\langle \rho \rangle$  for each one-year period and the score on DST for the one-year period as proxies for recall and search performance.

We fit four models of the form:

$$S^{X} = c_{0}^{X,Y} + c_{1}^{X,Y} \left\langle \rho \right\rangle + c_{2}^{X,Y} \tilde{a}^{Y} + c_{3}^{X,Y} b^{Y}.$$
(4.25)

The fits had R<sup>2</sup> and p values of  $(R^{A,I})^2 = 0.11$  and  $p^{A,I} = 0.0083$ ,  $(R^{B,I})^2 = 0.27$  and  $p^{B,I} < 0.0001$ ,  $(R^{A,II})^2 = 0.14$  and  $p^{A,II} = 0.0024$ , and  $(R^{B,II})^2 = 0.29$  and  $p^{B,II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters were statistically significant. The effect sizes associated with the addition of  $\langle \rho \rangle$  to the models in Eq. (4.22) were  $(f^{A,I})^2 = 0.068$ ,  $(f^{B,I})^2 = 0.27$ ,  $(f^{A,II})^2 = 0.12$ , and  $(f^{B,II})^2 = 0.18$ .

We chose the 102 one-year periods used to produce these models with the intent of identifying those periods corresponding to high cognitive performance by the subjects. We made this choice to produce a set of one-year periods where we could attribute the greatest amount of variation in TMT scores to differences in motor performance. If we assume that outliers to the model in Eq. (4.25) indicate one-year periods that nevertheless have poor cognitive performance, then trimming these one-year periods produces a better version of the intended data set. We trimmed the one-year periods for each model-case (i.e. A,I, B,I, A,II, or B,II) by looking at the residuals of the fit and trimming those one-year periods with residual values larger than expected in 95% of new observations for which the actual test score was larger than that predicted by the model. From the 102 one-year periods, we trimmed 3 one-year periods from the A,I model-case, 5 one-year

periods from the B,I model-case (there was one one-year period that was trimmed from both), 3 one-year periods from the A,II model-case, and 4 one-year periods from the B,II model-case (there was one one-year periods that was trimmed from both). For the A,I model-case, the mean and standard deviation of the actual test scores of the 3 trimmed one-year periods were  $64\pm4.9$  s and those of the model estimates were  $39\pm2.9$  s. For the B,I model-case, these were  $120\pm19$  s and  $73\pm15$  s, for the A,II model-case, these were  $62\pm7.8$  s and  $38\pm3.5$  s, and for the B,II model-case, these were  $125\pm14$  s and  $77\pm9.1$  s. The fits on the trimmed data sets had R<sup>2</sup> and p values of  $(R^{A,I})^2 = 0.070$  and  $p^{A,I} = 0.074$ ,  $(R^{B,I})^2 = 0.37$  and  $p^{B,I} < 0.0001$ ,  $(R^{A,II})^2 = 0.13$  and  $p^{A,II} = 0.0045$ , and  $(R^{B,II})^2 = 0.37$  and  $p^{B,II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters except  $c_0^{A,I}$  and  $c_0^{A,II}$  were statistically significant. In Figure 4.4, we compare the TMT scores estimated using class II movement data the actual TMT scores using the trimmed data sets.



**Figure 4.4.** Actual vs. estimated TMT scores across one-year periods for model in Eq. (4.25). Comparison of actual and estimated scores for TMT-A and TMT-B (estimated separately) for high performing subjects (MMSE = 30, ADL = 0). Scores were estimated using class II movement data and outliers with high test scores were trimmed (3 for TMT-A and 4 for TMT-B). The model for TMT-A fit with  $R^2 = 0.13$  and p = 0.0045, and the model for TMT-B fit with  $R^2 = 0.37$  and p < 0.0001.

TMT was designed so that the scores on TMT-A and TMT-B,  $S^A$  and  $S^B$  respectively, would be used together in the form  $S^B - S^A$  to arrive at an estimate of cognitive performance. [36] We next modified the model in Eq. (4.25) to produce an estimate of the difference in TMT scores  $S^B - S^A$ .

We fit two models of the form:

$$S^{B} - S^{A} = c_{0}^{Y} + c_{1}^{Y} \left\langle \rho \right\rangle + c_{2}^{Y} \tilde{a}^{Y} + c_{3}^{Y} b^{Y}.$$
(4.26)

The fits had  $R^2$  and p values of  $(R^I)^2 = 0.22$  and  $p^I < 0.0001$ , and  $(R^{II})^2 = 0.23$  and  $p^{II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters were statistically significant.

We trimmed the data for Eq. (4.26) using the same criteria we used to trim the data for Eq. (4.25). From the 102 one-year periods, we trimmed 3 one-year periods from the I

model-case, and 4 one-year periods from the II model-case (there were no one-year periods that were trimmed from both). For the I model-case, the mean and standard deviation of the actual values  $S^B - S^A$  of the 3 trimmed one-year periods were 93 ± 13 s and those of the model estimates were 43 ± 9.3 s. For the II model-case, these were 87 ± 15 s and 40 ± 6.2 s. The fits on the trimmed data sets had R<sup>2</sup> and p values of  $(R^I)^2 = 0.27$  and  $p^I < 0.0001$ , and  $(R^{II})^2 = 0.33$  and  $p^{II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters were statistically significant. In Figure 4.5, we compare the  $S^B - S^A$  values estimated using class II movement data the actual  $S^B - S^A$  values using the trimmed data sets.



**Figure 4.5.** Actual vs. estimated  $S^B - S^A$  values across one-year periods for model in Eq. (4.26). Comparison of actual and estimated values for  $S^B - S^A$  for high performing subjects (MMSE = 30, ADL = 0). Scores were estimated using class II movement data and DST scores with no outliers trimmed. The model fit with  $R^2 = 0.33$  and p < 0.0001.

We repeated the analysis of how well the score on one completion of TMT estimated the score on the next completion on TMT available that was done with 27 subject and 74 one-year periods above with the trimmed data sets. For the trimmed sets for the models using  $\tilde{a}^{I}$  and  $b^{I}$ , we found that, for TMT-A, we had 26 subjects, 71 one-year periods,  $R^{2} = 0.35$ , and p < 0.0001, for TMT-B, we had 25 subjects, 70 one-year periods,  $R^{2} = 0.43$ , and p < 0.0001. For the trimmed sets for the models using  $\tilde{a}^{II}$  and  $b^{II}$ , we found that, for TMT-A, we had 26 subjects, 71 one-year periods,  $R^{2} = 0.34$ , and p < 0.0001, for TMT-B, we had 25 subjects, 70 one-year periods,  $R^{2} = 0.42$ , and p < 0.0001.

We finally modified the model in Eq. (4.25) by including  $S^{DS}$  as a proxy for recall and search performance. As  $S^{DS}$  is measured in the clinic, the resulting model produces estimates that we cannot make using in-home MD measurements alone. However, the development of future techniques that estimate  $S^{DS}$  using in-home measurements would provide an estimate of TMT performance entirely using in-home measurements.

The observed mean and standard deviation of scores on the DST across one-year periods was  $S^{DS} = 45 \pm 7.1$ .  $S^{DS}$  did not show any statistically significant (p < 0.05) correlations with the motor parameters  $\tilde{a}^{I}$ ,  $b^{I}$ ,  $\tilde{a}^{II}$ , and  $b^{II}$  with  $R^{2} > 0.05$ .  $S^{DS}$  had a statistically significant (p < 0.05) correlation with  $\langle \rho \rangle$  with  $R^{2} > 0.072$ .

We looked at how well the score  $S^{DS}$  on DST estimates the TMT scores using models:

$$S^{X} = c_0^{X} + c_1^{X} S^{DS}. ag{4.27}$$

The fits had  $(R^A)^2 = 0.18$  and  $p^A < 0.0001$ ,  $(R^B)^2 = 0.27$  and  $p^B < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters were statistically significant.

We fit four models of the form:

$$S^{X} = c_{0}^{X,Y} + c_{1}^{X,Y} \left\langle \rho \right\rangle + c_{2}^{X,Y} S^{DS} + c_{3}^{X,Y} \tilde{a}^{Y} + c_{4}^{X,Y} b^{Y}.$$
(4.28)

We fit the models to the entire set of one-year periods rather than to random subsets. The fits had R<sup>2</sup> and p values of  $(R^{A,I})^2 = 0.23$  and  $p^{A,I} < 0.0001$ ,  $(R^{B,I})^2 = 0.42$  and  $p^{B,I} < 0.0001$ ,  $(R^{A,II})^2 = 0.25$  and  $p^{A,II} < 0.0001$ , and  $(R^{B,II})^2 = 0.44$  and  $p^{B,II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters were statistically significant except for the leading constant coefficients in the case of both TMT-A models. The effect sizes associated with the addition of  $S^{DS}$  to the models in Eq. (4.25) were  $(f^{A,I})^2 = 0.16$ ,  $(f^{B,I})^2 = 0.26$ ,  $(f^{A,II})^2 = 0.15$ , and  $(f^{B,II})^2 = 0.27$ .

We trimmed the data for Eq. (4.28) using the same criteria we used to trim the data for Eq. (4.25). From the 102 one-year periods, we trimmed 3 one-year periods from the A,I model-case, 4 one-year periods from the B,I model-case (there were no one-year periods that were trimmed from both), 2 one-year periods from the A,II model-case and 3 oneyear periods from the B,II model-case (there were no one-year periods that were trimmed from both). For the A,I model-case, the mean and standard deviation of the actual test scores of the 3 trimmed one-year periods were  $64 \pm 4.9$  s and those of the model estimates were  $42 \pm 1.0$  s. For the B,I model-case, these were  $117 \pm 23$  s and  $76 \pm 20$  s, for the A,II model-case, these were  $67 \pm 0.71$  s and  $42 \pm 2.0$  s, and for the B,II modelcase, these were  $120 \pm 21$  s and  $78 \pm 19$  s. The fits on the trimmed data sets had R<sup>2</sup> and p values of  $(R^{A,I})^2 = 0.18$  and  $p^{A,I} = 0.0010$ ,  $(R^{B,I})^2 = 0.48$  and  $p^{B,I} < 0.0001$ ,  $(R^{A,II})^2 = 0.18$ 0.22 and  $p^{A,II} = 0.00011$ , and  $(R^{B,II})^2 = 0.49$  and  $p^{B,II} < 0.0001$ . Examination of the 95% confidence intervals showed that all the fitted parameters except  $c_0^{A,I}$ ,  $c_1^{A,I}$ ,  $c_0^{A,I}$ , and  $c_1^{A,II}$  were statistically significant. In Figure 4.6, we compare the TMT scores estimated using class II movement data the actual TMT scores using the trimmed data sets.



**Figure 4.6.** Actual vs. estimated TMT scores across one-year periods for model in Eq. (4.28). Comparison of actual and estimated scores for TMT-A and TMT-B (estimated separately) for high performing subjects (MMSE = 30, ADL = 0). Scores were estimated using class II movement data and DST scores with no outliers trimmed. The model for TMT-A fit with  $R^2 = 0.22$  and p < 0.0001, and the model for TMT-B fit with  $R^2 = 0.49$  and p < 0.0001.

# 4.6 Discussion

Discussion of the modeling decisions made in the connect-the-dots model (Section 4.2.2), its application to TMT (Section 4.2.1), and empirical validation of the model using TMT is given in Chapter 3.

We have provided a technique for recording and storing large amounts of computer mouse-position data and identifying individual mouse movements using that data. We designed the techniques given in this paper for identifying individual mouse movements, and fitting a mouse movement model to each identified mouse movement to be computationally fast. Nevertheless, even the restricted data set of 55 subjects that we used for the bulk of the analysis in Section 4.5 was very large (102 years of mouse movements) and required substantial computer time to perform all the required calculations. The results that we have presented represent about the limit of what we

could reasonably do with the computational power we had available for this project. We expect that with more computational power available, we may obtain better results.

The observation that drives the analysis in this paper is that illustrated in Figure 4.1, namely the multimodal nature of the distribution of the logarithm of the mouse speed. Using this observation, we are able to impose a mathematical structure on the raw mouse data that appears, on average, to correspond to actual mouse movements as characterized by time duration and straightness of the movements, and the motor parameters of Fitts' law. We expect, however, that this simple procedure misclassifies a number of mouse movements. Likely, there are a number of cases where what should be a single mouse movement is divided into a number of smaller movements and where individual, consecutive movements have been combined into a single larger movement. Further refinements of the procedure given here may allow one to correct some of these misclassifications of movements. However, such refinements require a better understanding of how mouse movements are made, which we argue should proceed from a principled mathematical model of mouse movements.

We estimate the motor parameters a and b by fitting a set of observed movements to the empirical form of Fitts' law. The estimated values for the motor parameters of Fitts' law across the restricted set of subjects for the two classes of mouse movements were  $a^{l}$ = 96 ± 31 ms and  $b^{l}$  = 160 ± 31 ms/bit, and  $a^{ll}$  = 200 ± 42 ms and  $b^{ll}$  = 170 ± 61 ms/bit. We can compare these values to independently measured values for Fitts' law for several methods of using a computer mouse to select an icon: (1) a = 135 ms and b = 249 ms/bit for drag-select, (2) a = 230 ms and b = 166 ms/bit for point-select, and (3) a = 135 ms and b = 249 ms/bit for stroke-through. [110] The observed class I and class II movements appear, on average to more nearly resemble the point-select movements rather than drag-select or stroke through movements measured in [110].

Using the regression model in Eq. (4.25), the untrimmed data set was observed to fit with R<sup>2</sup> values of  $(R^{A,I})^2 = 0.11$ ,  $(R^{B,I})^2 = 0.27$ ,  $(R^{A,II})^2 = 0.14$ , and  $(R^{B,II})^2 = 0.29$ . The trimmed data was observed to fit with R<sup>2</sup> values of  $(R^{A,I})^2 = 0.070$ ,  $(R^{B,I})^2 = 0.37$ ,  $(R^{A,II})^2$ = 0.13, and  $(R^{B,II})^2 = 0.37$ . Using the regression model in Eq. (4.28), the untrimmed data set was observed to fit with R<sup>2</sup> of  $(R^{A,I})^2 = 0.23$ ,  $(R^{B,I})^2 = 0.42$ ,  $(R^{A,II})^2 = 0.25$ , and  $(R^{B,II})^2$ = 0.44. We can compare these R<sup>2</sup> values to the observed test-retest reliabilities for a cohort of normal adult controls with scores of 47 ± 25 s on TMT-A and 120 ± 86 s on TMT-B. For this cohort, the test-retest reliabilities were observed to be R<sup>2</sup> = 0.56 for TMT-A and R<sup>2</sup> = 0.72 for TMT-B. [28] The test-retest reliabilities provide a rough indication of how much noise there is inherent to TMT. We observe that our models for TMT-B account for roughly one-third to one-half the size of the total variation that one administration of TMT-B accounts for in another administration of TMT-B.

The simple model of computer mouse movements that we developed in Section 4.3.5 is only sufficient to analyze the mouse movements into the two classes of mouse movements that we have used. The model is only sufficient to determine whether the peak speed occurs in the first or second half of the mouse movement. For the high performing (MMSE = 30, ADL = 0) cohort of subject analyzed in this study, there appears to be no information gained by making the classification that is made in this study into class I and class II movements. We have retained the classification scheme in this study for two reasons related to future work. First, although this high performing cohort does not exhibit any significant difference in the measures values of the motor parameters a and b between the two classes, there is the possibility that other cohorts may show a difference. Second, the classification may prove useful in putting a classification on the idle intervals between movements.

If the movement classification scheme presented here proves inadequate, we can generalized the reasoning to produce a more sophisticated model of mouse movement both to generate better fits of the model to the identified mouse movements and to identify more classes of mouse movements. We can obtain a more general model by retaining the general form of  $\Phi(x, \dot{x}, \ddot{x})$  in Eq. (4.10). If we suppose that we can approximate  $\Phi(x, \dot{x}, \ddot{x})$  using a Taylor series to second-order, then we find:

$$\Phi(x, \dot{x}, \ddot{x}) \approx \phi_{00} + \begin{bmatrix} 0 \\ \phi_{02} \\ \phi_{03} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} x \\ \dot{x} \\ \ddot{x} \end{bmatrix} + \frac{1}{2} \begin{bmatrix} x \\ \dot{x} \\ \ddot{x} \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} \phi_{11} & \phi_{12} & \phi_{13} \\ 0 & \phi_{22} & \phi_{23} \\ 0 & 0 & \phi_{33} \end{bmatrix} \begin{bmatrix} x \\ \dot{x} \\ \ddot{x} \end{bmatrix}.$$
(4.29)

The zero appears in the place of  $\phi_{01}$  in Eq. (4.29) for the same reason that the comparable term in Eq. (4.12) is set to zero in Eq. (4.13). Taking the Lagrangian in Eq. (4.10) and using the form of  $\Phi(x, \dot{x}, \ddot{x})$  in Eq. (4.29) and plugging this into Eq. (4.14), we find the Euler-Lagrange equation for the more general model of mouse movements:

$$\ddot{\ddot{x}} + \phi_{33} \ddot{\ddot{x}} - (\phi_{22} - \phi_{13}) \ddot{x} + \phi_{11} x = 0.$$
(4.30)

There are now three parameters available to fit the identified mouse movements and to use to divide the identified mouse movements into classes. This model should also allow us to correct the classification of the raw mouse data into movements using the distribution of the logarithm of mouse speeds by providing a principled description of computer mouse movements. We have argued that the shorter idle intervals between consecutive mouse movements should contain information about a subject's recall and search performance. We did attempt to derive measures of recall and search performance using simple mathematical operations on the observed idle interval data (e.g., the median, the mode, or the minimum value of the time durations). For the operations we considered, the resulting measures did not appear to contain any information not already contained in the recall and search performance proxy  $\rho$  and, in all cases, did not perform as well as  $\rho$  for the purpose of estimating TMT scores.

# 4.7 Conclusion

In-home MD during everyday computer usage potentially provides a rich and easily unobtrusively and continuously monitored source of information about the cognitive performance of a subject. We have presented the foundation for a methodology of estimating cognitive function using in-home MD measured during everyday computer usage. We characterize cognitive function in terms of TMT and use a model of how a subject completes TMT to motivate the choice of what to measure from in-home MD. We have shown how to record and store computer mouse-position data reasonably compactly, how to identify individual mouse movements using the stored data, and how to measure motor performance from the identified mouse movements. Finally, we have outlined a program for making further measurements from in-home MD to estimate other, non-motor aspects of TMT performance.

# Chapter 5 – On the Relationship of Set-Switching, Movement Times, and Errors in Trail-Making Test Part B for Older Adults

# **5.0 Abstract**

Early detection of decline in cognitive performance in older adults allows early medical intervention to treat the decline. One way to detect performance changes earlier is though unobtrusive and ubiquitous monitoring of subject behavior in the home. For monitoring to be useful, we must relate the resulting measurements to clinically useful quantities. One way to establish these relationships is through models of the phenomena that we are measuring. Naturally, better models produce better clinical estimates of performance from the in-home measurements. Elsewhere, we have constructed the connect-the-dots model to describe the neuropsychological Trail-Making Test (TMT) and used it to derive estimates of TMT performance using measurements from a computer game and from everyday in-home computer usage. We revisit the model with the aim of improving it to produce better clinical estimates from the in-home measurements. We look at the data available from in-home measurements made using the computer game and every-day computer usage and show they are consistent with physical movements taking longer on TMT-B than on TMT-A. We make the connect-the-dots model consistent with this observation by allowing the subject to perform set-switching as a dual-task with movement on TMT-B causing movements to slow. We also note that
subjects make more errors on TMT-B than on TMT-A, and look at how errors relate to set-switching and movement.

# 5.1 Introduction

Early detection of declines in cognitive performance is an important component of providing proper medical treatment for maintaining the health and well-being of an older adult. One way of facilitating early detection is to make clinically useful measurements by from behaviors observed in the home. However, relating measurements that one can feasibly make in the home to established clinical values is often non-trivial. One approach to the problem of relating feasible in-home measurements to established clinical measurements is to use a mathematical model of the phenomenon that one is measuring. Naturally, one would like to use the most accurate model that one can as a more accurate model is expected to produce an estimation of established clinical measurements from in-home measurements.

Researchers have developed in-home measurements of cognitive performance for older adults using measurements of in-home activity. [71, 69, 70] Some in-home measurements of specific sorts or aspects of activity that have been developed include walking speed [Chapter 2, [79, 80]], time spent using a computer [113], typing speed [114], time away from home [115], and sleep [116, 117]. Researchers have also applied unobtrusive, in-home measurement techniques to fall risk assessment and fall detection for older adults. [84, 85] In each case, the quantity that is measured is a form of everyday behavior that one can measure in the background with the subject having to do as little as possible to facilitate the measurement process. The aim of in-home measurement techniques is to produce a clinically useful measurement within the practical limitations of what one can feasibly measure in the home. However, to be immediately useful to the clinician, we must relate the in-home measurements to established clinical values. Producing such a relationship is typically a non-trivial task.

We have put forward in-home monitoring techniques involving measurements made from the play of a computer game [Chapter 3], (computer) mouse dynamics [Chapter 4]. In both cases, we relate the measurements made from the techniques to an established clinical measurement, the Trail-Making Test (TMT), a standard pen-and-paper neuropsychological test. Also in both cases, we establish the relationship between the inhome measurement and TMT using a mathematical model (the connect-the-dots model) that describes how a subject completes TMT. The ability of the two techniques to use inhome measurements to estimate performance on TMT is as good as the connect-the-dots model used describe TMT and relate the in-home measurements to TMT, and otherwise understand the phenomena that we are measuring. One way to improve performance of the techniques is to improve the connect-the-dots model so that it more accurately describes the connect-the-dots task.

In the present paper, we look at the relationship of the in-home measurements from Chapters 3 and 4 to the connect-the-dots model and to TMT. We proceed by observing that the data in Chapter 4 appear to indicate a correlation between set-switching on TMT and motor speed. We argue that the simplest mechanism for producing such a correlation is a dual-tasking of set-switching and movement causing a slowing down of movement when set-switching is present. The connect-the-dots model, as developed in Chapter 3 assumes that the motor speed is the same whether or not set-switching is present. We look at modified forms of the connect-the-dots model using the data from [Chapter 3] and show that those data support the supposition that motor speed is slower when setswitching is present. Finally, we note that errors are more likely on TMT when setswitching is present. We look at data from Chapter 3 where the subjects make errors and begin developing a form of the connect-the-dots model that addresses errors on TMT.

#### 5.2 Background

The techniques put forward in Chapters 3 and 4 relate measurements made in the home to the established clinical measurement of the score on TMT using a mathematical model, the connect-the-dots model. Before looking the data from Chapters 3 and 4, we provide descriptions of TMT and the connect-the-dots model. We have slightly modified the form of the connect-the-dots model presented here from that given in Chapter 3. The model presented in Chapter 3 describes movement using a two-parameter form of Fitts' law. Here we replace that form of Fitts' law with a single parameter approximation that simplifies the analysis we use to relate set-switching to motor speed.

Section 5.2.1 provides a review of material appearing in Chapters 3 and 4

# 5.2.1 Trail-Making Test

TMT consists of two connect-the-dots tasks (TMT-A and TMT-B) which the subject completes by using a pen to draw a single line through a series of targets on a test page. Each test takes the form of a standard 8.5"x11" sheet of paper on which is printed 25 *targets* – small (12mm diameter), seemingly randomly scattered circles containing a label that may be a letter or a number. For TMT-A, the targets are labeled with numbers from 1 to 25, with one number labeling each circle, while in TMT-B targets are labeled with all

letters from A to L and all numbers from 1 to 13 with one letter or number labeling each circle. The individual completes the TMT-A by using a pen to draw a line (the "trail") through all the circles in ascending numerical order of the labels (i.e.  $(1,2,3,\ldots,24,25)$ ), while the TMT-B is completed by drawing the line through the circles in ascending alphanumeric order (i.e. '1,A,2,B,...,L,13'). We call the process of drawing a line from one target to the next in the sequence a *move*. The *score* on each part of the test is the time the subject needs to complete each task (there is no time limit). If the subject makes an error on the test (drawing the line through an incorrect target), the test administrator stops the subject as soon as the error is noted by the administrator and returns the subject to the last correctly selected target; timing of the subject is not stopped during this *error* recovery process. The total numbers of errors made on each part are included with the test score. The difference between TMT-A and TMT-B is the use of the numeric and alphanumeric sequence of labels respectively. The alphanumeric label sequences introduce the additional complication of *set-switching* in which the subject must not only recall the next element in an alphabetic or a numeric sequence, but must also switch sets between the alphabetic and numeric sequences.

# 5.2.2 Connect-the-Dots Model

The connect-the-dots model developed in Chapter 3 describes the process by which a subject makes each move by dividing the move in TMT into a sequence three additive stages. [103, 104] The stages are (1) *recall and updating*, in which the subject recalls the next target in the target sequence, (2) *search*, in which the subject searches for the recalled target, and (3) *motor*, in which the subject moves the mouse to the target according to Fitts' law. [105-107]

We characterize the recall and updating stage by the *recall time* required by the subject to recall the next target in the sequence. We treat the recall time as a random variable (RV)  $T_R$  with expected value  $\langle T_R \rangle = \tau_R$  (the *characteristic recall time*), and some standard deviation.

The search stage is characterized by the *search time*,  $T_s$ , required by the subject to locate on the test page the next target in the sequence after it has been recalled. We treat search as a series of discrete steps. [108] We assume that each step of search takes some fixed time  $\tau_s$  (the *characteristic search time*) and that the variation in search time is due to each search taking a variable number of steps; we treat the number of steps as a RV. At each step of search, the subject considers a randomly chosen unselected target on the test page; if that target is the desired target, the subject ceases search and moves on to the motor phase, otherwise the subject does not consider any unselected target twice during a particular search. In the v-th move during a part of TMT, there are 25 targets on the page, the subject is searching for the v-th target, and the subject has already found v-1 targets. Thus, there are 26-v targets unselected targets on the test page. The expected number of steps for the search is (26-v)/2, so the expected value for the search time for this target is given by:

$$\langle T_s \rangle = \left( \left( 26 - \nu \right) / 2 \right) \tau_s. \tag{5.1}$$

The distribution of  $T_s$  for a given value v is uniform on the discrete values  $\tau_s$ , ...,  $(26-v)\tau_s$ . The motor stage is characterized by the *motor time*  $T_M$ , a RV representing the time the subject takes to move the mouse to the next target. We assume that the speed of the movement is consistent with Fitts' law. Defining  $D_v$  to be center-to-center distance between the v-1-th and v-th targets, assuming a common width W for all targets, the expected time taken to move from the v-1-th to the v-th target is given by:

$$\langle T_M \rangle = a + b \log_2 (D_v / W + 1).$$
 (5.2)

The value  $\log_2(D/W+1)$  provides a measurement of the amount of information the subject must process to complete the movement as measured in bits; so the value *b* provides a measure of how much time the subject spends processing each bit of information.

The form of Fitts' law in Eq. (5.2) as the sum of two independent parameters presents technical difficulties for the present analysis. Eq. (5.2) is a standard form of Fitts' law, [105-107] and is the form used in Chapters 3 and 4. However, in the analysis in Chapter 3, we found we could not distinguish the parameters a and  $\tau_R$ , and instead could only measure  $\tau_R + a$ . In the present paper, we avoid this problem by using a single-parameter approximation to Eq. (5.2). For typical empirical values of a and b, over the range of typical distances and target widths for TMT, we can approximate Eq. (5.2) using the single parameter model:

$$\langle T_M \rangle \approx \tau_M \cdot \left( D_V / W \right)^{1/3}.$$
 (5.3)

Eq. (5.3) is a special case of Kvålseth's law for rapid, targeted movements. [143] In [110], the Fitts' law parameters a and b are measured for two classes of movements. The first class are movements made where the mouse button is not held down during the

movement; these are point-select movements and have parameter values a = 230 ms and b = 166 ms/bit. The second class are movements made where the mouse button *is* held down during the movement; these are drag-select and stroke-through movements and have parameter values a = 135 ms and b = 249 ms/bit. The choice of the power of 1/3 provides an average functional form of Fitts' law able to approximate the forms of Fitts' law for point-select, drag-select, and stroke-through movements reasonably well using a single free parameter  $\tau_M$ .

The expected total time  $\langle T \rangle$  required to complete TMT-A or TMT-B where no errors are made is the sum of the expected times spent making moving to each target in turn, that is:

$$\langle T \rangle \approx 25\tau_R + \kappa\tau_S + \xi (D_1, \dots, D_{25}, W) \tau_M,$$
  

$$\kappa = \sum_{\nu=1}^{25} ((26 - \nu)/2),$$
  

$$\xi (D_1, \dots, D_{25}, W) = \sum_{\nu=1}^{25} (D_\nu / W)^{1/3}.$$
(5.4)

We call  $\kappa$  the *search complexity*, and  $\xi(D_1, \dots, D_{25}, W)$  the *motor complexity*. We note that the estimate for expected total time required to complete the task is linear in the cognitive and motor parameters  $\tau_R$ ,  $\tau_S$ , and  $\tau_M$ .

Eq. (5.4) provides an estimate of the expected time required to complete one part of TMT without any errors using the cognitive and motor parameters  $\tau_R$ ,  $\tau_S$ , and  $\tau_M$ . We allow that different values may apply to each part of TMT and characterize subject performance on TMT as a whole by the set of parameter values  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ ,  $\tau_M^A$ , and  $\tau_M^B$  where the superscript indicates whether the parameter is for TMT-A or TMT-B. Whereas, when the subject makes no errors, TMT characterizes executive function by

two values, the two test scores, the connect-the-dots model characterizes executive function by up to eight values. The search complexity  $\kappa$  for TMT-A or TMT-B (they are the same value) can be calculated from the definition in Eq. (5.4); while the motor complexities  $\tilde{\xi}^A$ , and  $\tilde{\xi}^B$  for the remainder of the test after the first target has been selected can be calculated using the definition in Eq. (5.4) and direct measurement of the layout of the targets on the test page. Although we do not know where the hand begins before moving to the first target, we assume that the movement to the first target is typical for movements on the test and approximate the values  $\xi^A$  and  $\xi^B$  by  $\xi^A \approx$  $(25/24)\tilde{\xi}^A$  and  $\xi^B \approx (25/24)\tilde{\xi}^B$ .

In practice, we must modify the estimate in Eq. (5.4) to include times related to other effects present in the test-taking process. When we allow errors to occur on either part of TMT, a parameter  $\theta$  giving the expected time needed to recover from an error must also be included. In addition, the act of turning over the test page and moving to the first target adds some time  $T^0$  in addition to the recall, search and motor times to the total time. Using X as a placeholder for either of superscripts A or B, given a number of errors  $N^X$ , a subject's expected score on TMT-A or TMT-B (the expected time required to complete the test)  $S^X$  is:

$$S^{X} \approx T^{0} + 25\tau_{R}^{X} + \kappa\tau_{S}^{X} + (25/24)\tilde{\xi}^{X}\tau_{M}^{X} + \theta N^{X}.$$
(5.5)

# 5.3 In-Home Mouse Dynamics and TMT

In Chapter 4, we put forward a method for using measurements of in-home (computer) mouse dynamics (MD) made from everyday computer usage to estimate subject performance on TMT. That is, we observe how the subject moves the computer mouse

during normal, free computer usage and produce measurements based on how the subject moves the mouse. The measurements we make from MD are estimates of the parameters in Fitts' law in the form given in Eq. (5.2), and a further value that is related to both the lowest speed at which the subject makes meaningfully controlled movements of the mouse and the shortest interval with which the subject typically pauses between mouse movements. Using these measurements taken from MD, we found we were able to fit simple linear models using ordinary least squares to TMT-A with  $R^2 = 0.13$  and p = 0.0045, and to TMT-B with  $R^2 = 0.37$  and p < 0.0001. Thus, the MD measurements were able to account for twice the observed variation in TMT-B as the observed variation in TMT-A.

The difference between TMT-A and TMT-B is that TMT-B includes set-switching while TMT-A does not. As we have shown in Chapter 3 the lengths of the trails in TMT-A and TMT-B do not differ considerably. This suggests that the results in Chapter 4 indicate a relationship between the measurements made from in-home MD and set-switching during TMT-B. Although we might postulate a variety of mechanisms to account for such a relationship, a particularly simple mechanism is to suppose that set-switching for the move to a target during TMT-B is performed in part as a dual-task with the motor stage of the movement to the previous target (i.e. the set-switching and movement would be expected to cause the movement to be made slower than a comparable movement on TMT-A. The lengthened movement times on TMT-B and so

measures related to mouse movements would be able to account for a larger amount of the total variation in score on TMT-B than on TMT-A.

The remainder of this paper concerns itself with showing that the data from the computer game in Chapter 3 are consistent with a form of the connect-the-dots model in which the set-switching and motor stages are dual-tasked.

# 5.4 Single-Task and Dual-Task in TMT

We have allowed that the motor parameters  $\tau_M^A$  and  $\tau_M^B$  on TMT-A and TMT-B respectively, may differ between the two parts of the test. However, the motor tasks in TMT-A and TMT-B are the same – drawing a line from one target to the next. Since the motor tasks are the same, we would expect to find that  $\tau_M^A = \tau_M^B$  (which we assumed to be the case in Chapter 3). For the motor tasks to be carried out differently in the two cases, that is for  $\tau_M^A \neq \tau_M^B$  we need a model that makes the motor tasks in TMT-A and TMT-B differ in some way. One way to do this is to have the motor task occur in parallel with a cognitive task; that is have the motor task take place as part of a dual-task. We then allow that the subject carries out the motor task differently depending on the cognitive task. In this vein, poor performance on TMT has in fact been associated with altered dual-task prioritization in older adults. [47]

For movement a distance D to a target of width W, we define the motor parameter  $\tau_M$  to be the motor parameter value for the single-task case, that is, the case in which there is no cognitive dual-task with the motor task. The form of Fitts' law in this case is approximately:

$$T_M \approx \tau_M \cdot \left( D / W \right)^{1/3}. \tag{5.6}$$

The average velocity of the movement is approximately:

$$\langle v \rangle \approx D / T_M = W^{1/3} D^{2/3} / \tau_M.$$
 (5.7)

The motor time  $T_M$  changes when the subject performs a cognitive dual-task in parallel with the motor task. We assume that the dual-task lasts at least as long as the time taken to complete the motor task. We assume that the affect of the dual-task is to slow the movement by lowering the average movement speed. We model this by supposing that the dual-task lowers the average movement speed by a factor  $1/\gamma$ . We suppose that the value of the multiplicative factor  $\gamma$  is determined by the specific cognitive dual-task so that it takes about the same value across a population of subjects having a range of singletask motor times  $T_M$  for the same motor task. The average movement velocity when the cognitive dual-task is present is approximately:

$$\langle v \rangle \approx D/T_M \approx W^{1/3} D^{2/3} / \gamma \tau_M.$$
 (5.8)

Thus, when a cognitive dual-task is present, Eq. (5.7) becomes:

$$T_{M} \approx \gamma \tau_{M} \cdot \left( D / W \right)^{1/3}.$$
(5.9)

We assume that the effect of any cognitive dual-task associated with the motor stage of TMT-A or TMT-B is constant across the subject population in the sense given in deriving Eq. (5.9). We also assume that we know the motor parameter  $\tau_M$  for the single-task case for each subject in the population. Characterizing the presence of a cognitive dual-task in TMT-A or TMT-B by the parameters  $\gamma^A$  and  $\gamma^B$ , respectively, we may rewrite the TMT scores estimator in Eq. (5.5) as:

$$S^{X} \approx T^{0} + 25\tau_{R}^{X} + \kappa\tau_{S}^{X} + (25/24)\tilde{\xi}^{X}\gamma^{X}\tau_{M}^{X} + \theta N^{X}.$$

$$(5.10)$$

In Section 5.2.2, we have given a model for TMT that divides the process of moving from one target to the next into the three additive stages of (1) recall, (2) search, and (3) motor. In this model, we have merged the set-switching into the recall stage. If we separate set-switching out into its own stage we find we have still have the three additive stages model for TMT-A, but now have a four additive stages model for TMT-B of (1) set-switching, (2) recall, (3) search, and (4) motor. We would like a mechanism that causes the motor parameter  $\tau_M^A$  to take on a different value  $\tau_M^B$  through the presence of a cognitive dual-task during the motor stage. A simple mechanism is to have the setswitching stage take place as a cognitive dual-task with the motor stage whenever possible (i.e. for every set-switching stage except the first one). As there is no setswitching stage for TMT-A, we find the motor stage during TMT-A is a single-task and  $\gamma^A = 1$ . All motor stages in TMT-B, except the very last one, are dual-tasks. We illustrate the sequence of stages for all but the last move in TMT-B in Figure 5.1.



**Figure 5.1.** Stages of a TMT-B move. The process of selecting the next target in the sequence for TMT-B involves four sequential stages of (1) switching sets (2) recalling the next target, (3) serially searching for the next target by considering the available targets one after another, and (4) physically moving the mouse so that the cursor is on the target and clicking. The set-switching stage is made a secondary cognitive task with the motor stage whenever possible.

#### 5.5 Cognitive Performance and Scavenger Hunt Computer Game

In Chapter 3, we used Scavenger Hunt (SH), a simple computer game designed to mimic TMT, to estimate subjects' performance on TMT. We based the analysis relating SH to TMT on a slightly modified form of the connect-the-dots model presented in Section 5.2.2. In the treatment in Chapter 3, we assumed that the motor parameters during TMT-A and TMT-B were the same (i.e.  $\tau_M^A = \tau_M^B$ ). In the present treatment, we reconsider the data from Chapter 3 to see whether they provide any evidence of dual-tasking in TMT. In the interest of making the present treatment as self-contained as possible, we provide a brief review of the analysis in Chapter 3.

The subject plays SH by completing a series of self-contained rounds. Each round presents the subject with a board containing variable number of *markers* including both *targets* (typically about 4 to 10) and *distractors* that appear as circles containing a letter or number. The subject selects the targets in sequence by clicking each target in the order

given by the *search string* that appears at the top of the board. Search strings used in SH include: ascending or descending alphabetical sequences (e.g., 'A,B,C,...' or '...,C,B,A'), (ii) ascending or descending numeric sequences (e.g., '1,2,3,...' or '...,3,2,1'), and (iii) ascending or descending alphanumeric sequences (e.g., '1,A,2,B,...' or '...,B,2,A,1'). The search string and layout of targets and distractors on the SH board for each round is generated at random.

A marker in SH appears as a circle containing a single letter or number. The centers of the markers are arranged on the board in a 4x8 grid with a spacing of 80 pixels. Markers may have a diameter of 63 pixels or 77 pixels. The largest value D/W possible in SH as measured between the centers of targets is D/W = 11.

We make estimates of subject performance from play of SH using the connect-the-dots model given in Section 5.2.2. In the Sections 5.5.1 and 5.5.2, we provide a brief account of how we make the measurements. Section 5.5.3 provides a brief account of how TMT scores are estimated using measurements made from SH. The reader interested in the rationale behind the analysis in these sections can consult Chapter 3.

# 5.5.1 Estimating Motor Performance

We estimate SH motor performance using the subset of rounds with search string '1,2,3,4' and no distractors. These rounds make up about one in four SH rounds. We only use rounds in which the subject made no errors. We calculate motor performance by looking at the individual physical movements of the mouse made to select the four targets ignoring the movement to the first target. For each of the remaining three movements, we know the movement time  $t_i$ , distance moved  $D_i$ , and target width  $W_i$ , the number of targets in the search string n = 4, and the place of the target being searched for during

that move  $v_i \in \{2,3,4\}$ . We calculate the values  $c_0$ ,  $|c_1|$ , and  $|c_2|$  that minimize the total squared error given by:

$$\varepsilon = \sum_{i} \left( t_{i} - c_{0} - |c_{1}| \left( (5 - v_{i}) / 2 \right) - |c_{2}| \left( D_{i} / W_{i} \right)^{1/3} \right)^{2}.$$
(5.11)

We constrain the result so that  $c_1$  and  $c_2$  are non-negative; we retain the estimated motor parameter value  $\tau_M = |c_2|$ .

In [110], the Fitts' law parameters a and b are measured for two classes of movements. The first class are movements made where the mouse button is not held down during the movement; these are point-select movements and have parameter values a = 230 ms and b = 166 ms/bit. The second class are movements made where the mouse button *is* held down during the movement; these are drag-select and stroke-through movements and have parameter values a = 135 ms and b = 249 ms/bit. The motor performance estimates are made for SH for the range of values  $D/W = 3.5 \pm 1.7$ . Over this range, the estimated values for the characteristic motor time is  $\tau_M = 390$  ms for movements of the first class, and  $\tau_M = 440$  ms for movements of the second.

We may compare these values to the values we would estimate for the characteristic motor time  $\tau_M$  from typical moves in TMT. Typical values for TMT-A are D/W = 6.1 $\pm 2.6$ , and, over this range, the estimated values for the characteristic motor time is  $\tau_M =$ 380 ms for movements of the first class, and  $\tau_M = 450$  ms for movements of the second. Typical values for TMT-A are  $D/W = 8.2 \pm 3.6$ , and, over this range, the estimated values for the characteristic motor time is  $\tau_M = 380$  ms for movements of the first class, and  $\tau_M = 460$  ms for movements of the second. The largest value occurring on TMT is D/W = 17.

# 5.5.2 Estimating Recall and Search Performance

For SH, we characterize recall and search performance using the parameters  $\tau_R$ , and  $\tau_s$ . We do this using all the games played by the subject with search string belonging to some set. For each move we know the movement time  $t_i$ , distance moved  $D_i$ , and target width  $W_i$ , the number of targets in the search string  $n_i$ , the number of distractors in the search string  $d_i$ , and the place of the target  $v_i$  being searched for during that move. To accommodate the varying number of targets and the presence of distractors, we replace Eq. (5.1) in Section 5.2.2 with the expected search time:

$$\langle T_{s} \rangle = ((n_{i} - \nu_{i} + d_{i} + 1)/2)\tau_{s}.$$
 (5.12)

Given the value of the motor parameter *b* estimated in Section 5.5.1, we calculate the values  $c_0$ , and  $|c_1|$  that minimize the total squared error given by:

$$\varepsilon = \sum_{i} \left( \left\langle t_{i} \right\rangle - c_{0} - \left| c_{1} \right| \left( \left( n_{i} - \nu_{i} + d_{i} + 1 \right) / 2 \right) - \tau_{M} \cdot \left( D_{i} / W_{i} \right)^{1/3} \right)^{2}.$$
(5.13)

We constrain the result so that  $c_1$  is non-negative; we retain the estimated values  $\tau_R = c_0$ and  $\tau_S = |c_1|$ .

We make estimates using Eq. (5.13) separately for two sets of SH rounds. The first is the set of rounds with ascending or descending alphabetic or numeric search strings. The second is the set of rounds with ascending or descending alphanumeric search strings. This gives four recall and search parameters  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ , and  $\tau_S^B$ .

# 5.5.3 Estimating TMT Performance

The procedure in Sections 5.5.1 and 5.5.2 for measuring cognitive and motor performance using play of SH characterizes a subjects performance in terms of five cognitive and motor parameters  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and  $\tau_M$ . We assume the subject completes TMT using the related parameters  $\tau_R^{A'}$ ,  $\tau_R^{B'}$ ,  $\tau_S^{A'}$ ,  $\tau_S^{B'}$ , and  $\tau_M'$  so that the expected scores on TMT given numbers of errors  $N^X$  are given by Eq. (5.10) in the form:

$$S^{X} \approx T^{0} + 25\tau_{R}^{X'} + \kappa\tau_{S}^{X'} + (25/24)\tilde{\xi}^{X}\gamma^{X}\tau_{M}^{\ \prime} + \theta N^{X}.$$

$$(5.14)$$

We would like to estimate the TMT scores using the cognitive and motor parameter values measured in SH. Following Chapter 3, we assume that the parameters in TMT relate to those in SH by the set of transformations:

$$\tau_R^{X'} = \alpha + \beta \tau_R^X,$$
  

$$\tau_S^{X'} = \tau_S^X,$$
  

$$\tau_M^{X'} = \tau_M.$$
  
(5.15)

Eq. (5.15) assumes that the characteristic search time and the motor parameters for the single-task case are the same for TMT and SH. It allows that the characteristic recall time may differ between TMT and SH due to the subject being more careful in one than in the other. These assumptions are justified in more detail in Chapter 3. The final estimator for the expected TMT scores in terms of the cognitive and motor parameters estimated in SH is:

$$S^{X} \approx (T^{0} + 25\alpha) + 25\beta\tau_{R}^{X} + \kappa\tau_{S}^{X} + (25/24)\tilde{\xi}^{X}\gamma^{X}\tau_{M} + \theta N^{X}.$$

$$(5.16)$$

### 5.6 Errors in TMT and SH

As we show in Section 5.8, subjects make larger numbers of errors during TMT-B than during TMT-A. We have argued that the difference between TMT-A and TMT-B comes down to the presence of set-switching in TMT-B and its absence in TMT-A. This suggests that the increased rate of making errors during TMT-B is related to set-switching and the dual-tasking of the set-switching and motor stages. As we also show in Section 5.8, this observation holds in SH as well, with subjects making larger numbers of errors during SH rounds with ascending or descending alphanumeric search strings.

We treat errors as a probability for each move of moving to an incorrect target. When error probabilities are small, we can construct an approximate model in which the probability of making an error P during TMT or SH is the sum of the rates of making different types of errors, that is  $P \approx \sum_{i} P_{i}$ . Researchers have identified three types of TMT errors. [144, 145] The first type is *perseverative errors* in which the subject fails to alternate between the categories of letters and numbers in TMT-B. The second type is *sequential errors* in which the subject omits next element in the search string in either TMT-A or TMT-B. The third type is *proximity errors* in which the subject proceeds to an incorrect nearby target in either TMT-A or TMT-B. We assume that errors in setswitching appear only as additional perseverative errors (we consider this further in Section 5.9). Instead of using these three types of TMT errors, we divide the probability of making an error into two components. The first component is  $P_0$ , the baseline error probability (i.e. the probability of making an error on TMT-A, or the probability of making sequential or proximity errors). The second component is  $P_{sw}$ , the additional probability of error due set-switching (i.e. the difference of the probability of making an error on TMT-B and the probability of making an error on TMT-A). The baseline error probability represents the baseline probability of making sequential and proximity errors while the additional probability of error due to set-switching represents the presence of perseverative errors. So the probabilities  $P^A$  and  $P^B$  of making an error on TMT-A and TMT-B respectively, when the probability of making an error is small, are approximately:

$$P^{A} \approx P_{0},$$

$$P^{B} \approx P_{0} + P_{Sw}.$$
(5.17)

We provide a discussion of potential ways of performing a more detailed analysis of errors during SH in Chapter 3.

# **5.7 Notation Convention**

In the models we use in Section 5.8, we fit a single model to both TMT-A and TMT-B. However, we sectors of the model describing TMT-A and TMT-B may differ somewhat. In order to indicate clearly and compactly the way we calculate the model parameters in Section 5.8, and to make clear how the TMT-A and TMT-B sectors of the model differ, we adopt a convention for denoting the way we use ordinary least squares (OLS) regression to fit the scores on TMT. We assume we dependent variables y and Y, and independent variables  $x_n$  and  $X_n$ . From the dependent and independent variables, we calculate the parameters  $c_0, ..., c_n$  that minimize the square error:

$$\varepsilon = \sum_{i} \left( y_{i} - c_{0} - \sum_{n=1}^{N} c_{n} x_{n,i} \right)^{2} + \sum_{j} \left( Y_{j} - c_{0} - \sum_{n=1}^{N} c_{n} X_{n,j} \right)^{2}.$$
 (5.18)

We denote the problem of finding the parameters  $c_0, \ldots, c_n$  that minimize Eq. (5.18) by:

$$\begin{cases} y = c_0 + \sum_{n=1}^{N} c_n x_n \\ Y = c_0 + \sum_{n=1}^{N} c_n X_n. \end{cases}$$
(5.19)

For the models used in Section 5.8, the first equation in the formalism indicated in Eq. (5.19) corresponds to the TMT-A sector, and the second equation to the TMT-B sector of the model.

# **5.8 Empirical Study**

To provide a self-contained account of the analysis relevant to investigating evidence for dual-tasking in SH and TMT, we reproduce the portion of the analysis in Chapter 3 that is relevant to the present analysis. The reader can consult Chapter 3 for further information.

23 older adults (20 female and 3 male, average age  $81 \pm 6.8$  years, average level of education  $15 \pm 2.9$  years, MMSE =  $28 \pm 0.89$ , ADL =  $0.058 \pm 0.16$ ) participated in a one year study in which a set of computer games that included SH was placed into their homes. Subjects were given a battery of cognitive tests, including TMT, administered by trained clinical staff according to standard administrations procedures, at the beginning of the study, 6 months into the study and at the end of the study. The observed TMT-A and TMT-B scores and numbers of errors across all the tests taken by the subjects being included in this analysis and their standard deviations were  $S^A = 45 \pm 11$  s and  $N^A = 0.0073 \pm 0.14$ , and  $S^B = 100 \pm 28$  s and  $N^B = 1.0 \pm 0.64$ .

We looked at whether the measured SH motor performances indicated both singletasking and dual-tasking during SH. More specifically, we looked at whether the motor parameters estimated using only rounds with alphabetic or numeric search strings indicated significantly quicker motor performance than motor parameters estimated using only rounds with alphanumeric search strings.

For each subject, we estimated the motor parameter  $\tau_M$  in three ways. We measured a value  $\tau_M$  using Eq. (5.11) and data from rounds with search string '1,2,3,4' and no distractors (the method used to estimate the motor parameter  $\tau_M$  in Chapter 3). We measured a value  $\tau_M^A$  using a modified Eq. (5.11) that accounts for varying target numbers and the presence of distractors (see Eq. (5.13) for how distractors are accounted for) and data from rounds with alphabetic or numeric search strings (excluding rounds with search string '1,2,3,4' and no distractors). We measured a value  $\tau_M^B$  using a modified Eq. (5.11) that accounts for varying target numbers and the presence of distractors (see Eq. (5.13) for how distractors are accounted for) and data from rounds with alphabetic or numeric search strings (excluding rounds with search string '1,2,3,4' and no distractors). We measured a value  $\tau_M^B$  using a modified Eq. (5.11) that accounts for varying target numbers and the presence of distractors for varying target numbers and the presence of distractors of varying target numbers and the presence of distractors are accounted a value  $\tau_M^B$  using a modified Eq. (5.11) that accounts for varying target numbers and the presence of distractors and data from rounds with alphabetic or numeric search strings. The means and standard deviations of the estimated values were:

$$au_M = 670 \pm 240 \,\mathrm{ms},$$
 $au_M^A = 890 \pm 350 \,\mathrm{ms},$ 
 $au_M^B = 780 \pm 460 \,\mathrm{ms}.$ 
(5.20)

Although both the estimates  $\tau_M^A$  and  $\tau_M^B$  were somewhat larger than the estimate  $\tau_M$ , the estimates  $\tau_M^A$  and  $\tau_M^B$  gave comparable values. We consider these values and their relationship to the measurements further in Section 5.9.

An important difference between the SH rounds used to estimate  $\tau_M$  and those used to estimate  $\tau_M^A$  and  $\tau_M^B$  in Eq. (5.20) is that those used to estimate  $\tau_M$  contain no distractors, while those used to estimate  $\tau_M^A$  and  $\tau_M^B$  do contain distractors. To better understand the effect of distractors (if any), we restricted the analysis to SH rounds containing at most one distractor (limiting to zero distractors left too few subjects). We only performed the analysis on a subject if that subject had at least 15 moves total for rounds with alphabetic or numeric search strings and at most one distractor (excluding rounds with search string '1,2,3,4' and no distractors), and at least 15 moves total for rounds with alphanumeric search strings and at most one distractor.

This restriction left a cohort of 15 older adults (13 female and 2 male, average age 82 ± 6.6 years, average level of education 15 ± 2.9 years, MMSE = 29 ± 0.78, ADL = 0.044 ± 0.12). The observed TMT-A and TMT-B scores and numbers of errors across all the tests taken by the subjects being included in this analysis and their standard deviations were  $S^A = 45 \pm 12$  s and  $N^A = 0.044 \pm 0.12$ , and  $S^B = 100 \pm 23$  s and  $N^B = 0.98 \pm 0.62$ .

We estimated the values  $\tau_M$ ,  $\tau_M^A$ , and  $\tau_M^B$  in the same way as in Eq. (5.20); the means and standard deviations of the estimated values were:

In this case, all three estimates  $\tau_M$ ,  $\tau_M^A$ , and  $\tau_M^B$  give comparable values.

We observe in Eqs. (5.20) and (5.21) that the characteristic motor times  $\tau_M^B$  for movements during TMT-B-like rounds of SH are about the same as the characteristic motor times  $\tau_M^A$  for movements during TMT-A-like rounds (in fact, they are slightly smaller in both cases). This observation suggests that it is not the case during SH that moves during TMT-A-like rounds are made using single-tasking while moves made during TMT-B-like rounds are made using dual-tasking. Rather, moves in both types of rounds are made in approximately the same way. As set-switching only occurs in TMT-B-like rounds, this means that subjects are not dual-tasking set-switching with movement during play of SH.

We continued our analysis using the original set of 23 older adults.

SH characterizes a subject's cognitive performance with five measured parameters:  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and  $\tau_M$ . We calculated these using Eqs. (5.11) and (5.13). We estimated the motor parameter  $\tau_M$  as in Eqs. (5.20) and (5.21) using rounds with search string '1,2,3,4' and no distractors. The superscripts A and B denote estimates made using rounds with alphabetic and numeric, or alphanumeric search strings respectively. The means and the standard deviations of the parameter estimates across all subjects were:

$$au_{R}^{A} = 110 \pm 270 \,\mathrm{ms},$$
 $au_{R}^{B} = 410 \pm 330 \,\mathrm{ms},$ 
 $au_{S}^{A} = 96 \pm 45 \,\mathrm{ms},$ 
 $au_{S}^{B} = 110 \pm 40 \,\mathrm{ms},$ 
 $au_{M}^{B} = 670 \pm 240 \,\mathrm{ms}.$ 
 (5.22)

If we assume that the difference of values between  $\tau_R^A$  and  $\tau_R^B$  in Eq. (5.22) are due to set-switching, we estimate the average set-switching time to be 290 ms which compares well with independently measured values of about 200 ms. [111] We looked at the correlations of the parameter  $\tau_M$  with parameters measured in Eq. (5.22). We also looked at the correlation of the parameter  $\tau_M$  to the estimated characteristic set-switching time  $\tau_R^B - \tau_R^A$ . We summarize the correlations in Figure 5.2.

| Parameter correlated to $\tau_{M}$ | $\mathbb{R}^2$ | р      |
|------------------------------------|----------------|--------|
| $	au_R^A$                          | 0.18           | 0.044  |
| $	au_R^B$                          | 0.35           | < 0.01 |
| $	au_{S}^{A}$                      | 0.10           | 0.13   |
| $	au_{S}^{B}$                      | 0.056          | 0.28   |
| $\tau^B_R - \tau^A_R$              | 0.17           | 0.053  |

**Figure 5.2.** Correlations of  $\tau_M$  with other parameters measured from Scavenger Hunt.

The stronger correlations between  $\tau_M$ , and  $\tau_R^A$  and  $\tau_R^B$  were expected due to the method by which the estimates are made. We note that  $\tau_M$  had no statistically significant correlation with  $\tau_S^A$  and  $\tau_S^B$  and a weak (but statistically significant in the sense of p < 0.05) correlation with the estimated characteristic set-switching time  $\tau_R^B - \tau_R^A$ . The correlation between *b* and  $\tau_R^B - \tau_R^A$  is negative. We consider this further in Section 5.9.

We began by repeating the analysis in Chapter 3 but characterizing motor performance using Eq. (5.3) rather than Eq. (5.2). We used the characterization of subject cognitive performance given by SH in Eq. (5.22) to estimate TMT scores in the case where the motor stages of both TMT-A and TMT-B were assumed to be single tasks, that is  $\gamma^{A} =$  $\gamma^{B} = 1$ . We calculated the TMT scores estimator by finding the global parameters  $T^{0} + 25\alpha$ ,  $\beta$ , and  $\theta$  using OLS regression with the model (following the notation convention in Section 5.7):

$$\begin{cases} \left\langle S^{A} \right\rangle - \kappa \tau_{S}^{A} - (25/24) \tilde{\xi}^{A} \tau_{M} \\ = (T^{0} + 25\alpha) + 25\beta \tau_{R}^{A} + \theta \left\langle N^{A} \right\rangle \\ \left\langle S^{B} \right\rangle - \kappa \tau_{S}^{B} - (25/24) \tilde{\xi}^{B} \tau_{M} \\ = (T^{0} + 25\alpha) + 25\beta \tau_{R}^{B} + \theta \left\langle N^{B} \right\rangle. \end{cases}$$

$$(5.23)$$

We measured values for the motor complexities  $\tilde{\xi}^A$  and  $\tilde{\xi}^B$  from the TMT of  $\tilde{\xi}^A = 43$ and  $\tilde{\xi}^B = 47$ . The model fit with  $R^2 = 0.80$  and p < 0.0001. Inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$  and  $\theta$ were statistically significant, while that of  $T^0 + 25\alpha$  was not. The parameter values were:

$$T^{0} + 25\alpha = 1.9 \mathrm{s},$$
  
 $\beta = 1.7,$  (5.24)  
 $\theta = 32 \mathrm{s}.$ 

The model performance and estimates for  $\beta$  and  $\theta$  were comparable to those in Chapter 3.

As we observed in Chapter 3, there was a subject that produces a notable outlier in the data. Again, as in Chapter 3, removal of the outlier subject does not produce a notable change in the performance of the model or the estimated parameter values. We retain this subject for the analysis, as in Chapter 3. In the models analyzed here, removal of the outlier subject produced comparable or slightly better results than those produced while retaining the outlier.

As in Chapter 3, it was of interest to see how the model would perform in the case where subjects made no errors on TMT. We restricted the analysis to include only administrations of TMT in which both TMT-A and TMT-B had no errors. There were 16 subjects that had at least one error-free administration of TMT. We fitted a truncated version of the model in Eq. (5.23) lacking the terms in  $\theta N^A$  and  $\theta N^B$ , and used the average of all error-free TMT administrations for the test scores. The model fit with  $R^2 = 0.54$  and p < 0.0001, and estimated global parameter values of  $T^0 + 25\alpha = 12$  s and  $\beta = 2.7$ . Inspection of the confidence intervals for the coefficient estimates showed that the estimates of  $\beta$  and  $T^0 + 25\alpha$  were both statistically significant.

Having established that using Eq. (5.3) to characterize motor performance provides comparable results to those obtained when Eq. (5.2) is used (as is done in Chapter 3), we next looked at whether the available data provided evidence that movement times were longer during TMT-B than during TMT-A.

To estimate values for  $\gamma^{A}$  and  $\gamma^{B}$ , we weakened the model in Eq. (5.23) to allow us to fit values for  $\gamma^{A}$  and  $\gamma^{B}$  to the data. We calculated the new TMT scores estimator by finding the global parameters  $T^{0} + 25\alpha$ ,  $\beta$ ,  $\gamma^{A}$ ,  $\gamma^{B}$ , and  $\theta$  using OLS regression with the model:

$$\begin{cases} \left\langle S^{A} \right\rangle - \kappa \tau_{S}^{A} = \left(T^{0} + 25\alpha\right) + 25\beta \tau_{R}^{A} \\ + \left(25/24\right) \tilde{\xi}^{A} \gamma^{A} \tau_{M} + \theta \left\langle N^{A} \right\rangle \\ \left\langle S^{B} \right\rangle - \kappa \tau_{S}^{B} = \left(T^{0} + 25\alpha\right) + 25\beta \tau_{R}^{B} \\ + \left(25/24\right) \tilde{\xi}^{B} \gamma^{B} \tau_{M} + \theta \left\langle N^{B} \right\rangle. \end{cases}$$

$$(5.25)$$

This model fit with  $R^2 = 0.87$  and p < 0.0001; inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$ ,  $\theta$ ,  $\gamma^A$ , and  $\gamma^B$  were statistically significant, while that of  $(T^0 + 25\alpha)$  was not. The parameter values were:

$$T^{0} + 25\alpha = -9.1s,$$
  
 $\beta = 2.1,$   
 $\gamma^{A} = 1.2,$  (5.26)  
 $\gamma^{B} = 1.8,$   
 $\theta = 21s.$ 

We repeated this analysis 10000 times using the data for 12 randomly selected subjects each time. In this way, we estimated the means and standard deviations for parameters  $\gamma^{A}$  and  $\gamma^{B}$  of  $\gamma^{A} = 0.79 \pm 0.42$  and  $\gamma^{B} = 1.6 \pm 0.31$ . The smallest observed value for  $\gamma^{B} - \gamma^{A}$  was  $\gamma^{B} - \gamma^{A} = 0.15$ .

To reduce any effects due to the observed correlations between  $\tau_M$ , and  $\tau_R^A$  and  $\tau_R^B$ , we weakened the model in Eq. (5.25) to use the estimated characteristic set-switching time  $\tau_R^B - \tau_R^A$  rather than the recall times  $\tau_R^A$  and  $\tau_R^B$ . We calculated the new TMT scores estimator by finding the global parameters  $T^0 + 25\alpha$ ,  $\beta$ ,  $\gamma^A$ ,  $\gamma^B$ , and  $\theta$  using OLS regression with the model:

$$\begin{cases} \langle S^{A} \rangle - \kappa \tau_{S}^{A} = (T^{0} + 25\alpha) \\ + (25/24) \tilde{\xi}^{A} \gamma^{A} \tau_{M} + \theta \langle N^{A} \rangle \\ \langle S^{B} \rangle - \kappa \tau_{S}^{B} = (T^{0} + 25\alpha) + 25\beta \cdot (\tau_{R}^{B} - \tau_{R}^{A}) \\ + (25/24) \tilde{\xi}^{B} \gamma^{B} \tau_{M} + \theta \langle N^{B} \rangle. \end{cases}$$

$$(5.27)$$

This model fit with  $R^2 = 0.79$  and p < 0.0001; inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$ ,  $\theta$ , and  $\gamma^B$  were statistically significant, while those of  $(T^0 + 25\alpha)$  and  $\gamma^A$  were not. The parameter values were:

$$T^{0} + 25\alpha = 11s,$$
  
 $\beta = 2.5,$   
 $\gamma^{A} = 0.56,$  (5.28)  
 $\gamma^{B} = 1.1,$   
 $\theta = 20s.$ 

We repeated this analysis 10000 times using the data for 12 randomly selected subjects each time. In this way, we estimated the means and standard deviations for parameters  $\gamma^{A}$  and  $\gamma^{B}$  of  $\gamma^{A} = 0.26 \pm 0.29$  and  $\gamma^{B} = 0.99 \pm 0.22$ . The probability that  $\gamma^{B} - \gamma^{A}$  was less than zero was p = 0.011.

The model that we have developed in Chapter 3 is approximately Eq. (5.23) which is Eq. (5.23) with  $\gamma^A = \gamma^B = 1$ . In the models in Chapter 3 and Eq. (5.23), we assume that movements in both TMT-A and TMT-B are made as single-tasks and that the motor parameters that we measure in SH are measurements for single-task movements. The simplest improvement of the models in Chapter 3 and Eq. (5.23) is to include dualtasking during TMT-B is to maintain that the assumptions we have made in Chapter 3 and Eq. (5.23) still hold in the case of TMT-A, but allow  $\gamma^B$  to assume a larger value to reflect dual-tasking. We do this by allowing  $\gamma^B$  to be a parameters that we fit to the data while holding  $\gamma^A = 1$ . This provides us with a better model to use to estimate TMT performance from SH data to replace the one used in Chapter 3. The new model is able to include the dual-tasking effect with the addition of only one additional free parameter over the model used in Chapter 3, and constitutes the simplest model that can replace that used in Chapter 3 while including dual-tasking. We strengthened and simplified the weakened form the model in Eq. (5.23) by fixing the value  $\gamma^{A}$  to one. The TMT scores estimator was found by calculating the global parameters  $T^{0} + 25\alpha$ ,  $\beta$ ,  $\gamma^{B}$ , and  $\theta$  using OLS regression with the model:

$$\begin{cases} \left\langle S^{A} \right\rangle - \kappa \tau_{S}^{A} - \left(25 / 24\right) \tilde{\xi}^{A} \tau_{M} \\ = \left(T^{0} + 25\alpha\right) + 25\beta \tau_{R}^{A} + \theta \left\langle N^{A} \right\rangle \\ \left\langle S^{B} \right\rangle - \kappa \tau_{S}^{B} = \left(T^{0} + 25\alpha\right) \\ + 25\beta \tau_{R}^{B} + \left(25 / 24\right) \tilde{\xi}^{B} \gamma^{B} \tau_{M} + \theta \left\langle N^{B} \right\rangle. \end{cases}$$

$$(5.29)$$

The model fit with  $R^2 = 0.87$  and p < 0.0001; inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$ ,  $\theta$ , and  $\gamma^B$  were statistically significant, while that of  $(T^0 + 25\alpha)$  was not. The effect size associated with the addition of  $\gamma^B$  to the model in Eq. (5.23) was  $f^2 = 0.54$ . The parameter values were:

$$T^{0} + 25\alpha = -3.0 \,\mathrm{s},$$
  
 $\beta = 1.9,$   
 $\gamma^{B} = 1.6,$   
 $\theta = 21 \,\mathrm{s}.$   
(5.30)

The lower bound for the 95% confidence intervals for  $\gamma^{B}$  was 1.3. This model is a simple modification of the model in Chapter 3 to allow for dual-tasking during TMT-B. We illustrate the performance of this model in Figure 5.3.



**Figure 5.3.** Actual vs. estimated TMT scores across subjects. Each of the 23 subjects has two values shown, one for TMT-A and one for TMT-B, each representing the average of the three administrations of TMT. The model fit has  $R^2 = 0.87$  and p < 0.0001. A line with slope one passing through the origin is shown for reference.

As we did for the model in Eq. (5.23) above, we looked at how the model would perform in the case where subjects made no errors on TMT. We again restricted to the 16 subjects that had at least one error-free administration of TMT. We fitted a truncated version of the model in Eq. (5.29) lacking the terms in  $\theta N^A$  and  $\theta N^B$ , and used the average of all error-free TMT administrations for the test scores. The model fit with R<sup>2</sup> = 0.80 and p < 0.0001, and estimated global parameter values of  $T^0 + 25\alpha = -1.9$  s,  $\beta =$ 1.9, and  $\gamma^B = 1.9$ . Inspection of the confidence intervals for the coefficient estimates showed that the estimates of  $\beta$  and  $\gamma^B$  were statistically significant while  $T^0 + 25\alpha$  was not.

Again, to reduce any effects due to the observed correlations between  $\tau_M$ , and  $\tau_R^A$  and  $\tau_R^B$ , we weakened the model in Eq. (5.29) to use the estimated characteristic set-

switching time  $\tau_R^B - \tau_R^A$  rather than the recall times  $\tau_R^A$  and  $\tau_R^B$ . The new TMT scores estimator was found by calculating the global parameters  $T^0 + 25\alpha$ ,  $\beta$ ,  $\gamma^B$ , and  $\theta$  using OLS regression with the model:

$$\begin{cases} \langle S^{A} \rangle - \kappa \tau_{S}^{A} - (25/24) \tilde{\xi}^{A} \gamma^{A} \tau_{M} \\ = (T^{0} + 25\alpha) + \theta \langle N^{A} \rangle \\ \langle S^{B} \rangle - \kappa \tau_{S}^{B} = (T^{0} + 25\alpha) + 25\beta \cdot (\tau_{R}^{B} - \tau_{R}^{A}) \\ + (25/24) \tilde{\xi}^{B} \gamma^{B} \tau_{M} + \theta \langle N^{B} \rangle. \end{cases}$$

$$(5.31)$$

This model fit with  $R^2 = 0.78$  and p < 0.0001; inspection of the 95% confidence intervals for the coefficient estimates showed that the estimates of  $\beta$ ,  $\theta$ , and  $\gamma^B$  were statistically significant, while that of  $(T^0 + 25\alpha)$  was not. The parameter values were:

$$T^{0} + 25\alpha = -1.6s,$$
  
 $\beta = 2.9,$   
 $\gamma^{B} = 1.4,$   
 $\theta = 21s.$   
(5.32)

The lower bound for the 95% confidence intervals for  $\gamma^{B}$  was 1.0.

We finally looked at the rates at which the subjects made errors on TMT and SH. For SH rounds with ascending or descending alphabetic or numeric search strings (excluding rounds with search string '1,2,3,4' and no distractors) the range of the total number of errors observed for each subject across subjects was 0 to 70 with a median of 6.5. For SH rounds with ascending or descending alphanumeric search strings the range of errors across subjects was 0 to 127 with a median of 12. To compare the two rates of errors we looked at the probability of making an error each time a subject makes a movement to a target. For TMT, this was defined to be the total number of errors observed divided by

the total number of movements made, and denoted the probabilities for TMT-A and TMT-B by  $P^A$  and  $P^B$ . For SH, we used all the observed rounds of SH, including those in which errors were made, these probabilities, defined for SH rounds with ascending or descending alphabetic or numeric search strings (excluding '1,2,3,4' and no distractors) and ascending or descending alphanumeric search strings; these probabilities of error are denoted by  $p^A$  and  $p^B$  respectively. The observed means and standard deviations were:

$$P^{A} = 0.0028 \pm 0.0054,$$

$$P^{B} = 0.037 \pm 0.022,$$

$$p^{A} = 0.022 \pm 0.019,$$

$$p^{B} = 0.048 \pm 0.026.$$
(5.33)

The observed means and standard deviations values of the errors due to set-switching  $P_{Sw}$ 

$$= P^{B} - P^{A}$$
 and  $p_{Sw} = p^{B} - p^{A}$  were:

$$P_{Sw} = P^{B} - P^{A} = 0.034 \pm 0.023,$$
  

$$p_{Sw} = p^{B} - p^{A} = 0.026 \pm 0.021.$$
(5.34)

We did not find any statistically significant relationships among the probabilities  $P^A$ ,  $P^B$ ,  $p^A$ , and  $p^B$  except for the linear model:

$$p^{B} = [0.031] + [0.81] p^{A}.$$
(5.35)

The model in Eq. (5.35) fit the data with  $R^2 = 0.37$  and p < 0.0022. Inspection of the 95% confidence intervals indicted that both parameters were statistically significant. The coefficient value of 0.81 multiplying  $p^A$  is close to the value of 1.0 predicted by the model in Eq. (5.17).

### **5.9 Discussion**

The study presented in Section 5.8 is discussed more fully in Chapter 3. We restrict the present discussion to issues related to the observed correlation of the total set-switching and motor times, evidence for dual-tasking of set-switching with movement during TMT-B, and their relationship to the making of errors.

In Chapters 3 and 4, we presented two techniques for in-home monitoring of subject performance using data obtained from interactions with a computer. In both cases, we used the obtained data to estimate performance on TMT. The aim of the present paper has been to modify the model developed in Chapter 3 so that it would be able to account for the observations in Chapters 3 and 4. When we monitor subjects in the home, we sacrifice some ability to control what the subjects are doing in order to gather large quantities of data over a long period. We make up for the loss of some amount of control over what the subjects are doing by using a mathematical model to understand how the in-home measurements relate to other quantities of interest (e.g. cognitive performance as characterized by the score on TMT). As a result, it is important to produce the most accurate model of what we are measuring as we can.

We measure a value for the characteristic motor time of  $\tau_M = 730 \pm 270$  ms in Eq. (5.20) which differs from both the expected value of  $\tau_M = 390$  ms for point-select movements, and  $\tau_M = 440$  ms for drag-select and stroke-through movements. This result is comparable to the values of  $b = 300 \pm 110$  ms/bit for Eq. (5.2) measured in Chapter 3 using SH for in relation to the values of b = 166 ms/bit for point-select movements, and b = 249 ms/bit for drag-select and stroke-through movements measured in [110]. Using a comparable cohort of older adults in Chapter 4 we measured values of  $b = 160 \pm 36$ 

ms/bit and  $b = 170 \pm 61$  ms/bit using in-home measurement of MD. Point-select movements are made with the mouse button up, while drag-select and stroke through movements are made with the mouse button held down. The results in Chapter 4 suggest that most mouse movements during everyday computer usage resemble point-select movements, that is, the movements are made with the mouse button held up. However, the estimates of b or  $\tau_M$  made using SH appear to be nearer to drag-select or strokethrough movements made with the mouse button held down. We note that when the mouse button is held down, it is also the case that there is a greater force by the hand pushing the mouse onto the table; thus the change in Fitts' law may not be due as much to the button being held down as the greater force associated with holding the button down. One interpretation for the larger values for b or  $\tau_M$  is that the subjects push on the mouse with a greater force when playing SH than during everyday computer usage. This may relate to a general phenomenon of people pushing harder on controls when playing in video games.

In the SH computer game, Eqs. (5.20) and (5.21) indicate there is no change in the motor parameter  $\tau_M$  between rounds with alphabetic or numeric search strings, and rounds with alphanumeric search strings. This suggests that rounds of SH with alphabetic or numeric search strings are completed according to the simple additive stages model with three stages: (1) recall, (2) search, and (3) motor; while rounds with alphanumeric search strings are completed according to the simple four stages model: (1) set-switching, (2) recall, (3) search, and (4) motor.

In Chapter 4, we observed that purely motor speed measurements were able fit TMT-A with  $R^2 = 0.13$  and p = 0.0045, and TMT-B with  $R^2 = 0.37$  and p < 0.0001. So the motor

speed measurements were able to account for about twice as much of the total variation in TMT-B as in TMT-A. We have suggested that a simple explanation for this observation is that the subjects spends a longer time in the motor stage during TMT-B than during TMT-A. In the present paper, we wanted to see if the data from SH indicated a similar effect. We fit several models using the data for SH. The first model we fit (Eq. (5.23)) is a slightly modified version of the model used in Chapter 3 where we have replaced the original form of Fitts' law (Eq. (5.2) (used in Chapter 3) with an approximation (Eq. (5.3)). We have shown that this slightly modified model gives results comparable to the model that we actually used in Chapter 3. The next model we fit (Eq. (5.25) took the first model and allowed the total times spent in the motor stages to differ in TMT-A and TMT-B by allowing the model to fit the parameters  $\gamma^A$  and  $\gamma^B$  giving values  $\gamma^{A} = 1.2$  and  $\gamma^{B} = 1.8$ . Thus, we have observed that the SH data appear to support a longer motor stage in TMT-B than in TMT-A in agreement with the observations in Chapter 4. In the fourth model (Eq. (5.29), we looked at the simplest extension of the first model that allows for the effect of more time being spent in the motor stage in TMT-B over TMT-A and have observed that it provides a significant improvement over the model used in Chapter 3. We also looked at two additional models, the third and fifth models, to determine whether the fit values  $\gamma^A$  and  $\gamma^B$  could be related to the presence of the recall time values. We found that the use of the characteristic recall times  $\tau_R^X$  and the estimated characteristic set-switching times  $\tau_R^B - \tau_R^A$ both show that the data are consistent with a longer motor stage in TMT-B than in TMT-A.

In the models in Eqs. (5.25), (5.27), (5.29), and (5.31), we included the characteristic recall times  $\tau_R^X$  or the estimated characteristic set-switching times  $\tau_R^B - \tau_R^A$  and allowed them to be fit to the models. The effect of the inclusion of these parameters in this way is that we are allowing additional time for set-switching beyond the time devoted to set-switching during dual-tasking with the motor stage. Physically, we can interpret these models as saying that the set-switching stage for a move begins when the movement to the target of the previous move begins and continues for some time after the movement is finished. The apparent extension of set-switching beyond the period spend dual-tasking with the motor stage in the motor times due to dual-tasking with set-switching would indicate that both the total time spent making movements and the total time spent in-between movements increase from TMT-A to TMT-B.

When set-switching is included in the connect-the-dots model explicitly as an additive stage, the recall and update stage becomes the process of recalling the next target in the sequence given that the current target in the sequence is in mind. In TMT-A, the current target in the sequence is in mind when the subject has just moved to the current target, while in TMT-B the subject has to switch-sets and recall the most recent target selected in the set before the current target in the sequence is in mind. The set-switching stage in the model contains all the processing that a subject does in a TMT-B move prior to the recall and update stage. In a simple formulation of the connect-the-dots model, we can suppose that the characteristic times associated with these more narrowly defined recall and update stages are the same for TMT-A and TMT-B and that the set-switching stage captures extra time needed to complete TMT-B.
Following [144, 145], we have divided errors on TMT into three types: perseverative, sequential, and proximity. The three types of errors correspond well with different stages of the connect-the-dots model. Perseverative errors, in which the subject fails to switch sets between letters and numbers, appear to be errors made during the set-switching stage. Sequential errors, in which the subject omits the next element in the search string in either TMT-A or TMT-B, appear to be errors made during the recall stage. Proximity errors, in which the subject proceeds to an incorrect nearby target, appear to be errors made during the search stage. We expect that the subject can reduce the probability of making an error during a particular stage by being more careful during that stage; that is by spending more time performing that stage. In relation to this, we note that in the fits of the models in Eqs. (5.25), (5.27), (5.29), and (5.31), that the time spent performing recall approximately doubles. This is consistent with the subjects spending more time performing the recall stage during TMT than during SH. We expect that subjects are less likely to make sequential errors during TMT than during SH, possibly accounting for the observed lower rates of errors during TMT than during SH. In the models in Eqs. (5.25), (5.27), (5.29), and (5.31), we have assumed that the time devoted to search does not change from SH to TMT. If we assume that changes in the times devoted to the various stages during TMT are in part chosen to lower the associated rates of error, this assumption is consistent with the assumption that the probability of making a proximity error is already very low in SH.

The probabilities of perseverative errors are comparable for TMT-B and SH. Although the data appear to indicate a higher probability of probability of perseverative errors on TMT-B the standard deviations of the two estimated probabilities of perseverative errors overlap. As it appears that subjects spend more time performing the set-switching stage in TMT-B than in SH, we would expect the probability of perseverative errors to be lower during TMT-B. If, in fact, it were to turn out that the probabilities of perseverative errors is higher during TMT-B, then it would seem that the longer time spent performing the set-switching stage, or the dual-tasking of the set-switching and motor stages increases the error probability. In this case, there would appear to be no benefit to lengthening the set-switching stage or dual-tasking set-switching with motor during TMT-B, so it would be of interest to determine why the subjects would do this. Alternatively, such a result may indicate that a model that assumes that set-switching introduces the possibility of perseverative errors, but does not affect other types of errors is incorrect.

We postulated that older adults dual-task the set-switching and motor stages during TMT-B as a way of modifying the connect-the-dots model to make the observations in Chapters 3 and 4 consistent with each other. We have argued that it is the simplest mechanism to account for the apparent relationship between set-switching and motor. One should be able to confirm or disprove this postulate simply by observing a cohort of older adults complete TMT and measuring the time they spend physically drawing the trail. This could be done by recording older adults completing the test or by using a computer-based annotation technique like that in [102]. It is still possible that there is no difference in the physical movement times between TMT-A and TMT-B. If this turns out to be the case, the measurements in Chapter 4 and the analysis presented in Section 5.8 that quantifies a relationship between the set-switching and motor stages still hold and a

more complicated model that dual-tasking would have to be found to account for the observations.

#### 5.10 Conclusion

In Chapters 3 and 4, we used a mathematical model describing how a subject completes TMT to make measurements using a simple computer game [Chapter 3], and in-home MD [Chapter 4] to provide estimates of subject performance on TMT. The performance of these techniques follows from both the quality of the measurements and the quality of the model used. We expect improvements in the model to yield better estimates of TMT performance. Data from Chapters 3 and 4 point to an improvement in the model in the form of allowing dual-tasking of set-switching and movement on TMT-B, although this remains to be confirmed by direct observation. By generating better models of the physical phenomenon being measured for clinical purposes we point the way to more clinically useful measurements of the physical phenomenon, not only in-home, but also in future clinical instruments which may improve upon TMT.

#### 5.11 Appendices

#### 5.11.1 Appendix 1

We can analyze the model in Eq. (5.29) further using the same set of 23 subjects analyzed in Section 5.8. Rather than fitting the model in Eq. (5.29) once for all 23 subjects, we fit it 23 times, once for each subset of 22 subjects. These models fit with a mean and standard deviation of  $R^2$  of  $R^2 = 0.93 \pm 0.0040$  and in all cases p < 0.0001. The means and standard deviations of the estimated global parameter values were:

$$T^{0} + 25\alpha = -3.0 \pm 0.54 \,\mathrm{s},$$
  

$$\beta = 1.9 \pm 0.079,$$
  

$$\gamma^{B} = 1.6 \pm 0.024,$$
  

$$\theta = 21 \pm 1.4 \,\mathrm{s}.$$
  
(5.36)

The global parameter values estimated for each of the 23 models in Eq. (5.36) are close to those estimated in the single model using all 23 subjects in Eq. (5.30). We arrive at the same model using any set of 22 subjects that we do when using all 23 subjects.

We define  $S^x$  to be the actual average TMT scores and  $\tilde{S}^x$  to be the estimated average TMT scores for the individual subject left out of the estimation of the global parameter values in Eq. (5.36). The means and standard deviations of the resulting differences in the actual and estimated average TMT scores  $S^x - \tilde{S}^x$  using the estimated global parameter values for each of the 23 subject for TMT-A and TMT-B were:

$$S^{A} - \tilde{S}^{A} = 0.10 \pm 12 \mathrm{s},$$
  
 $S^{B} - \tilde{S}^{B} = 0.017 \pm 18 \mathrm{s}.$ 
(5.37)

We can compare the values in Eq. (5.37) to the means and standard deviations of the errors when all 23 subjects are fit using a single model:

$$S^{A} - \tilde{S}^{A} = -0.15 \pm 11 \mathrm{s},$$
  
 $S^{B} - \tilde{S}^{B} = 0.15 \pm 15 \mathrm{s}.$ 
(5.38)

The errors in the case where the model is fit using 22 subjects and the estimated global parameters were used to estimate TMT performance for the remaining subject were comparable to the errors when all 23 subjects were used to estimate the global parameter values.

### 5.11.2 Appendix 2

We can reconsider the analysis in Chapter 3 and this paper using a simpler analysis of SH rounds. In the simpler analysis, we use only average observed movement times for

the TMT-A-like and TMT-B-like rounds used for analysis in Chapter 3 and in this paper rather than characteristic recall, search, and motor times. We denote the average movement times by  $t^{A}$  and  $t^{B}$ . Using the notation in this paper, the model we fit is:

$$\begin{cases} \left\langle S^{A} \right\rangle = 25c_{0} + 25c_{1}t^{A} + \theta \left\langle N^{A} \right\rangle \\ \left\langle S^{B} \right\rangle = 25c_{0} + 25c_{1}t^{B} + \theta \left\langle N^{B} \right\rangle. \end{cases}$$
(5.39)

This model fit with  $R^2 = 0.82$  and p < 0.0001; inspection of the 95% confidence intervals for the coefficient estimates showed that all the parameter values were statistically significant. The parameter values were:

$$c_0 = -1.7 \,\mathrm{s},$$
  
 $c_1 = 2.5,$  (5.40)  
 $\theta = 28 \,\mathrm{s}.$ 

So, denoting the average movement times during TMT by  $T^A$  and  $T^B$ , we find that SH movements transform into TMT movements according to the relationship:

$$T^{X} = [-1.7s] + 2.5t^{X}.$$
(5.41)

We observe that the model in Eq. (5.39) performs somewhat better than the model in Eq. (5.23) where  $\gamma^{A} = \gamma^{B} = 1$  that had  $R^{2} = 0.80$  and p < 0.0001, but somewhat worse than the model in Eq. (5.29) where  $\gamma^{A} = 1$  but  $\gamma^{B}$  is free that had  $R^{2} = 0.87$  and p < 0.0001.

As we did for the models in Eqs. (5.23) and (5.29) above, we can look at how the model would perform in the case where subjects made no errors on TMT. We again restricted to the 16 subjects that had at least one error-free administration of TMT. We fit a truncated version of the model in Eq. (5.29) lacking the terms in  $\theta N^A$  and  $\theta N^B$ , and used the average of all error-free TMT administrations for the test scores. The model fit with  $R^2 = 0.61$  and p < 0.0001, and estimated global parameter values of  $c_0 = -2.5$  s and

 $c_1 = 3.3$ . Inspection of the confidence intervals for the coefficient estimates showed that all the estimated parameter values were statistically significant. We again observe that the model in Eq. (5.39) performs somewhat better than the model in Eq. (5.23) where  $\gamma^A$  $= \gamma^B = 1$  that had R<sup>2</sup> = 0.54 and p < 0.0001. However, in this case, the model in Eq. (5.29) where  $\gamma^A = 1$  but  $\gamma^B$  is free that had R<sup>2</sup> = 0.80 and p < 0.0001 performs substantially better. In this case, SH movements transform into TMT movements according to the relationship:

$$T^{X} = [-2.5s] + 3.3t^{X}.$$
(5.42)

The model using the average observed movement times in SH,  $t^A$  and  $t^B$ , performs better than the model in Eq. (5.23) for the purpose of estimating the TMT scores because the model in  $t^A$  and  $t^B$ , implicitly includes the contribution of dual-tasking during the motor stage while this is left out in the model in Eq. (5.23). However, the more detailed analysis developed in Chapter 3 and in this paper performs better for the purpose of estimating the TMT scores when dual-tasking is allowed as is the case in the model in Eq. (5.29). We can make sense of the forms of the transformations in Eqs. (5.41) and (5.42) in terms of the additive stages model we developed in Chapter 3 and this paper. The models in Eqs. (5.41) and (5.42) transform the averge SH movement times, while the models developed in Chapter 3 and this paper transform each of the stages separately to produce the estimated movement times in TMT. In terms of the additive stages model, we can identify the large negative value  $c_0$  as a necessary correction for stages that have been transformed to too large of a value using the coefficient  $c_1$ . The aim of developing the connect-the-dots model in Chapter 3 and of modifying it in this paper to make it consistent with the observations that we have made in Chapter 4 is not to simply produce an in-home estimator of TMT performance as represented entirely by the scores on TMT. The aim is to produce a physical model that can describe the entire behavior of the subject completing the connect-the-dots task as it appears in a number of situations (in particular in SH and TMT). This physical model takes the form of an additive stages model where we can monitor the performance of each stage separately. By monitoring the subject's performance of each of the stages through measurements of the cognitive and motor parameters  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and  $\tau_M$ , we aim to arrive at a more detailed description of the subject than is available through simply monitoring the average movement times in SH,  $t^A$  and  $t^B$ . Nevertheless, it is important to show that we can produce an estimate of performance on TMT, as TMT is an established clinical instrument. Conclusion

# **Chapter 6 – Conclusion and Future Directions**

# 6.1 Conclusion

In this thesis, we have looked at three techniques for user-centered, in-home monitoring of cognitive performance using available cost-effective technologies. We have made measurements of the walking speed in the home using an inexpensive system of passive infrared motion sensors placed in the home. We have made measurements from which we can estimate TMT performance in the home by observing the subject's interactions with a personal computer using both a computer game and everyday, real-life computer usage. We have validated the in-home monitoring techniques using two established clinical performance measures: (1) the walking speed, and (2) the Trail-Making Test. Finally, we have shown that the data obtained by observing in-home computer usage support a model of TMT performance in which the subject dual-tasks set-switching and movement during TMT-B.

### 6.1.1 Part I – In-Home Monitoring of Walking Speed

In Part I, we have developed a technique for monitoring of walking speed in the home using a system of passive infrared motion sensors.

In Chapter 2, we have presented a technique for measuring the walking speed v inhome using a system of PIR sensors. As we have shown in Chapter 1, the walking speed has been shown to be an indicator of a variety of aspects of health and performance in older adults including: survival, [16, 17] health, [13] risk of future hospitalization, [18] risk of cognitive impairment [20] and dementia, [21] and disability. [22, 23] The technique that we have presented in Chapter 2 allows one to monitor the walking speed v continuously as an indicator of these various health and performance issues.

# 6.1.2 Part II – In-Home Monitoring of Trail-Making Test Performance

In Part II, we have developed two techniques of monitoring a subject's cognitive performance in the home using the subject's interactions with a computer. We have also shown that the data obtained using these techniques support a model in which the subject dual-tasks set-switching and movement during TMT-B.

In Chapter 3, we have presented a technique for measuring the cognitive and motor parameters  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and  $b^{SH}$  in-home using the SH computer game. We have facilitated the analysis of SH game-play by constructing a computational model – the connect-the-dots model – to describe the process of playing SH. We have also constructed estimators for performance on TMT using these cognitive and motor performance parameters. As we have shown in Chapter 1, the TMT performance has been shown to be clinically useful: in the diagnosis of many neurological conditions, [33] for measuring of cognitive and set-switching ability, [36] and for measuring executive function. [37] The technique that we have presented in Chapter 3 allows one to monitor the TMT performance continuously and thus monitor cognitive and set-shifting ability, and executive function continuously. It also allows for a more detailed diagnosis of observed problems by providing a more detailed analysis of a subject's performance than does TMT due to the separate measurement of each of the cognitive and motor parameters. In Chapter 4, we have presented a technique for measuring the cognitive and motor parameters  $\langle \rho \rangle$ ,  $\tilde{a}^{I}$ ,  $b^{MD,I}$ ,  $\tilde{a}^{II}$ , and  $b^{MD,II}$  using in-home MD. We have also constructed estimators for performance on TMT using these cognitive and motor performance parameters. We have observed that the cognitive and performance parameters perform poorly in estimating performance on TMT-A but much better in estimating performance on TMT-B. The technique that we have presented in Chapter 4 allows one to monitor continually those aspects of TMT-B performance that are related closely to the control and movement of the hand.

In Chapter 5, we have argued that the reason that the parameters measured in Chapter 4 perform better in estimating TMT-B than in estimating TMT-A is due to the subjects dual-tasking set-switching with movement during TMT-B causing movements to be made more slowly. Following this observation, and after reviewing the data used in Chapter 3, we have modified the connect-the-dots model developed in Chapter 3 to allow the dual-tasking of the set-switching and motor stages during TMT-B. This has allowed the description of TMT made using the connect-the-dots model to capture the observations in both Chapters 3 and 4.

# 6.1.3 Connect-the-Dots Model

In Chapter 3, we have constructed the connect-the-dots model to understand how an older adults plays the SH computer game, and have used the model to generate measures of performance form play of SH. This initial version of the connect-the-dots model presented in Chapter 3 consists of three additive stages: (1) recall and update, (2) search, and (3) motor. In Chapter 4, we have used the connect-the-dots model as a model to understand in-home MD qualitatively. In Chapter 5, we have observed that the data we

have appear to support the idea that mouse movements take longer during TMT-B than during TMT-A. To account for this observation, we have modified in Chapter 5 the connect-the-dots model by dividing the recall and update stage into two stages: (1) setswitching stage, and (2) recall and update stage. We have accounted for the longer movement times during TMT-B by allowing the subject to perform set-switching as a dual-task with the motor stage causing the subject to make the movements more slowly. We now summarize the final form of the connect-the-dots model developed over Chapters 3, 4, and 5 in detail.

We characterize a general connect-the-dots task as having *n* targets and *d* distractors arrayed on a page or computer screen. The targets are labeled using elements from some number of sets, and a trail must be drawn through the targets in an indicated order. In the sequence of targets, the transitions between some targets may involve set-switching, and the transitions between other might not. Elements from each set appear in a fixed ascending or descending sequence, we are simply allowing an interweaving of sets in the overall target sequence. The distances between consecutive targets in the sequence are values  $D_{\nu}$  and the widths of the targets are (possibly varying) values  $W_{\nu}$ . The task begins with the subject's pencil, pen, or computer pointer at a known position.

We describe the process of drawing the trail from target v-1 to target v. The process of selecting target v requires the subject complete a sequence of four additive stages: (1) set-switching, (2) recall and update, (3) search, and (4) motor. We allow the subject to dual-task the set-switching stage with the motor stage from the previous movement to a target if the subject likes, bearing in mind that dual-tasking of set-switching and movement is optional. The subject must perform set-switching if the labels of targets v-1 and v come from different sets. A subject switches sets when the subject changes focus from the set to which the label of target v-1 belongs to the set to which the label of target v belongs. This process takes some amount of time with an expected value of  $\tau_{Sw}$ . The quantity  $\tau_{Sw}$ is the *characteristic set-switching time*. We address this process with more detail in Chapter 5. The expected set-switching time is:

$$\left\langle T_{Sw}\right\rangle = \tau_{Sw}.\tag{6.1}$$

This stage may be dual-tasked with the motor stage of the previous movement. This dual-tasking is optional.

In Chapter 5, we attribute perseverative errors on TMT (i.e. errors in which the subject fails to alternate between the categories of letters and numbers in TMT-B) [144, 145] to a failure to carry out the set-switching stage correctly.

Once the subject is working with the correct set, the subject must determine the next element in the sequence for that set. This process takes some amount of time with an expected value of  $\tau_R$ . The quantity  $\tau_R$  is the *characteristic recall and update time*. We address this process with more detail in Chapters 3 and 5. In Chapter 3, we folded the set-switching stage into the recall and update stage, while in Chapter 5, we separated the two as we do here. The expected recall and update time is:

$$\langle T_R \rangle = \tau_R. \tag{6.2}$$

In Chapter 5, we attribute sequential errors on TMT (i.e. errors in which the next element in the search string in either TMT-A or TMT-B is omitted) [144, 145] to a failure to carry out the recall and update stage correctly.

Having decided what the next label in the target sequence is, the subject next searches for the next target. We assume the subject has perfect memory of previously selected targets but otherwise no memory of the positions of the remaining targets or the distractors. The subject searches the remaining targets and distractors at random until the next target is found. Each step of the search takes a time  $\tau_s$ . The quantity  $\tau_s$  is the *characteristic search time*. We address this process with more detail in Chapter 3. The expected search time is:

$$\langle T_s \rangle = \left( \left( n - \nu + d + 1 \right) / 2 \right) \tau_s. \tag{6.3}$$

In Chapter 5, we attribute proximity errors on TMT (i.e. errors in which the subject proceeds to an incorrect nearby target in either TMT-A or TMT-B) [144, 145] to a failure to carry out the search stage correctly.

Once the subject finds the next target, the subject must move from the present position to the target just located. The subject does this according to Fitts' law and is characterized by the Fitts' law motor parameters a and b. The expected motor time is:

$$\langle T_M \rangle = a + b \log_2 \left( D_v / W_v + 1 \right)$$
  
  $\approx \tau_M \cdot \left( D_v / W_v \right)^{1/3}.$  (6.4)

The two-parameters form of Fitts' law in Eq. (6.4) is the form used in Chapters 3 and 4, while the single-parameter approximation to Eq. (6.4) is an approximation that we have developed in Chapter 5.

We assume that, during the motor stage, the subject always moves to the target decided upon by the end of the previous stages. Motor stage errors involve failing to end the movement within the region of the target that the subject is moving to by either undershooting or overshooting the target during the movement.

### 6.1.4 Using the Three Techniques Together

Taken together, the three in-home monitoring techniques presented in Chapters 2, 3, and 4 provide a characterization of a subject's performance in terms of eleven measured parameters: v,  $\tau_R^A + a$ ,  $\tau_R^B + a$ ,  $\tau_S^A$ ,  $\tau_S^B$ ,  $\langle \rho \rangle$ ,  $\tilde{a}^I$ ,  $\tilde{a}^{II}$ ,  $b^{SH}$ ,  $b^{MD,I}$ , and  $b^{MD,II}$ . Although we have not done so in this thesis, we may modify the techniques used in Chapters 3 and 4 to use the approximate single-parameter form of Fitts' law given in Eq. (6.4). This form approximates the two parameter form of Fitts' law in the motor parameters a and busing a single parameter form with motor parameter  $\tau_M$ ., Using this approximation, the cognitive and motor parameters measured in the technique in Chapter 3 become:  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ , and  $\tau_M^{SH}$ . The cognitive and motor parameters measured in the technique in Chapter 4 become:  $\langle \rho \rangle$ ,  $\tau_M^{MD,I}$ , and  $\tau_M^{MD,II}$ . Using this approximation, a subject monitored using all three techniques would now be characterized by nine measured parameters: v,  $\tau_R^A$ ,  $\tau_R^B$ ,  $\tau_S^A$ ,  $\tau_S^B$ ,  $\langle \rho \rangle$ ,  $\tau_M^{SH}$ ,  $\tau_M^{MD,I}$ , and  $\tau_M^{MD,II}$ .

When the three techniques presented in Chapters 2, 3, 4, and 5 are all used to monitor a single subject, we are able to combine the measurements made using the three techniques (either 11 or 9 cognitive and motor parameters) to produce a single picture of the subject's performance. The picture that we build is one that combines information about how subject walks with information about how a subject completes the connect-the-dots task. This information takes the form of how quickly the subject: (1) walks, (2) performs set-switching, (3) recalls a symbol in a sequence, (4) performs a visual search, (5) moves a pencil, pen, or computer mouse while playing SH, and (6) moves a pencil, pen, or computer mouse when going about everyday activities. Thus, the picture we build of the

subject is one of the times the subject takes to carry out a variety of behaviors. The aim of monitoring, then, is to detect significant slowing in the performance of one or more of the monitored behaviors. By noting which activities have slowed significantly and which have not, we may arrive at a detailed description of the subject's decline in performance.

### **6.2 Future Directions**

We conclude by going over a number of possible future directions for extending the research that we have presented in this thesis.

#### 6.2.1 Measure Walking Cadence In-Home

Grieve's law [146] provides an empirical relationship between the walking speed and the step-size of the form:

$$s \sim v^{0.42}$$
. (6.5)

We can augment the technique for measuring walking speed by instrumenting shoes worn by the subjects in the home to measure when the subject takes a step. The measurement of the intervals between steps provides an in-home measurement of the walking cadence (the number of steps in a given time period). If these measurements of when the subject takes a step are time-aligned with the sensor data from the PIR sensor line, then we can obtain the cadences for walks through the sensor line. The walking speed v, step-size s, and cadence c must satisfy the relationship:

$$s = v / c. \tag{6.6}$$

Thus, by combining measurements of walking speed and walking cadence, we may obtain estimates of the parameters in Grieve's law.

### 6.2.2 Augment Computer Game Suite

We can augment the computer game suite that includes SH with further measures that we expect to relate to the various additive stages of the connect-the-dots model of SH. First, we can include a computer game designed to mimic DST, and create a model that relates performance on the computer game version of DST with performance on DST itself. As DST involves the remembering of an assignment of digits to symbols, we expect a model of DST to bear some relationship to the recall stage of the connect-thedots model used to describe SH and TMT. Second, we can use a webcam to try to detect saccades of the eye during play of SH. By counting the saccades, we can produce an estimate of the actual number of steps in search during SH, and use this information to produce a better empirical estimate of the characteristic search time. Finally, one could add a further game that amounts to a straightforward implementation of a laboratory measurement of Fitts' law. The game would consist of targets of various widths appearing at various positions on the computer screen which the subject must click on. This would provide a more solid characterization of motor performance than that estimated using SH play, and would allow principled investigation of changes of motor performance during play of SH due to either dual-tasking or the presence of distractors.

#### 6.2.3 Analyze Errors Made During SH

There is reason to believe that the numbers of errors made during TMT contain important information about subject performance, and we expect the probabilities with which a subject makes errors during SH contain related information. We have limited the use of the available data on errors made during SH to a quick summary in Chapter 5, where we have also related an existing classification of the types of TMT errors to stages in the connect-the-dots model. In Chapter 3, we have provided a number of avenues for exploring the SH error data.

# 6.2.4 Reduce In-Home MD Processing Time

We have designed the technique of measuring performance using in-home MD presented in Chapter 4 to work with the limited computational resources that were available for the analysis there. Improvements in the way the data are stored or are processed would lower the time required to analyze the data, as would the availability of more computational resources. Lowered processing times would allow more sophisticated analyses of the data to be completed in reasonable amounts of time. Potential improvements to the analysis given reduced processing times are given in Chapter 4.

#### 6.2.5 Measure Dual-Tasking During Mouse Movement

For mouse-movements, we expect there to be a slowdown in the movement when dualtasking is present. If we can classify mouse-movement times into those that take longer or shorter times given all other relevant factors (such as the distance moved for a mousemovement), then we can potentially generate an estimate of how much a subject's movements slow when dual-tasking. In cases where we can assume the cognitive part of the dual-task is the same across subjects, then we try to differentiate subjects by the degree to which movement slows down when dual-tasking.

### 6.2.6 Optimal Control Model of Mouse Movements

In Chapter 4, we have used a simple optimal control model of computer mouse movements to divide the observed mouse movements into two classes. For the high performing older adults examined in that Chapter, the classification does not appear to be useful. However, the same sort of analysis used in Chapter 4 may prove useful for a lower performing set of older adults. The two classes may also prove useful for classifying pauses between mouse movements. In the discussion in Chapter 4, we have proposed using a more general model of mouse movements to provide a more detailed description of mouse movements in order to arrive at a better division of mouse movements into classes. We would like to use such a better classification of mouse movements to provide a more detailed description of how a subject uses a computer. However, as we have characterized mouse movements in Chapters 3, 4, and 5 using Fitts' law, it is important that we know how to relate a detailed model of individual movements to Fitts' law.

We can set out a brief outline of the optimal control formalism in which the model in Chapter 4 and generalizations of it fit. The movement problem is one of finding the movement orbit x(t) that minimizes some cost functional  $J[\cdot]$  subject to some set of initial and final conditions. In the case of mouse movements, the initial and final conditions are that the movement begin at rest at some position -D and end at rest at the origin. Mathematically, we express the optimal control problem as:

$$\begin{split} I[x] &= \int_{0}^{T} L(x, \dot{x}, \ddot{x}, \ddot{x}) dt = \int_{0}^{T} \left( \frac{1}{2} \ddot{x}^{2} - \Phi(x, \dot{x}, \ddot{x}) \right) dt, \\ &\qquad x(0) = -D, \quad x(T) = 0, \\ &\qquad \dot{x}(0) = v_{0}, \quad \dot{x}(T) = v_{T}, \\ &\qquad \ddot{x}(0) = a_{0}, \quad \ddot{x}(T) = a_{T}. \end{split}$$
(6.7)

The integrand  $L(x, \dot{x}, \ddot{x}, \ddot{x})$  of the cost functional is the Lagrangian of the system. We minimize the cost functional in Eq. (6.7) by setting the first variation to zero. This is done by solving the following differential equation (see e.g., [140] Part I Chapter 4):

$$\frac{d^3}{dt^3}\frac{\partial L}{\partial \ddot{x}} - \frac{d^2}{dt^2}\frac{\partial L}{\partial \ddot{x}} + \frac{d}{dt}\frac{\partial L}{\partial \dot{x}} - \frac{\partial L}{\partial x} = 0.$$
(6.8)

The orbit x(t) of the movement is the solution to Eq. (6.8) that satisfies the initial and final conditions in Eq. (6.7). Eqs. (6.7) and (6.8) provide the Lagrangian formulation of the optimal control problem. There is an alternative formulation of the optimal control problem called the Hamiltonian formulation. This formulation is a mathematically dual and equivalent formulation to the Lagrangian formulation. We construct the Hamiltonian formulation by first defining the *generalized coordinates* vector Q(t):

$$Q^{\mathrm{T}} = \begin{bmatrix} q_1 & q_2 & q_3 \end{bmatrix} = \begin{bmatrix} x & \dot{x} & \ddot{x} \end{bmatrix},$$
  
$$\dot{Q} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 0 \end{bmatrix} Q + \begin{bmatrix} 0 \\ 0 \\ u \end{bmatrix}.$$
 (6.9)

The quantity u(t) is the control of the system and, in the case of Eq. (6.9), satisfies  $u = \ddot{x}$ . We next define the *generalized momenta* vector P(t):

$$\boldsymbol{P}^{\mathrm{T}} = \begin{bmatrix} \boldsymbol{p}_1 & \boldsymbol{p}_2 & \boldsymbol{p}_3 \end{bmatrix}. \tag{6.10}$$

The vectors Q(t) and P(t) together provide a complete description of the state of the system at any time. We calculate the Hamiltonian by taking the Legendre transform (see e.g., [147]) of Eq. (6.7). The Legendre transform of the Lagrangian has the form:

$$H = P^{\mathrm{T}}\dot{Q} - L. \tag{6.11}$$

The optimal orbit x(t) satisfies the equations (see e.g., [148] Chapter 5):

$$\dot{Q} = \frac{\partial H}{\partial P}, \quad \dot{P} = -\frac{\partial H}{\partial Q}, \quad \frac{\partial H}{\partial u} = 0.$$
 (6.12)

The first two equations in Eq. (6.12) are Hamilton's equations. Hamilton's equations provide a description of how the state of the system evolves in time. Given an expression for the Hamiltonian, we may use Eq. (6.12) to calculate the form of the generalized momenta of the system:

$$P = \begin{bmatrix} p_1 \\ p_2 \\ p_3 \end{bmatrix} = \begin{bmatrix} \ddot{x} \\ -\ddot{x} \\ \ddot{x} \end{bmatrix} - \begin{bmatrix} 0 & 1 & -d / dt \\ 0 & 0 & 1 \\ 0 & 0 & 0 \end{bmatrix} \frac{\partial \Phi}{\partial X}.$$
 (6.13)

Finally, Pontryagin's minimum principle (see e.g., [148] Chapter 5) tells us that the Hamiltonian takes a constant value when calculated along an optimal orbit.

$$H = \Psi. \tag{6.14}$$

We call the constant of the motion  $\Psi$  the *generalized energy*.

In the discussion of Chapter 4, we suggested a more general formulation of the optimal control movement of mouse movements of the form:

$$L = \frac{1}{2}\ddot{x}^{2} + \phi_{00} + \begin{bmatrix} 0\\\phi_{02}\\\phi_{03} \end{bmatrix}^{T} \begin{bmatrix} x\\\dot{x}\\\dot{x} \end{bmatrix} - \frac{1}{2} \begin{bmatrix} x\\\dot{x}\\\dot{x} \end{bmatrix}^{T} \begin{bmatrix} \phi_{11} & \phi_{12} & \phi_{13}\\0 & \phi_{22} & \phi_{23}\\0 & 0 & \phi_{33} \end{bmatrix} \begin{bmatrix} x\\\dot{x}\\\dot{x} \end{bmatrix},$$

$$x(0) = -D, \quad x(T) = 0,$$

$$\dot{x}(0) = 0, \quad \dot{x}(T) = 0,$$

$$\ddot{x}(0) = 0, \quad \ddot{x}(T) = 0,$$

$$\ddot{x}(0) = 0, \quad \ddot{x}(T) = 0.$$
(6.15)

We calculate the equation of motion for the system in Eq. (6.15) using Eq. (6.8) and find:

$$\ddot{x} + \phi_{33} \ddot{x} - (\phi_{22} - \phi_{13}) \ddot{x} + \phi_{11} x = 0.$$
(6.16)

The Hamiltonian in this case is:

$$H = \frac{1}{2} P^{\mathrm{T}} \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 1 \end{bmatrix} P + P^{\mathrm{T}} \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 0 \end{bmatrix} Q + \frac{1}{2} Q^{\mathrm{T}} \begin{bmatrix} \phi_{11} & 0 & \phi_{13} \\ 0 & \phi_{22} & 0 \\ 0 & 0 & \phi_{33} \end{bmatrix} Q = \Psi.$$
(6.17)

Evaluating the Hamiltonian at time zero, we find:

$$\frac{1}{2}\ddot{x}(0)^2 + \frac{1}{2}\phi_{11}D^2 = \Psi.$$
(6.18)

The initial value of the jerk  $\ddot{x}(0)$  is approximately (see Section 6.3):

$$\ddot{x}(0) \approx \left(60 - 24\phi_{33}T^2 - \frac{\phi_{11}T^6}{120} + \frac{54\phi_{33}^2T^4}{5} + \frac{9\phi_{11}\phi_{33}T^8}{20}\right)\frac{D}{T^3}.$$
(6.19)

Combining Eqs. (6.18) and (6.19), we find the distance moved D and the movement time T satisfy the polynomial equation:

$$1800 - 1440\phi_{33}T^{2} + 816\phi_{33}^{2}T^{4} - \frac{\Psi}{D^{2}}T^{6} + 27\phi_{11}\phi_{33}T^{8} + \frac{\phi_{11}^{2}}{28800}T^{12} \approx 0.$$
(6.20)

We expect Eq. (6.20) to be consistent with Fitts' law. To do this, we must include the target width W in Eq. (6.20) so that every appearance of D becomes D/W. We can do this by requiring the generalized energy  $\Psi$  to satisfy:

$$\Psi = \frac{1800}{\tau_M^{-6}} W^2.$$
(6.21)

Combining Eqs. (6.20) and (6.21) in the special case where  $\phi_{11} \approx 0$  and  $\phi_{33} \approx 0$ , we arrive at an approximate form of Fitts' law:

$$T \approx \tau_M \cdot \left(\frac{D}{W}\right)^{1/3}.$$
 (6.22)

Eq. (6.22) is a special case of Kvålseth law for rapid, targeted movements. [143] Eq. (6.22) provides a reasonably good approximation to Fitts' law in the form we have used in Chapters 3 and 4 and is the approximate form of Fitts' law that we have used in Chapter 5 and Eq. (6.4). We observe that models of the form given in Eq. (6.15) are consistent with Fitts' law given the assumption that mouse movements satisfy Eq. (6.21).

We have left open the nature of the cost represented by the functional  $J[\cdot]$ . One choice for the cost is the metabolic work of the movement. In this case the minimization of the cost corresponds to making the movement so that the minimal amount of metabolic work is expended. We look at a model of human movement describing an individual exercise in which  $J[\cdot]$  represents the metabolic work in [149].

# 6.3 Appendix

We would like to find an approximation for the orbit x(t) of a computer mouse movement that begins motionless at a position -D and ends motionless at the origin at some time T. We begin with the equation of motion given in Eq. (6.16). Although Eq. (6.16) can be solved exactly using standard mathematical technique, it is more convenient to generate an approximate solution to Eq. (6.16) in the form of a truncated Taylor series expansion about time t = 0:

$$x(t) \approx x(0) + \dot{x}(0)t + \frac{1}{2}\ddot{x}(0)t^{2} + \frac{1}{6}\ddot{x}(0)t^{3} + \frac{1}{24}\ddot{x}(0)t^{4} + \frac{1}{120}\dot{x}(0)t^{5} + \frac{1}{720}\ddot{x}(0)t^{6}.$$
(6.23)

The unknown coefficients on the RHS of Eq. (6.23) must satisfy Eqs. (6.15) and (6.16), and as there are seven unknown coefficients in Eq. (6.23) and seven equations in Eqs. (6.15) and (6.16), there is a unique solution. Combining Eqs. (6.15) and (6.23) using the conditions for time t = 0, and evaluating Eq. (6.16) at time t = 0 gives the following system of equations:

$$\begin{aligned} x(t) &\approx -D + \frac{1}{6} \ddot{x}(0) t^{3} + \frac{1}{24} \ddot{x}(0) t^{4} + \frac{1}{120} \ddot{x}(0) t^{5} + \frac{1}{720} \ddot{x}(0) t^{6}, \\ \dot{x}(t) &\approx \frac{1}{2} \ddot{x}(0) t^{2} + \frac{1}{6} \ddot{x}(0) t^{3} + \frac{1}{24} \dot{x}(0) t^{4} + \frac{1}{120} \ddot{x}(0) t^{5}, \\ \ddot{x}(t) &\approx \ddot{x}(0) t + \frac{1}{2} \ddot{x}(0) t^{2} + \frac{1}{6} \dot{x}(0) t^{3} + \frac{1}{24} \ddot{x}(0) t^{4}, \\ \ddot{x}(0) &+ \phi_{33} \ddot{x}(0) = \phi_{11} D. \end{aligned}$$

$$(6.24)$$

Evaluating Eq. (6.24) at t = T gives:

$$D \approx \frac{1}{6} \ddot{x}(0) T^{3} + \frac{1}{24} \ddot{x}(0) T^{4} + \frac{1}{120} \dot{x}(0) T^{5} + \frac{1}{720} \ddot{x}(0) T^{6},$$
  

$$0 \approx \frac{1}{2} \ddot{x}(0) T^{2} + \frac{1}{6} \ddot{x}(0) T^{3} + \frac{1}{24} \dot{x}(0) T^{4} + \frac{1}{120} \ddot{x}(0) T^{5},$$
  

$$0 \approx \ddot{x}(0) T + \frac{1}{2} \ddot{x}(0) T^{2} + \frac{1}{6} \dot{x}(0) T^{3} + \frac{1}{24} \ddot{x}(0) T^{4},$$
  

$$\ddot{x}(0) + \phi_{33} \ddot{x}(0) = \phi_{11} D.$$
(6.25)

We can rewrite the system of equations in Eq. (6.25) compactly in matrix form as:

$$\begin{bmatrix} 720D \\ 0 \\ 0 \\ (\phi_{11}T^{6})D \end{bmatrix} \approx \begin{bmatrix} 120 & 30 & 6 & 1 \\ 60 & 20 & 5 & 1 \\ 24 & 12 & 4 & 1 \\ 0 & (\phi_{33}T^{2}) & 0 & 1 \end{bmatrix} \begin{bmatrix} \ddot{x}(0)T^{3} \\ \ddot{x}(0)T^{4} \\ \vdots \ddot{x}(0)T^{5} \\ \ddot{x}(0)T^{6} \end{bmatrix}.$$
(6.26)

We note that the values  $\phi_{11}T^6$  and  $\phi_{33}T^2$  are unitless, that is they are simply real numbers. We can simplify Eq. (6.26) using standard row operations. First, we use the third equation to eliminate terms in  $\ddot{x}(0)$  from the first and second equations:

$$\begin{bmatrix} 2880D \\ 0 \\ 0 \\ 0 \\ (\phi_{11}T^{6})D \end{bmatrix} \approx \begin{bmatrix} 336 & 48 & 0 & -2 \\ 120 & 20 & 0 & -1 \\ 24 & 12 & 4 & 1 \\ 0 & (\phi_{33}T^{2}) & 0 & 1 \end{bmatrix} \begin{bmatrix} \ddot{x}(0)T^{3} \\ \ddot{x}(0)T^{4} \\ \ddot{x}(0)T^{5} \\ \ddot{x}(0)T^{6} \end{bmatrix}.$$
(6.27)

Next, we use the fourth equation to eliminate terms in  $\ddot{x}(0)$  from the remaining equations:

$$\begin{bmatrix} (2880+2(\phi_{11}T^{6}))D\\ (\phi_{11}T^{6})D\\ -(\phi_{11}T^{6})D\\ (\phi_{11}T^{6})D \end{bmatrix} \approx \begin{bmatrix} 336 & 48+2(\phi_{33}T^{2}) & 0 & 0\\ 120 & 20+(\phi_{33}T^{2}) & 0 & 0\\ 24 & 12-(\phi_{33}T^{2}) & 4 & 0\\ 0 & (\phi_{33}T^{2}) & 0 & 1 \end{bmatrix} \begin{bmatrix} \ddot{x}(0)T^{3}\\ \ddot{x}(0)T^{4}\\ \dot{\ddot{x}}(0)T^{5}\\ \ddot{x}(0)T^{6} \end{bmatrix}.$$
(6.28)

Finally, we use the second equation to eliminate terms in  $\ddot{x}(0)$  from the first equation leaving:

$$\left( 336 \left( 20 + \left( \phi_{33} T^2 \right) \right) - 120 \left( 48 + 2 \left( \phi_{33} T^2 \right) \right) \right) \ddot{x} (0) T^3$$

$$\approx \left( \left( 20 + \left( \phi_{33} T^2 \right) \right) \left( 2880 + 2 \left( \phi_{11} T^6 \right) \right) - \left( 48 + 2 \left( \phi_{33} T^2 \right) \right) \left( \phi_{11} T^6 \right) \right) D.$$

$$(6.29)$$

This gives:

$$\ddot{x}(0) \approx \left(\frac{7200 + 360(\phi_{33}T^2) - (\phi_{11}T^6)}{120 + 54(\phi_{33}T^2)}\right) \frac{D}{T^3}.$$
(6.30)

The remaining terms  $\ddot{x}(0)$ ,  $\dot{\ddot{x}}(0)$ , and  $\ddot{\ddot{x}}(0)$  may be evaluated in a similar manner. If we assume that  $\phi_{11}T^6$  and  $\phi_{33}T^2$  is relatively small, then we can approximate Eq. (6.30) using a Taylor series expansion. Keeping to at most second order in  $\phi_{11}T^6$  and  $\phi_{33}T^2$ , we find that Eq. (6.30) is approximately:

$$\ddot{x}(0) \approx \left(60 - 24\phi_{33}T^2 - \frac{\phi_{11}T^6}{120} + \frac{54\phi_{33}^2T^4}{5} + \frac{9\phi_{11}\phi_{33}T^8}{20}\right)\frac{D}{T^3}.$$
(6.31)

# References

- McMorrow, K., Roeger, W., *The economic consequences of ageing populations*, in *European Economy - Economic Papers 138*. 1999, Directorate General Economic and Monetary Affairs, European Commission (DG ECFIN), European Commission.
- Bloom, D.E., Canning, D., Fink, G., *Implications of population ageing for* economic growth. Ox Rev Econ Pol, 2010. 26(4): p. 583-612.
- The state of aging and health in America 2013. 2013, Atlanta, GA: Centers for Disease Control and Prevention, US Dept of Health and Human Services.
- 4. Gray, A., *Population ageing and health care expenditure*. Ageing Horizons, 2005.
  2: p. 15-20.
- Kaye, J.A., Zitzelberger, T.A., *Overview of healthcare, disease, and disability*. Pervasive Computing in Healthcare, ed. J.E. Bardram, Mihailidis, A., Wan, D. 2006: CRC Press.
- 6. *The unsustainable cost of healthcare*. 2009, Social Security Advisory Board.
- Rechel, B., Doyle, Y., Grundy, E., McKee, M., *How can health systems respond* to population ageing. 2009, Health Systems and Policy Analysis - Policy Brief 10, World Health Organization.
- 8. Woolf, S.H., Huston, C.G., Lewin, L.S., Marks, J.S., Fielding, J.E., Sanchez, E.J., *The economic argument for disease prevention: Distinguishing between value and savings.* 2009, Partnership for Prevention.

- Banerjee, S., Wittenberg, R., Clinical and cost effectiveness of services for early diagnosis and intervention in dementia. Int J Geriatr Psychiatry, 2009. 24(7): p. 748-754.
- 10. Prince, M., Bryce, R., Ferri, C., *World Alzheimer Report 2011: The benefits of early diagnosis and intervention.* Alzheimer's Disease International, 2011.
- Getsios, D., Blume, S., Ishak, K.J., Maclaine, G., Hernández, L., An economic evaluation of early assessment for Alzheimer's disease in the United Kingdom. Alzheimers Dement, 2012. 8: p. 22-30.
- Arnrich, B., Mayora, O., Bardam, J., Tröster, G., *Pervasive healthcare: paving the way for a pervasive, user-centered and preventive healthcare model.* Methods Inf Med, 2010. 49(1): p. 67-73.
- Muñoz-Mendoza, C.L., Cabrero-García, J., Reig-Ferrer, A., Cabañero-Martínez, M.J., Evaluation of walking speed tests as a measurement of functional limitations in elderly people: A structured review. Int J Clin Hlth Psyc, 2010. 10(2): p. 359-378.
- 14. Himann, J.E., Cunningham, D.A., Rechnitzer, P.A., Paterson, D.H., *Age-related changers in speed of walking*. Med Sci Sports Exerc, 1988. **20**(2): p. 161-166.
- Schimpl, M., Moore, C., Lederer, C., Neuhaus, A., Sambrook, J., Danesh, J., Ouwehand, W., Daumer, M., Association between walking speed and age in healthy, free-living individuals using mobile accelerometry—A cross-sectional study. PLoS ONE, 2011. 6(8).
- Hardy, S.E., Perera, S., Roumani, Y.F., Chandler, J.M., Studenski, S.A., Improvement in usual gait speed predicts better survival in older adults. JAGS, 2007. 55: p. 1724-1734.

- Studenski, S., Perera, S., Patel, K., Rosano, C., Faulkner, K., Inzitari, M., Brach, J., Chandler, J., Cawthon, P., Connor, E.B., Nevitt, M., Visser, M., Kritchevsky, S., Badinelli, S., Harris, T., Newman, A.B., Cauley, J., Ferrucci, L., Guralnik, J., *Gait speed and survival in older adults*. JAMA, 2011. **305**(1): p. 50-58.
- Studenski, S., Perera, S., Wallace, D., Chandler, J.M., Duncan, P.W., Rooney, E., Fox, M., Guralnik, J.M., *Physical performance measures in the clinical setting*. JAGS, 2003. 51(3): p. 314-322.
- Bramell-Risberg, E., Jarnlo, G.B., Minthon, L., Elmståhl, S., *Lower gait speed in older women with dementia compared with controls*. Dement Geriatr Cogn Disord, 2005. 20(5): p. 298-305.
- Camicioli, R., Howieson, D., Oken, B., Sexton, G., Kaye, J., *Motor slowing precedes cognitive impairment in the oldest old*. Neurology, 1998. 50(5): p. 1496-1498.
- Beauchet, O., Allali, G., Berrut, G., Hommet, C., Dubost, V., Assal, F., *Gait analysis in demented subjects: Interests and perspectives*. Neuropsychiatr Dis Treat, 2008. 4(1): p. 155-160.
- Guralnik, J.M., Ferrucci, L., Simonsick, E.M., Salive, M.E., Wallace, R.B., Lower-extremity function in persons over the age of 70 years as a predictor of subsequent disability. N Engl J Med, 1995. 332(9): p. 556-561.
- Guralnik, J.M., Ferrucci, L., Pieper, C.F., Leveille, S.G., Markides, K.S., Ostir,
  G.V., Studenski, S., Berkman, L.F., Wallace, R.B., *Lower extremity function and* subsequent disability: consistency across studies, predictive models, and value of gait speed alone compared with the short physical performance battery. J
  Gerontol A Biol Sci Med Sci, 2000. 55(4): p. M221-M231.

- Welmer, A.K., Rizzuto, D., Qiu, C., Caracciolo, B., Laukka, E.J., Walking speed, processing speed, and dementia: A population-based longitudinal study. J Gerontol A Biol Sci Med Sci, 2014 Apr. 4 [Epub ahead of print].
- Holtzer, R., Verghese, J., Xue, X., Lipton, R.B., *Cognitive processes related to gait velocity: results from the Einstein Aging Study*. Neuropsychology, 2006.
   20(2): p. 215-223.
- 26. Fitzpatrick, A.L., Buchanan, C.K., Nahin, R.L., Dekosky, S.T., Atkinson, H.H., Carlson, M.C., Williamson, J.D., Gingko Evaluation of Memory (GEM) Study Investigators, *Associations of gait speed and other measures of physical function with cognition in a healthy cohort of elderly persons.* J Gerontol A Biol Sci Med Sci, 2007. **62**(11): p. 1244-1251.
- 27. Tombaugh, T.N., *Trail making test A and B: Normative data stratified by age and education*. Arch Clin Neuropsychol, 2004. **19**: p. 203-214.
- Giovagnoli, A.R., Del Pesce, M., Mascheroni, S., Simoncelli, M., Laiacona, M., Capitani, E., *Trail making test: normative values from 287 normal adult controls*. Ital J Neurol Sci, 1996. 17: p. 305-309.
- 29. Bowie, C.R., Harvey, P.D., *Administration and interpretation of the trail making test.* Nature Protocols, 2006. **1**(5): p. 2277-2281.
- Benedict, R.H.B., Zgaljardic, D.J., Practice effects during repeated administrations of memory tests with and without alternate forms. J Clin Exp Neuropsychol, 1998. 20(3): p. 339-352.
- Collie, A., Maruff, P., Darby, D.G., McStephen, M., *The effects of practice on the cognitive test performance of neurologically normal individuals assessed at brief test-retest intervals.* J Int Neuropsychol Soc, 2003. 9: p. 419-428.

- Bird, C.M., Papadoboulou, K., Ricciardelli, P., Rossor, M.N., Cipolotti, L., *Monitoring cognitive changes: psychometric properties of six cognitive tests.* Br J Clin Psychol, 2004. 43: p. 197-210.
- Rabin, L.A., Barr, W.B., Burton, L.A., Assessment practices of clinical neuropsychologists in the United States and Canada: A survey of INS, NAN, and APA division 40 members. Arch Clin Neuropsychol, 2005. 20: p. 33-65.
- Sanchez-Cubillo, I., Perianez, J.A., Adrover-Roig, D., Rodriguez-Sanchez, J.M., Rios-Lago, M., Tirapu, J., Barcelo, F., *Construct validity of the trail making test: Role of task-switching, working memory, inhibition/interference control, and visuomotor abilities.* J Int Neuropsychol Soc, 2009. 15: p. 438-450.
- 35. Salthouse, T.A., *What cognitive abilities are involved in trail-making performance?* Intelligence, 2011. **39**: p. 222-232.
- 36. Corrigan, J.D., Hinkeldey, N.S., *Relationships between parts A and B of the trail making test.* J Clin Psychol, 1987. **43**(4): p. 402-409.
- Arbuthnott, K., Frank, J., *Trail making test, part B as a measure of executive control: validation using a set-switching paradigm.* J Clin Exp Neuropsychol, 2000. 22(4): p. 518-528.
- Davies, A.D.M., *The influence of age on trail making test performance*. J Clin Psychol, 1968. 24(1): p. 96-98.
- Lindsey, B.A., Coppinger, N.W., Age-related deficits in simple capabilities and their consequences for trail making performance. J Clin Psychol, 1969. 25(2): p. 156-159.
- 40. Goul, W.R., Brown, M., *Effects of age and intelligence on trail making test performance and validity*. Percept Mot Skills, 1970. **30**(1): p. 319-326.

- Gordon, N.G., *The trail making test in neuropsychological diagnosis*. J Clin Psychol, 1972. 28(2): p. 167-169.
- 42. Gordon, N.G., *Diagnostic efficiency of the trail making test as a function of cutoff score, diagnosis, and age.* Percept Mot Skills, 1978. **47**(1): p. 191-195.
- Kennedy, K.J., Age effects on trail making test performance. Percept Mot Skills, 1981. 52(2): p. 671-675.
- Vazzana, R., Bandinelli, S., Lauretani, F., Volpato, S., Lauretani, F., Di Iorio, A., Abate, M., Corsi, A.M., Milaneschi, Y., Guralnik, J.M., Ferrucci, L., *Trail Making Test predicts physical impairment and mortality in older persons.* JAGS, 2010. 58(4): p. 719-723.
- 45. von Hippel, W., Ng, L., Abbot, L., Caldwell, S., Gill, G., Powell, K., *Executive functioning and gambling: performance on the trail making test is associated with gambling problems in older adult gamblers*. Neuropsychol Dev Cogn B Aging Neuropsychol Cogn, 2009. **16**(654-670).
- Classen, S., Horgas, A., Awadzi, K., Messinger-Rapport, B., Shechtman, O., Joo, Y., *Clinical predictors of older driver performance on a standardized road test*. Traffic Inj Prev, 2008. 9(5): p. 456-462.
- 47. Hobert, M.A., Niebler, R., Meyer, S.I., Brockmann, K., Becker, C., Huber, H., Gaenslen, A., Godau, J., Eschweiler, G.W., Berg, D., Maetzler, W., *Poor Trail Making Test performance Is directly associated with altered dual task prioritization in the elderly Baseline results from the TREND study* PLoS ONE, 2011. 6(11).
- 48. Ruffolo, L.F., Guilmette, T.J., Willis, G.W., *Comparison of time and error rates* on the trail making test among patients with head injuries, experimental

*malingerers, patients with suspect effort on testing, and normal controls.* Clin Neuropsychol, 2000. **14**(2): p. 223-230.

- Ashendorf, L., Jefferson, A.L., O'Connor, M.K., Chaisson, C., Green R.C., Stern, R.A., *Trail making test errors in normal aging, mild cognitive impairment, and dementia.* Arch Clin Neuropsychol, 2008. 23: p. 129-137.
- 50. Orwat, C., Graefe, A., Faulwasser, T., *Towards pervasive computing in health care A literature review*. BMC Medical Informatics and Decision Making, 2008.
   8(26).
- Bardram, J.E., *Pervasive healthcare as a scientific discipline*. Methods Inf Med, 2008. 47(3): p. 178-185.
- 52. Dishman, E., *Inventing wellness systems for aging in place*. Computer, 2004.37(4-34).
- 53. Varshney, U., *Pervasive healthcare*. Computer, 2003. **36**: p. 138-140.
- Wild, K., Boise, L., Lundell, J., Foucek, A., Unobtrusive in-home monitoring of cognitive and physical health: Reactions and perceptions of older adults. J Appl Gerontol, 2008. 27(2): p. 181-200.
- 55. Garber, A.M., *Long-term care, wealth, and health of the disabled elderly living in the community*, in *The economics of aging*, D.A. Wise, Editor. 1989, University of Chicago Press: Chicago.
- 56. Peterson, R.C., Stevens, J.C., Ganguli, M., Tangalos, E.G., Cummings, J.L., DeKosky, S.T., Practice parameter: early detection of dementia: Mild cognitive impairment (an evidence-based review): Report of the Quality Standards Subcommittee of the American Academy of Neurology. Neurology, 2001. 56(9): p. 1133-1142.

- 57. Reijinders, J., van Heugten, C., van Boxtel, M., Cognitive interventions in healthy older adults and people with mild cognitive impairment: a systematic review.
  Ageing Res Rev, 2013. 12(1): p. 263-275.
- Jimison, H.B., Pavel, M., *Integrating computer-based health coaching into elder home care*, in *Technology and aging*, A. Mihailidis, Boger, J., Kautz, H., Normie, L., Editor. 2008, IOS Press: Amsterdam, the Netherlands.
- 59. Jimison, H.B., Pavel, M., Bissell, P., McKanna, J., *A framework for cognitive monitoring using computer game interactions*. Stud Health Technol Inform, 2007.
  129: p. 1073-1077.
- 60. Andrade, C., Radhakrishnan, R., *The prevention and treatment of cognitive decline and dementia: An overview of recent research on experimental treatments.* Indian Journal of Psychiatry, 2009. **51**(1): p. 12-25.
- 61. Williams, K., Kemper, S., *Exploring interventions to reduce cognitive decline in aging*. J Psychosoc Nurs Ment Health Serv, 2010. **48**(5): p. 42-51.
- Theill, N., Schumacher, V., Adelsberger, R., Martin, M., Jänke, L., *Effects of simultaneously performed cognitive and physical training in older adults*. BMC Neuroscience, 2013. 14(103).
- 63. Bherer, L., Erickson, K.I., Liu-Ambrose, T., *A review of the effects of physical activity and exercise on cognitive and brain functions in older adults.* Journal of Aging Research, 2013.
- Gildner, T.E., Liebert, M.A., Kowal, P., Chatterji, S., Snodgrass, J.J., Associations between sleep duration, sleep quality, and cognitive test performance among older adults from six middle income countries: Results from the study on global ageing and adult health (SAGE). Journal of Clinical Sleep Medicine, 2014. 10(6): p. 613-621.

- 65. Jimison, H.B., Klein, K.A., Marcoe, J.L., A socialization intervention in remote health coaching for older adults in the home. Conf Proc IEEE Eng Med Biol Soc 2014, 2014: p. 7025-7028.
- 66. Paganelli, F., Giuli, D., *An ontology-based context model for home health monitoring and alerting in chronic patient care.* AINAW '07, 2007. **838-845**.
- Jimison, H.B., McKanna, J., Ambert, K., Hagler, S., Hatt, W.J., Pavel, M., Models of cognitive performance based on home monitoring data. Conf Proc IEEE Eng Med Biol Soc 2010, 2010: p. 5235-5237.
- Mileo, A., Merico, D., Bisiani, R., *Support for context-aware monitoring in home healthcare*. Journal of Ambient Intelligence and Smart Environments, 2010. 2: p. 49-66.
- 69. Dawadi, P.N., Cook, D.J., Schmitter-Edgecomb, M., *Automated assessment of cognitive health using smart home technologies*. Technol Health Care, 2013.
  21(4): p. 323-343.
- Dawadi, P.N., Cook, D.J., Schmitter-Edgecomb, M., Automated cognitive health assessment using smart home monitoring of complex tasks. IEEE Trans Syst Man Cybern Syst Hum, 2013. 43(6): p. 1302-1313.
- 71. Chan, M., Campo, E., Estève, D., *Assessment of activity of elderly people using a home monitoring system.* Int J Rehabil Res, 2005. **28**(1): p. 69-76.
- 72. Murphy, S.L., *Review of physical activity measurement using accelerometers in older adults: considerations of research design and conduct.* Prev Med, 2009.
  48: p. 108-114.

- Garatachea, N., Torres Luque, G., González Gallego, J., *Physical activity and* energy expenditure measurements using accelerometers in older adults. Nutr Hosp, 2010. 25(2): p. 224-230.
- 74. Yang, C.-C., Hsu, Y.-L., *A review of accelerometry-based wearable motion detectors for physical activity monitoring*. Sensors, 2010. **10**: p. 7772-7788.
- 75. Cheung, V.H., Gray, L., Karunanithi, M., *Review of accelerometry for determining daily activity among elderly patients*. Arch Phys Med Rehabil, 2011.
  92: p. 998-1014.
- Bento, T., Cortinhas, A., Leitão, J.C., Mota, M.P., *Use of accelerometry to measure physical activity in adults and the elderly*. Rev Saude Publica, 2012.
  46(3): p. 561-570.
- 77. Patel, S., Park, H., Bonato, P., Chan, L., Rodgers, M., A review of wearable sensors and systems with application in rehabilitation. Journal of NeuroEngineering and Rehabilitation, 2012. 9(21).
- Goble, D.J., Cone, B.L., Fling, B.W., Using the Wii Fit as a tool for balance assessment and neurorehabilitation: The first half decade of "Wii-search". Journal of NeuroEngineering and Rehabilitation, 2014. 11(12).
- 79. Stone, E.E., Skubic, M., *Unobtrusive, continuous, in-home gait measurement using the Microsoft Kinect.* IEEE Trans Biomed Eng, 2013. **60**(10): p. 2952-2953.
- Wang, F., Stone, E., Skubic, M., Keller, J.M., Abbott, C., Rantz, M., *Toward a passive low-cost in-home gait assessment system for older adults*. IEEE J Biomed Health Inform, 2013. 17(2): p. 346-355.

- Beattie, Z.T., Hayes, T.L., Guilleminault, C., Hagen, C.C., Accurate scoring of the apnea-hypopnea index using a simple non-contact breathing sensor. J Sleep Res, 2013. 22(3): p. 356-362.
- 82. Lê, Q., Nguyen, H.B., Barnett, T., *Smart homes for older people: Positive aging in a digital world*. Future Internet, 2012. **4**: p. 607-617.
- Morris, M.E., Adair, B., Miller, K., Ozanne, E., Hansen, R., Pearce, A.J., Santamaria, N., Viegas, L., Long, M., Said, C.M., *Smart-home technologies to assist older people to live well at home*. Journal of Aging Science, 2013. 1(1).
- Ariani, A., Redmond, S.J., Chang, D., Lovell, N.H., *Simulated unobtrusive falls detection with multiple persons*. IEEE Trans Biomed Eng, 2012. **59**(11): p. 3185-3196.
- Rantz, M.J., Skubic, M., Abbott, C., Galambos, C., Pak, Y., Ho, D.K., Stone,
   E.E., Rui, L., Back, J., Miller, S.J., *In-home fall risk assessment and detection sensor system.* J Gerontol Nurs, 2013. **39**(7): p. 18-22.
- Rashidi, P., Mihailidis, A., A survey on ambient-assisted living tools for older adults. IEEE J Biomed Health Inform, 2013. 17(3): p. 579-590.
- Kaye, J.A. and T.A. Zitzelberger, *Overview of Healthcare, Disease, and Disability*, in *Pervasive Computing in Healthcare*, J.E. Bardram, A. Mihailidis, and D. Wan, Editors. 2006, CRC Press.
- 88. Kaye, J., *Home-based technologies: a new paradigm for conducting dementia prevention trials.* Alzheimers Dement, 2008. **4**(1 Suppl 1): p. S60-6.
- Aminian, K., Robert, P., Jéquier, E., Schutz, Y., *Incline, speed, and distance assessment during unconstrained walking.* Med Sci Sports Exerc, 1995. 27(2): p. 226-234.

- 90. Miyazaki, S., Long-term unrestrained measurement of stride length and walking velocity utilizing a piezoelectric gyroscope. IEEE Trans Biomed Eng, 1997.
  44(8): p. 753-759.
- Sabatini, A.M., Martelloni, C., Scapellato, S., Cavallo, F., Assessment of walking features from foot inertial sensing. IEEE Trans Biomed Eng, 2005. 52(3): p. 486-494.
- 92. Hayes, T.L., Pavel, M., Larimer, N., Tsay, I.A., Nutt, J., Adami, A.G., *Distributed healthcare: Simultaneous assessment of multiple individuals*. IEEE Pervasive Comput, 2007. 6(1): p. 36-43.
- 93. Hayes, T.L., Abendroth, F., Adami, A., Pavel, M., Zitzelberger, T.A., Kaye, J.A., Unobtrusive assessment of activity patterns associated with mild cognitive impairment. Alzheimers Dement, 2008. 4(6): p. 395-405.
- 94. Hayes, T.L., Hagler, S., Austin, D., Kaye, J., Pavel, M., Unobtrusive assessment of walking speed in the home using inexpensive PIR sensors. Conf Proc IEEE Eng Med Biol Soc 2009, 2009: p. 7248-7251.
- 95. Bjork, A., Numerical methods for least squares problems. 1996: SIAM.
- 96. Golub, G.H., Van Loan, C.F., *Matrix computations*. 3rd ed. 1996: The John Hopkins University Press.
- 97. Schatz, P., Browndyke, J., *Applications of computer-based neuropsychological assessment*. J Head Trauma Rehabil, 2002. **17**(5): p. 395-410.
- Wild, K., Howieson, D., Webbe, F., Seelye, A., Kaye, J., *The status of computerized cognitive testing in aging: A systematic review*. Alzheimers Dement, 2008. 4(6): p. 428-437.

- 99. Espeland, M.A., Katula, J.A., Rushing, J., Kramer, A.F., Jennings, J.M., Sink, K.M., Nadkarni, N.K., Reid, K.F., Castro, C.M., Church, T., Kerwin, D.R., Williamson, J.D., Marottoll, R.A., Rushing, S., Marsiske, M., Rapp, S.R., LIFE Study Group, *Performance of a computer-based assessment of cognitive function measures in two cohorts of seniors*. Int J Geriatr Psychiatry, 2013. 28(12): p. 1239-1250.
- Salthouse, T.A., Fristoe, N.M., Process analysis of adult age effects on a computer-administered Trail Making Test. Neuropsychology, 1995. 9(4): p. 518-528.
- 101. Drapeau, C.E., Bastien-Toniazzo, M., Rous, C., Carlier, M., Nonequivalence of computerized and paper-and-pencil versions of Trail Making Test. Percept Mot Skills, 2007. 104(3 Pt 1): p. 785-791.
- 102. Poreh, A., Miller, A., Dines, P., Levin, J., Decomposition of the Trail Making Test
   Reliability and validity of a computer assisted method for data collection.
  Archives of Assessment Psychology, 2012. 2(1).
- 103. Donders, F.C., On the speed of mental processes. Acta Psychol (Amst), 1869/1969. 30: p. 412-431.
- 104. Sternberg, S., *The discovery of processing stages*. Acta Psychol (Amst), 1969. 30: p. 276-315.
- 105. Fitts, P.M., *The information capacity of the human motor system in controlling the amplitude of movement.* J Exp Psychol Gen, 1954/1992. **121**(3): p. 262-269.
- Fitts, P.M., Peterson J.R., *Information capacity of discrete motor responses*. J Exp Psychol Gen, 1964. 67(2): p. 103-112.

- MacKenzie, I.S., A note on the information-theoretic basis for Fitts' law. J Motor Behav, 1989. 21: p. 323-330.
- Treisman, A.M., Gelade, G., *A feature-integration theory of attention*. Cognit Psychol, 1980. 12: p. 97-136.
- 109. Jimison, H., Pavel, M., McKanna, J., Pavel, J., Unobtrusive monitoring of computer interactions to detect cognitive status in elders. IEEE Trans Inf Technol Biomed, 2004. 8(3): p. 248-252.
- MacKenzie, I.S., Movement time prediction in human-computer interfaces, in Readings in Human Computer Interaction, R.M. Baecker, Buxton, W.A.S., Grudin, J., Greenberg, S., Editor. 1995, Kaufmann: Los Altos, CA. p. 483-493.
- 111. Monsell, S., Task switching. Trends Cogn Sci, 2003. 7(3): p. 134-140.
- 112. Sperling, G., Melchner, M.J. Visual search and visual attention. in Proceedings of the Fourth Symposium of Sensory System Physiology: Information Processing in Visual System. 1976: Leningrad: Academy of Sciences, Pavlov Institute of Physiology.
- 113. Kaye, J., Mattek, N., Dodge, H.H., Campbell, I., Hayes, T., Austin, D., Hatt, W.,
  Wild, K., Jimison, H., Pavel, M., *Unobtrusive measurement of daily computer use* to detect mild cognitive impairment. Alzheimers Dement, 2014. 10(1): p. 10-17.
- 114. Austin, D., Jimison, H., Hayes, T., Mattek, N., Kaye, J., Pavel, M., Measuring motor speed through typing: a surrogate for the finger tapping test. Behav Res Meth, 2011. 43(4): p. 903-909.
- 115. Peterson, J., Austin, D., Kaye, J.A., Pavel, M., Hayes, T., Unobtrusive in-home detection of time spent out-of-home with applications to loneliness and physical activity. IEEE J Biomed Health Inform, 2013 Dec 13. [Epub ahead of print].

- Hayes, T.L., Riley, T., Mattek, N., Pavel, M., Kaye, J.A., *Sleep habits in mild cognitive impairment*. Alzheimer Dis Assoc Disord, 2014. 28(2): p. 145-150.
- 117. Jung, D.W., Hwang, S.H., Yoon, H.M., Lee, Y.J., Jeong, D.-U., Park, K.S., Nocturnal awakening and sleep efficiency estimation using unobtrusively measured ballistocardiogram. IEEE Trans Biomed Eng, 2014. 61(1): p. 131-138.
- Hagler, S., Jimison, H., Pavel, M., *Unobtrusive in-home assessment by means of everyday computer mouse usage*. Conf Proc IEEE Eng Med Biol Soc 2011, 2011: p. 3816-3819.
- Shen, C., Cai, Z., Guan, X., Du, Y., Maxion, R.A., User authentication through mouse dynamics. IEEE Trans Inf Forensics Security, 2013. 8(1): p. 16-30.
- Ahmed, A.A.E., Traore, I., A new biometric technology based on mouse dynamics. IEEE T Depend Secure, 2007. 4(3): p. 165-179.
- 121. Feher, C., Elovici, Y., Moskovitch, R., Rokach, L., Schclar, A., *User identity verification via mouse dynamics*. Inf Sci, 2012. **201**: p. 19-36.
- Gorad, B.J., Kodavade, D.V., User identity verification using mouse signature. IOSR-JCE, 2013. 12(4): p. 33-36.
- 123. Katsumata, Y., Todoriki, H., Yasura, S., Dodge, H.H., *Timed up and go test predicts cognitive decline in healthy adults aged 80 and older in Okinawa: Keys to optimal cognitive aging (KOCOA) project.* JAGS, 2011. **59**(11): p. 2188-2189.
- 124. Donaghue, O.A., Horgan, N.F., Savva, G.M., Cronin, H., O'Regan, C., Kenny, R.A., Association between timed up-and-go and memory, executive function, and processing speed. JAGS, 2012. 60(9): p. 1681-1686.

- 125. Buracchio, T., Dodge, H.H., Howieson, D., Wasserman, D., Kaye, J., *The trajectory of gait speed preceding mild cognitive impairment*. Arch Neurol, 2010.
  67(8): p. 980-986.
- 126. Dodge, H.H., Mattek, N.C., Austin, D., Hayes, T.L., Kaye, J.A., *In-home walking speeds and variability trajectories associated with mild cognitive impairment*. Neurology, 2012. **78**(24): p. 1946-52.
- 127. Salazar-Fraile, J., Balanzá-Martínez, V., Selva-Vera, G., Martínez-Aran, A., Sánchez-Moreno, J., Rubio, C., Vieta, E., Gómez-Beneyto, M., Tabarés-Seisdedos, R., *Motor speed predicts stability of cognitive deficits in both schizophrenic and bipolar I patients at one-year follow-up*. Eur J Psychiat, 2009.
  23(3): p. 184-197.
- 128. Dale, R., Duran, N.D., *The cognitive dynamics of negated sentence verification*. Cogn Sci, 2011. 35(5): p. 983-996.
- 129. Freeman, J.B., Ambady, N., Hand movements reveal the time-course of shape and pigmentation processing in face categorization. Psychon Bull Rev, 2011. 18(4): p. 705-712.
- Papesh, M.H., Goldinger, S.D., *Memory in motion: Movement dynamics reveal memory strength*. Psychonomic Bulletin & Review, 2012. 19(5): p. 906-913.
- Xiao, K., Yamauchi, T., Semantic priming revealed by mouse movement trajectories. Conscious Cogn, 2014. 27: p. 42-52.
- 132. Pointer ballistics for Windows XP. 2002, Microsoft Corporation.
- 133. Nubar, Y., Contini, R., *A minimum principle in biomechanics*. Bull Math Biol, 1961. 23(4): p. 377-391.

- Nelson, W.L., *Physical principles for economies of skilled movement*. Biol Cybern, 1983. 46: p. 135-147.
- Hogan, N., An organizing principle for a class of voluntary movements. J Neurosci, 1984. 4(11): p. 2745-2754.
- 136. Flash, T., Hogan, N., *The coordination of arm movements: an experimentally confirmed mathematical model.* J Neurosci, 1985. **5**(7): p. 1688-1703.
- 137. Uno, Y., Kawato, M., Suzuki, R., *Formation and control of optimal trajectory in human multijoint arm movement*. Biol Cybern, 1989. **61**: p. 89-101.
- 138. Pandy, M.G., Garner, B.A., Anderson, F.C., Optimal control of non-ballistic muscular movements: a constraint-based performance criterion for rising from a chair. J Biomech Eng, 1995. 117(1): p. 15-26.
- Goldstein, H., Poole, C.P., Safko, J.L., *Classical mechanics*. 3rd ed. 2001: Addison-Wesley.
- 140. Elsgolc, L.E., *Calculus of variations*. 1962, Reading, MA: Addison-Wesley Publishing Company, Inc.
- 141. Davis, A.S., Pierson, E.E., *The relationship between the WAIS-III digit symbol coding and executive functioning*. Appl Neuropsychol Adult, 2012. 19(3): p. 192-197.
- 142. Card, S.K., Moran, T.P., Newell, A., *The psychology of human-computer interaction*. 1986: Lawrence Erlbaum Associates.
- 143. Kvålseth, T.O., *An alternative to Fitts' law.* B Psychonomic Soc, 1980. 16(5): p. 371-373.

- 144. McCaffery, R.J., Krahula, M.M., Heimberg, R.G., An analysis of the significance of performance errors on the trail making test in polysubstance users. Arch Clin Neuropsychol, 1989. 4: p. 393-398.
- 145. Mahurin, R.K., Velligan, D.I., Hazleton, B., Mark Davis, J., Eckert, S., Miller, A.L., *Trail making test errors and executive function in schizophrenia and depression*. Clin Neuropsychol, 2006. 20(2): p. 271-288.
- 146. Grieve, D. W., *Gait patterns and the speed of walking*. Biomed Eng (NY), 1968.3: p. 119-122.
- 147. Zia, R.K.P., Redish, E.F., McKay, S.R., *Making sense of the Legendre transform*. Am J Phys, 2009. **77**(7): p. 614-622.
- Kirk, D.E., *Optimal control theory: An introduction*. 1970, Englewood Cliffs, NJ: Prentice-Hall.
- Hagler, S., Jimison, H., Bajcsy, R., Pavel, M., *Quantification of human movement for assessment in automated coaching*. Conf Proc IEEE Eng Med Biol Soc 2014, 2014 (to be published).